

Empowering people who serve the public Questions

Remittance

Tyler Technologies, Inc. (FEIN 75-2303920) P.O. Box 203556 Dallas, TX 75320-3556

THIS IS NOT AN INVOICE **PROFORMA**

Company	Order No.	Date	Page
020	5161	08/16/2016	1 of 2

Tyler Technologies - Courts & Justice Phone: 1-800-772-2260 Press 2, then 3

Fax: 1-866-673-3274 Email: ar@tylertech.com

To: Fort Bend County Information Technology

301 Jackson Street Attn: Robyn Doughtie Richmond, TX 77469

Ship To: Fort Bend County Information Technology

301 Jackson Street Attn: Robyn Doughtie Richmond, TX 77469

	Cust	omer Grp/No.	Customer PO#	Payr	nent Terms	Currency Cod	de	Ship Via	Salesperso	on Cd
	1	45414			Net 30	USD		MISC		
No. It	tem/ Des	cription/ Comments	Dro	op Ship	# Users	Quantity	U/M	Unit Price	Disc %	Total Cost
ract No	o.: C&J									
1	Ft. Ber		Гуре Maintenance and Suppor 0/01/2016, End: 09/30/2017;		1 12 months	1	EA	98,283.97	.00	98,283.9
2	Ft. Ber		se Type Maintenance and Sup 0/01/2016, End: 09/30/2017;	•	1 12 months	1	EA	52,922.13	.00	52,922.13
3	Ft. Ber		(per office) Maintenance and 0/01/2016, End: 09/30/2017;		1 12 months	1	EA	9,720.39	.00	9,720.3
4	Ft. Ber		Standard Maintenance and Sup 0/01/2016, End: 09/30/2017;	•	1 12 months	1	EA	16,412.76	.00	16,412.7
5	Brazos	ral: 6OTXTTWANSM Technology Ticket Writ nance: Start: 10/01/20	er Maintenance 16, End: 09/30/2017; Term:	No 12 mo	1 nths	1	EA	1,069.99	.00	1,069.9
6	Ft Ben	_	es Edition Software Maintenar 16, End: 09/30/2017; Term:		1 nths	1	EA	39,769.38	.00	39,769.3
7	Ft Ben		ed to TIFF Software Maintena 16, End: 09/30/2017; Term:		1 nths	1	EA	13,046.04	.00	13,046.0
8	eNotice	ral: 6OCMEANSM es Standard Annual Mai nance: Start: 10/01/20	intenance 16, End: 09/30/2017; Term:	No 12 mo	1 nths	1	EA	9,450.00	.00	9,450.0



Empowering people who serve the public® Questions

Remittance

Tyler Technologies, Inc. (FEIN 75-2303920) P.O. Box 203556 Dallas, TX 75320-3556

THIS IS NOT AN INVOICE PROFORMA

Company	Order No.	Date	Page
020	5161	08/16/2016	2 of 2

Tyler Technologies - Courts & Justice Phone: 1-800-772-2260 Press 2, then 3

Fax: 1-866-673-3274 Email: ar@tylertech.com

No. Item/ Description/ Comments Drop Ship # Users Quantity U/M Unit Price Disc % Total Cost

	Does not include any applicable taxes	Order Total:	240,674.66
We	would like to continue on the current maintenance plan		
	Our Purchase Order is enclosed		
	Our Purchase Order is		
	Purchase Order is not required to be invoiced		
	AN INVOICE WILL BE SENT <u>AFTER</u> CONFIRMATION OF ORDER		
Comme	nts:		

AMENDMENT

This amendment ("Amendment") is made this day of 2015 by and between Tyler Technologies, Inc. ("Tyler") and Fort Bend County, Texas ("End User").						
WHEREAS, Tyler and the End User are parties to a certain Revised End User License Agreement and Professional Services Agreement dated January 5, 2010 ("Agreement"); and						
WHEREAS, End User was party to an agreement with the Agreement with Tyler; and	e CUC whereby End User was authorized to enter into the					
WHEREAS, End User has discontinued its agreement with	ith the CUC; and					
WHEREAS, End User desires to continue to use the Lice provided in the Agreement and to remove references to, a						
NOW THEREFORE, in consideration of the mutual profollows:	mises hereinafter contained, Tyler and End User agree as					
 Maintenance and Support Services Agreement, Sentirety and replaced with the following as of the 	Section 11.1 (a) of the Agreement is hereby deleted in its effective date of this Amendment:					
	nnually in advance for the Maintenance and Support Fees due newal of the Maintenance and Support Term and End User					
termination of its agreement with the CUC is not of the Agreement. The parties further expressly a validly entered into while the CUC purchasing a	nent, the parties understand and agree that End User's tintended, and shall not be deemed, to result in a termination agree that because the Agreement was appropriately and greements were in place, it shall continue as a direct ng termination of the CUC purchasing agreement.					
 This Amendment shall be governed by and const Agreement. 	trued in accordance with the terms and conditions of the					
4. All other terms and conditions of the Agreement	shall remain in full force and effect.					
IN WITNESS WHEREOF, the parties hereto have execu	atted this Amendment as of the dates set forth below.					
Tyler Technologies, Inc. Courts & Justice Division	Fort Bend County, Texas					
Ву:	By: Jelley					
Name: Mitchell Spence	Name: Robert E. Hebert					
Title: VP of Customer Success & Support Services	Title: Fort Bend county Judge					
Date: 7 (8 / 2) (5	Date: July 28, 2015					



December 04, 2009

Chairman

Judge Mike Bradford

Midland County

Chair Elect

Commissioner Mike Cantrell Dallas County

Immediate Past Chair

Judge Glen Whitley

Glen Whitley Tarrant County

Vice-Chairmen

Comm. Tommy Adkisson Bexar County

Comm. Sylvia Garcia Harris County

Comm. Andy Meyers Fort Bend County

Comm. Jay Millikin Comal County

Executive Director

Donald Lee

Member Counties

Bell ~ Bexar Brazoria ~ Brazos

Cameron ~ Chambers

Collin ~ Comal

Dallas ~ Denton

El Paso ~ Ellis

Fort Bend ~ Galveston

Grayson ~ Gregg

Guadalupe ~ Harris

Hays ~ Hidalgo

Hunt ~ Jefferson

Johnson ~ Kaufman

Lubbock ~ McLennan

Midland ~ Nueces Rockwall ~ San Patricio

Tarrant ~ Travis

Williamson ~ Wise

500 West 13th Street Austin, TX 78701

Phone: 512.476.6174 Fax: 512.476.5122

www.cuc.org

Ms. Brandy Shepherd Tyler Technologies, Inc 6500 International Parkway, Suite 2000

Re: Tyler Contract Amendment for Fort Bend County

Dear Ms. Shepherd:

Plano TX 75093

Enclosed with this letter is 1 original document of the Master Software License and Professional Services Agreement-Amendment 1 signed by Mr. Don Lee and County Judge from Fort Bend County.

Please execute the contract and please return it to Texas Conference of Urban Counties as soon as possible.

Should you have any questions regarding this matter, please do not hesitate to contact me at (512) 476-6174.

Sincerely,

Nadia Budhiwan TechShare Program

Texas Conference of Urban Counties

TEXAS Conference of Urban Counties



October 23, 2009

Chairman

Judge Mike Bradford Midland County

Chair Elect

Commissioner Mike Cantrell Dallas County

Immediate Past Chair

Judge Glen Whitley Tarrant County

Vice-Chairmen

Comm. Tommy Adkisson Bexar County

Comm. Sylvia Garcia Harris County

Comm. Andy Meyers Fort Bend County

Comm. Jay Millikin Comal County

Judge J.D. Salinas Hidalgo County

Executive Director Donald Lee

Member Counties

Bell ~ Bexar Brazoria ~ Brazos Cameron ~ Chambers Collin ~ Comal Dallas ~ Denton El Paso ~ Ellis Fort Bend ~ Galveston Grayson ~ Gregg Guadalupe ~ Harris Hays ~ Hidalgo Hunt ~ Jefferson Johnson ~ Kaufman Lubbock ~ McLennan Midland ~ Nueces Randall ~ Rockwall San Patricio ~ Tarrant

500 West 13th Street Austin, TX 78701

Travis ~ Williamson

Wise

Phone: 512.476.6174 Fax: 512.476.5122 www.cuc.org Honorable Bob Hebert County Judge Fort Bend County 301 Jackson, Ste. 719 Richmond, TX 77469

Re: Consideration of Changes to the Agreement with the Tyler Technologies for the Common Integrated Justice System (CIJS) Court Administration System

Dear Judge Hebert:

Thank you for Fort Bend County's continued participation in the Urban Counties TechShare program, sharing the Common Integrated Justice System Court Administration System called Odyssey, licensed from Tyler Technologies through a TechShare procured enterprise license agreement.

By joining and working together, participating counties have saved more than \$30 million sharing this license agreement. Fort Bend County has saved \$2.5 million.

Now, action is required of the Fort Bend County Commissioners Court to continue this strong partnership and potentially enhance the savings to Fort Bend County taxpayers. The action does not require any increase in costs.

In our agreements with Tyler, deadlines for adding additional counties and realizing rebates to existing counties have passed. The Urban Counties has negotiated an amendment that extends agreements and enhances the benefits to participating counties. Only counties that vote to adopt the amended agreement will enjoy the enhanced benefits.

The amendment has been reviewed by, and is recommended by, the CJIS Oversight Board on which Fort Bend County is represented. The amendment has also been approved by, and is recommended by, the Urban Counties Board of Directors.

A memo providing a comprehensive analysis is attached. In summary, the amendment:

- Extends the deadline past 12/31/09 for additional counties to join in sharing the license agreement;
- Pays participating counties \$800,000 total (distributed pro-rata) if all license existing participants completely pay for their court case management system licenses by December 31, 2009;

- Removes a cap on payments to Tyler by new counties joining in the resource sharing (this cap is not going to be reached by December 31, 2009);
- Extends the popular enterprise modifications program that is saving participating counties more than \$200,000 each year; and
- Continues the reduced annual software maintenance fees, assuring the counties of continued savings through the program.

If Fort Bend County ratifies the amendment, the County will continue to realize the benefits from the collaborative effort, including the potential for the license fee rebates included in the amendment.

Section 25.7 of the Master Software License and Professional Services Agreement requires that all participating counties ratify the proposed contract amendment. Therefore, if the County does not ratify the amendment, then all participating counties will forfeit any potential rebates and future benefits from the amended agreement. The right to use the Odyssey system will of course continue, under the current agreement.

We request your immediate consideration of this amendment. If the participating counties do not approve the amendment by November 25, 2009 then the benefits of it will be lost.

Please adopt the attached draft resolution, or a similar resolution that approves the attached Amendment No. 1 to the Master Software License and Professional Services Agreement with Tyler Technologies.

If you would like additional information please do not hesitate to contact me. Either myself or Charles Gray are at your disposal either by phone or to visit your county and discuss this matter.

Thank you again for your participation in the Urban Counties and the TechShare program.

Sincerely,

Donald Lee

Executive Director

cc: Honorable Richard Morrison, Fort Bend County Commissioner, Pct. 1
Honorable Grady Prestage, Fort Bend County Commissioner, Pct. 2
Honorable Andy Meyers, Fort Bend County Commissioner, Pct. 3
Honorable James Patterson, Fort Bend County Commissioner, Pct. 4
Ms. Connie Heinecke, Fort Bend County IT Projects Manager

Enclosed: Detailed Memo from Charles Gray

Draft Commissioners Court Resolution

Master Software License and Professional Services Agreement Amendment No.

1



October 23, 2009

Chairman

Judge Mike Bradford Midland County

Chair Elect

Commissioner Mike Cantrell Dallas County

Immediate Past Chair

Judge Glen Whitley Tarrant County

Vice-Chairmen

Comm. Tommy Adkisson Bexar County

Comm. Sylvia Garcia Harris County

Comm. Andy Meyers Fort Bend County

Comm. Jay Millikin Comal County

Judge J.D. Salinas Hidalgo County

Executive Director Donald Lee

Member Counties

Bell ~ Bexar
Brazoria ~ Brazos
Cameron ~ Chambers
Collin ~ Comal
Dallas ~ Denton
El Paso ~ Ellis
Fort Bend ~ Galveston
Grayson ~ Gregg
Guadalupe ~ Harris
Hays ~ Hidalgo
Hunt ~ Jefferson
Johnson ~ Kaufman
Lubbock ~ McLennan
Midland ~ Nueces

500 West 13th Street Austin, TX 78701

Rockwall ~ San Patricio Tarrant ~ Travis Williamson ~ Wise

Phone: 512.476.6174 Fax: 512.476.5122 www.cuc.org To: Judge Hebert, Fort Bend County

CC: Ms. Connie Heinecke, IT Projects Manager

From: Charles Gray, TechShare Program Manager

Re: Consideration of Changes to the Agreement with the Tyler

Technologies for the Common Integrated Justice System (CIJS)

Court Administration System

The purpose of this memo is to request that Fort Bend County ratify the amendment to the contract between Texas Conference of Urban Counties and Tyler Technologies making the Odyssey Court Case Management System available to our members.

The contract amendment was adopted by the Urban Counties Board of Directors on Wednesday, September 8, 2009.

Each of the member counties participating in the collaborative project to implement the Odyssey Court Case Management System through the Urban Counties TechShare program must determine whether or not it will ratify the contract amendment.

This letter provides an overview of the business reasons that lead to this amendment and, further, describes the contents of the amendment at a high level.

By ratifying this agreement, Fort Bend County will continue to realize the benefits from sharing the Odyssey Court Administration System through the Urban Counties TechShare program.

The Urban Counties will provide a detailed briefing for Fort Bend County upon request.

Action on this item is requested by November 25, 2009. The amendment is included with this letter. Please return a signed copy of the amendment to the Urban Counties by November 25, 2009. If Fort Bend County decides not to ratify the amendment, please advise the Urban Counties of that decision so that the County's contract can be transferred directly to Tyler Technologies.

The Master Software License and Professional Services Agreement requires that all participating counties ratify any amendment. If Fort Bend County elects not to ratify the amendment, then none of the participating counties will realize the benefits included in the amendment.

Fort Bend County; CIJS Contract Changes with Tyler Technologies October 23, 2009, Page 2 of 5

Background:

In March 2006, the Urban Counties entered into a contract with Tyler Technologies to provide the option to license and implement a Court Administration System, along with other justice system modules, to the member counties. The contract anticipated that, over a three year period, the Urban Counties could obtain an "enterprise" software license that would provide unlimited use of the Court Administration System for member counties by enrolling a sufficient number of counties to pay the combined enterprise license fee of \$12,395,000.

Over the next three years, the Urban Counties, working closely with Tyler Technologies, enrolled thirteen (13) counties in the contract, with a combined license fee commitment (for the Court Administration System) of \$10,652,500.

In order to continue to offer the Court Administration System to member counties and to extend the collaborative efforts of the participating counties, an amendment to the original contract is needed. Collaborating with Tyler Technologies new contract documents have been developed extending the relationship. The new agreement would be effective if approved by the Urban Counties Board and the participating member counties.

The proposed contract documents have been reviewed with the CIJS Oversight Board whose members recommended adoption by the Urban Counties Board. The Urban Counties Board of Directors approved the contract amendment on September 8, 2009. A summary of the contract changes are presented below.

Contract Amendment Overview:

The process for amending the Master Software License and Professional Services agreement between the Texas Conference of Urban Counties and Tyler Technologies includes two steps:

- Approve Amendment No. 1 to the Master Software License and Professional Services Agreement to provide for additional benefits to the participating counties and to effectively close out the current Tyler contract with the Urban Counties; and
- Replace the Master Software License and Professional Services Agreement with a new Master Agreement that provides the member counties with the continued opportunity to acquire the Court Administration System and other optional modules from Tyler Technologies through December 2014 at significant cost savings.

This approach resolves the issues with the current agreement and allows the TechShare program to continue the collaborative effort with Tyler Technologies that has delivered significant value and savings to the participating counties. The elements of each step of the project are outlined below:

Approve Amendment No. 1 to the Master Software License and Professional Services Agreement:

- Resolves Urban Counties/Tyler issues regarding license fee payment by eliminating the enterprise license fee.
- Requires that pending license fees for participating counties be paid through the Urban Counties by the end of December 2009.
- If all pending license fees are paid by the end of December 2009, provides a license fee rebate in the amount of \$800,000 to be distributed to participating counties by the Urban Counties.
- Continues current implementation projects for the thirteen (13) participating counties with no increase in cost.
- Provides Tyler with a twelve percent (12%) increase in hourly services rates for any change orders or new projects executed for the thirteen (13) participating counties, effective immediately.
- Continues the current Software Maintenance agreement terms for the
 participating counties at the same low rate of 16% of license fees, saving the
 counties about 25% over the industry standard maintenance fee rates of 22 to
 24% of license fees.
- Provides the option for the participating counties to expand Software
 Maintenance services to include expanded help desk coverage through an incremental increase in annual software maintenance fees.
- Continues administration and payment of Software Maintenance services through the Urban Counties so counties can leverage their collaborative purchasing structure.
- Provides for the completion of all approved enterprise modifications up to the contract value of \$661,986, providing almost \$2 million in value to participating counties.
- Closes out the current contract on December 31, 2009 while recognizing that some contract commitments, such as ongoing implementation projects and maintenance contracts, survive the termination of the agreement.

New Master Agreement:

- Provides the Urban Counties members who have not already elected to participate in the program with the option to obtain the Court Administration System, optional justice software modules such as Jail Management, Adult Probation, Sheriff's Records Management and Hot Checks at a discounted license fee of approximately 10% through December 31, 2010.
- Provides all counties in Texas with the option to obtain the Court Administration System, optional justice software modules such as Jail Management, Adult Probation, Sheriff's Records Management and Hot Checks through the Urban Counties through December 31, 2014.

Fort Bend County; CIJS Contract Changes with Tyler Technologies October 23, 2009, Page 4 of 5

- Provides an additional license fee rebate of \$390,000 if the participating county license fees are expanded by a minimum of \$2 million by the end of December 2010. The rebate would be paid to the Urban Counties to be distributed to the participating member counties.
- Provides Software Maintenance Services, including expanded Help Desk and Software Assurance Services, at discounted fees for all participating counties.
 Basic maintenance would be offered at a 25% savings by establishing the maintenance fees at 16% of licenses fee instead of the industry standards of 22 to 24%.
- Expands the enterprise modification program so participating counties can continue to obtain software enhancements at a rate discounted by at least 15%, with costs shared by all participants, resulting in continued savings through collaboration.
- Continues administration and payment of for Software Maintenance services through the Urban Counties.
- Structures the parties' contractual relationship in a manner that is more customary in the software development and sales marketplace through the use of a more traditional reseller agreement

Recommended Resolution:

Each participating county is asked to adopt the Amendment Number 1 to the Master Software License and Professional Services Agreement with the following understandings. Amendment No. 1:

- 1. Does not obligate the Urban Counties or participating counties to any further payments to Tyler than are currently obligated;
- 2. Provides existing participating counties a right to an \$800,000 rebate if all license fees due by existing participants are paid by December 31, 2009;
- 3. Provides for an additional \$390,000 rebate if an additional \$2,000,000 in license fees are paid by new participating counties before the end of 2010;
- Does not increase the potential costs to current participants nor affect their rights to use and have implemented what they have currently contracted for;
- Provides counties that are not currently participating the opportunity to do so thru 2014;
- 6. Increases the hourly rate paid to Tyler by 12% for change orders and new projects requested by current participants;
- 7. Continues the existing 16% of license fees maintenance fee cost for participating counties;
- 8. Allows participating counties to expand maintenance to include help desk support by increasing maintenance fees by five percent (5%); and

Fort Bend County; CIJS Contract Changes with Tyler Technologies October 23, 2009, Page 5 of 5

Allows for continuation and expansion of the existing Enterprise Modification program whereby participating counties share in the cost of enhancements to the system.

Please do not hesitate to contact me if I can be of assistance on this or any other matter. Copies of the original contract materials are available upon request.

Enclosure: Master Software License and Professional Services Agreement Amendment No. 1

Master Software License and Professional Services Agreement Amendment No. 1

This Amendment No. 1 (this "Amendment") to the Master Software License and Professional Services Agreement is entered into by and between Tyler Technologies, Inc., a Delaware corporation ("Software Provider"), and the Texas Conference of Urban Counties, Inc. (the "CUC"). This Amendment shall become effective as of the last date set forth on the signature page below (the "Effective Date").

WHEREAS, on March 14, 2006, Software Provider and the CUC entered into that certain Master Software License and Professional Services Agreement (as amended, the "Agreement") pursuant to which, among other things, Software Provider agreed to grant to the CUC a non-exclusive, royalty-free, revocable license for the sole purpose of sublicensing the Licensed Property to Participating Member Counties only; and

WHEREAS, pursuant to Section 25.7 of the Agreement, Software Provider and the CUC desire to amend the terms of the Agreement to provide for the following, among other things: (a) the license grant to the CUC for the sole purpose of sublicensing the Licensed Property shall be limited to sublicensing the Licensed Property to the thirteen Participating Member Counties (defined in this Amendment); (b) the License Fee associated with such license shall be reduced to \$10,757,500 and shall be due and payable as set forth in this Amendment; (c) the Term of the Agreement shall expire on December 31, 2009; and (d) the Participating Member Counties shall contract directly with Software Provider for maintenance and support services related to the Licensed Property, all as more particularly described herein; and

NOW, THEREFORE, in consideration of the mutual promises contained herein, along with other good and valuable consideration, the receipt and sufficiency of which both parties mutually acknowledge, the parties agree as follows:

- Defined Terms. The following definitions hereby supersede and replace the definitions contained in the Agreement:
 - a. <u>License Fee</u> means \$10,757,500 made payable in various payments pursuant to Section 4.1;
 - b. <u>Participating Member Counties</u> means: (i) Collin County, Texas; (ii) Fort Bend County, Texas; (iii) Gregg County, Texas; (iv) Tarrant County, Texas; (v) Denton County, Texas; (vi) Grayson County, Texas; (vii) Guadalupe County, Texas; (viii) Williamson County, Texas; (ix) Galveston County, Texas; (x) Hidalgo County, Texas; (xi) Nueces County, Texas; (xii) Cameron County, Texas; and (xiii) San Patricio County, Texas. No other Member County shall have the right to become a Participating Member County under the terms of the Agreement.
 - c. <u>Price List</u> means the license fees and maintenance fees associated with each Participating Member County as set forth in <u>Schedule A</u> to this Amendment.

- d. <u>Standard Maintenance Agreement</u> means Software Provider's standard maintenance and support services agreement to be executed by Software Provider and each Participating Member County. Each Participating Member County shall have the option of purchasing "Tier 1 Support" or "Tier 2 Support", as defined in the Standard Maintenance Agreement.
- e. <u>Tier 1 Support</u> means support services including (a) defect remediation; (b) help desk user support; (c) server/system support; and (d) desktop support for help desk users.
- f. <u>Tier 2 Support</u> means support services including defect remediation by Tyler after initial troubleshooting by Participating Member County technical staff.
- g. Terms used but not otherwise defined in this Amendment shall have the meaning assigned to such terms in the Agreement.
- 2. Relationship with Participating Member Counties. The Agreement is hereby amended by deleting Section 2.4(b) in its entirety and replacing it with the following:
 - "(b) Notwithstanding the foregoing, the Software Provider shall work directly with each Participating Member County, without CUC participation, for the day-to-day implementation of the Project for such Participating Member County, including, without limitation, the issuance of Authorization Orders, Authorization Confirmations, Change Orders, and Change Confirmations. In connection therewith, Software Provider shall invoice each Participating Member County directly for all fees incurred by Software Provider on behalf of such Participating Member County (including, without limitation, Local Modification Fees) in accordance with the respective Implementation Plan, which shall be paid in accordance with Section 4. Software Provider shall also provide each Participating Member County directly with all maintenance and support services under the terms of the Standard Maintenance Agreement, and the invoicing and payment of all related maintenance fees shall occur between Software Provider and each respective Participating Member County."
- 3. <u>License Fee</u>. The Agreement is hereby amended by deleting <u>Section 4.1</u> in its entirety and replacing it with the following:
 - "4.1 <u>License Fee</u>. The License Fee is \$10,652,500. As of the Effective Date of this Amendment, the parties acknowledge and agree that the CUC has paid \$_____ of the License Fee, which shall be allocated among the Participating Member Counties as set forth on <u>Schedule B</u> to this Amendment. The remaining License Fee shall be due and payable as follows:

Payment Date 12/312009 Amount to be Paid \$2,185,750 Software Provider shall invoice the CUC thirty (30) days prior to each Payment Date, which shall be paid in accordance with Section 4.4. With each License Fee payment, the CUC shall identify for Software Provider the license fee amounts attributable to each Participating Member County and to be credited against the Discounted County Price List for each Participating Member County in the event of a termination of this Agreement, which amounts shall, in the aggregate, equal the total amount of the License Fee paid by the CUC on each such Payment Date. In the event that the CUC timely pays the License Fee in full by December 31, 2009, Tyler shall remit to the CUC a one-time incentive bonus in the amount of \$800,000. Upon the CUC's timely payment of the License Fee in full, the foregoing licenses shall become irrevocable as set forth in Section 3.2 of this Agreement. If the licenses are revoked by Software Provider because of the CUC's failure to timely pay the License Fee in full, each Participating Member County may reinstate its respective sublicense using the procedure set forth in Section 3.8 of this Agreement."

4. <u>Invoice and Payment</u>. Section 4.4 of the Agreement is hereby amended by deleting the following sentence in its entirety:

"Software Provider shall invoice the CUC for maintenance fees in accordance with the terms of the Standard Maintenance Agreement."

5. <u>Implementation Services</u>. The Agreement is hereby amended by adding the following sentence to the end of Section 5.4:

"Software Provider shall continue to deliver implementation and other services as set forth in any previously approved Implementation Plan, without regard to any expiration of the Term of this Agreement provided, however, that such continuation of delivery does not relieve any Participating Member County from its obligation to pay for software licenses and services as set forth in this Agreement. Software Provider's hourly service rates for implementation and other services included in any existing approved Implementation Plan shall be as set forth for the hours in such Implementation Plan. Software Provider's hourly service rates for implementation and other services quoted in a new Budget Proposal or Change Order to an existing implementation plan hereunder shall be according to Software Provider's then current published annual rates as set forth in Schedule C and as may be amended annually by Software Provider."

- 6. <u>Maintenance and Support Agreement</u>. The Agreement is hereby amended by deleting <u>Section 11.1</u> in its entirety and replacing it with the following:
 - "11.1 Maintenance and Support Agreement. Each Participating Member County shall execute its own respective Standard Maintenance Agreement, which shall be between Software Provider and the Participating Member County. Such Standard Maintenance Agreement shall provide that the Maintenance and Support Fees shall remain fixed for a period of [3] years after the Effective Date of Participating Member County's Standard Maintenance Agreement previously executed with Software Provider and the CUC, after which time the fees may be adjusted to the level of the then current standard

Maintenance and Support Fees. At each Participating Member County's election, the Standard Maintenance Agreement shall provide either "Tier 1 Support" or "Tier 2 Support". Software Provider shall provide each respective Participating Member County with maintenance and support services for the Judicial Software in accordance with the terms of the executed Standard Maintenance Agreement, and the applicable maintenance fees shall be collected from Participating Member County as set forth below:

- (a) Software Provider shall invoice the CUC for maintenance and support fees incurred by a Participating County in accordance with the terms of the Standard Maintenance Agreement, which, in any event, shall provide that the maintenance and support fees shall be billed annually in advance. Software Provide shall use reasonable efforts to submit such invoices to the CUC sixty (60) days prior to the initial term of the Standard Maintenance Agreement or the anniversary thereof, as applicable. The CUC shall be responsible for invoicing each Participating County and collecting payments from the applicable Participating County with respect to the maintenance and support fees. The CUC shall promptly, but in any event within five (5) business days, remit to Tyler all maintenance and support fees collected on behalf of a Participating County. In the event that the CUC fails to timely perform its invoice, collection, and remittance obligations under this section, Software Provider shall have the right, in its sole discretion and upon written notice to the CUC and the affected Participating County, to invoice such Participating County directly for all future maintenance and support services.
- In order to encourage prompt and timely payment of the maintenance (b) and support fees by each Participating County, each Participation Agreement shall provide for the following: (i) all maintenance and support fees shall be due and payable by the Participating County to the CUC within thirty (30) days of its receipt of an invoice from the CUC; (ii) interest shall accrue, and the Participating County shall be liable for such interest, on all late payments of maintenance and support fees as set forth in Section 4.5; (iii) Software Provider may, in its sole discretion, suspend maintenance and support services with respect to a Participating County for all outstanding invoices that are sixty (60) days or more past due; and (iv) in the event that Software Provider suspends maintenance and support services for a Participating County, such Participating County may reinstate maintenance and support (1) if the lapsed period were less than 12 months, by paying 75% of the maintenance and support fees that would have been owed by the Participating County during the lapsed period plus the maintenance and support fees for the then upcoming maintenance year, or (2) if the lapsed period were more 12 months or more, by paying 100% of the maintenance and support fees that would have been owed by the Participating County during the lapsed period plus the maintenance and support fees for the then upcoming maintenance vear."
- 7. <u>Term.</u> The Agreement is hereby amended by deleting Section 22.1 in its entirety and replacing it with the following:
 - "22.1 <u>Term</u>. The term of this Agreement shall commence on the Effective Date and shall continue until the sooner of: (a) December 31, 2009; (b) each Participating

Member County reaches Operational Use; or (c) the Agreement is terminated as provided herein (the "Term")."

- 8. Conflict; Remaining Terms Unchanged; Entire Agreement; Further Amendment. To the extent any of the terms or provisions set forth in this Amendment conflict with the terms or provisions of the Agreement, the terms and provisions of this Amendment shall control. Except as specifically provided in this Amendment, all remaining terms and provisions of the Agreement shall remain unchanged and in full force and effect. The Agreement and this Amendment constitutes the entire understanding and contract between the Parties and supersedes any and all prior or contemporaneous oral or written representations or communications with respect to the subject matter thereof. This Amendment shall not be modified, amended, or in any way altered except by an instrument in writing signed by the properly delegated authority of each Party. All amendments or modifications to this Amendment shall be binding upon the Parties despite any lack of consideration.
- 9. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 10. Governing Law. Any dispute arising out of or relating to this Amendment or the breach thereof shall be governed by the laws of the State of Texas, without regard to or application of choice of law rules or principles.

IN WITNESS WHEREOF, this Amendment has been executed by the Parties hereto to be effective as of the last date set forth below.

TEXAS CONFERENCE OF URBAN COUNTIES, INC. TYLER TECHNOLOGIES, INC.

By:

Name:

Donald Lee

Title: Date:

October 1

/6 . 200

Ву:

Name:

itle:

e

2000-

Fort Bend County

Name: Robert Hebert

Title: County Judge Date: November 24

, 2009

Master Software License and Professional Services Agreement Amendment No. 1

Schedule "A" Price List

County	County Enterprise Pricing	Initial Case Type	Second Case Type	Support Init. Case Type	Support Second Case Type	Support Both Case Types
Tarrant (Note1)	2,000,000	1,387,500	612,500	194,250	.87,500	250,000
Hidalgo	1,050,000	682,500	367,500	126,750	68,250	195,000
Collin	1,050,000	682,500	367,500	110,500	59,500	170,000
Denton	945,000	N/A	N/A	N/A	N/A	170,000
Fort Bend	875,000	568,750	306,250	91,000	49,000	140,000
Cameron	875,000	568,750	306,250	91,000	49,000	140,000
Nueces	875,000	568,750	306,250	91,000	49,000	140,000
Galveston	875,000	568,750	306,250	84,500	45,500	130,000
Williamson	700,000	568,750	306,250	91,000	49,000	140,000
Grayson	410,000	N/A	N/A	N/A	N/A	105,000
Gregg	630,000	409,500	220,500	68,250	36,750	105,000
Guadalupe	490,000	318,500	171,500	45,500	24,500	70,000
San Patricio	490,000	318,500	171,500	45,500	24,500	70,000

Note 1. Tarrant County's "Initial Case Type" License Fee reflects an increase of \$250,000 and a proportional Annual Maintenance and Support Fee increase of \$31,750. This "Initial Case Type" license covers all case types and case manager lines of business except adult criminal.

Master Software License and Professional Services Agreement Amendment No. 1

Schedule "B" Allocation of License Fees Paid by CUC as of Effective Date

County	Enterprise License Fee	License Fee Paid as of Effective Date	Remaining License Fee Payments Due
Tarrant	\$1,387,500	\$1,137,500	\$250,000
Hidalgo	\$1,050,000	\$341,250	\$708,750
Collin	\$1,050,000	\$1,050,000	\$0
Denton	\$945,000	\$945,000	\$0
Fort Bend	\$875,000	\$875,000	. \$0
Cameron	\$875,000	\$525,000	\$350,000
Nueces	\$875,000	\$390,000	\$485,000
Galveston	\$875,000	\$875,000	0
Williamson	\$700,000	\$700,000	\$0
Grayson	\$410,000	\$410,000	\$0
Gregg	\$630,000	\$630,000	\$0
Guadalupe	\$490,000	\$294,000	\$196,000
San Patricio	\$490,000	\$294,000	\$196,000
Totals	\$10,652,500	\$7,735,250	\$2,185,750

Master Software License and Professional Services Agreement Amendment No. 1

Schedule "C" 2009 Implementation Services Hourly Rates

Project Management \$

\$160.00 per hour

Technical Services

\$149.00 per hour

Training Services

\$138.00 per hour

AMENDED AND SUPERSEDED CUC MASTER AGREEMENT

This Amended and Superseded Master Agreement is entered into by and between Tyler Technologies, Inc., a Delaware corporation, with its principal place of business located at 6500 International Parkway, Suite 2000, Plano, Texas 75093, and the Texas Conference of Urban Counties, Inc., a Texas corporation, with its principal place of business located at 500 W. 13th Street, Austin, Texas 78701.

Background

WHEREAS, Tyler and the CUC entered into that certain CUC Master Agreement dated as of September 8, 2009 (the "Original Agreement"); and

WHEREAS, the term of the Original Agreement expires as of December 31, 2014, and the parties agree to extend and amend the Original Agreement as set forth herein.

NOW, THEREFORE, in consideration of the mutual promises contained herein, along with other good and valuable consideration, the receipt and sufficiency of which all parties mutually acknowledge, the parties agree as follows:

1. **DEFINITIONS**

- 1.1 Agreement means this Amended and Superseded CUC Master Agreement, including all exhibits attached hereto and to be attached throughout the term of this Agreement, all of which are incorporated by reference herein.
- 1.2 <u>Budget Proposal</u> means the proposal developed by Tyler for a County interested in becoming an End User, which shall include an outline of the estimated Project costs, the scope of the Project, the assumptions made by Tyler associated with such costs, the County staffing expectations for the implementation of the Project, the estimated project start date and duration, and the estimated cost for completion of the Implementation Plan.
- 1.3 <u>Claims</u> mean any and all claims, liens, demands, damages, liability, actions, causes of action, losses, judgments, costs, and expenses, including reasonable attorneys' fees and expenses.
 - 1.4 <u>County</u> means, as applicable, a Member County or a Non-Member County.
- 1.5 <u>Court Admin System</u> means a common integrated court administration system, which includes (a)(i) the Current Production Software Version plus Prosecutor, or (ii) the Individual Case Types, as applicable, plus (b) Enterprise Modifications.
- 1.6 <u>CPD Document</u> means a conceptual process document, a high-level design document.
 - 1.7 <u>CUC</u> means the Texas Conference of Urban Counties, Inc.
 - 1.8 <u>CUC Technical Committee</u> has the meaning set forth in <u>Section 2.9</u>.

- 1.9 <u>Current Production Software Version</u> means the most recent production version of Tyler's court management system that is then running on Tyler's client sites as of the relevant date.
- 1.10 <u>Documentation</u> means the user's manuals and any other materials in any form or media provided by Tyler to the users of the Judicial Software for the users' respective operation of the Judicial Software in accordance with this Agreement.
 - 1.11 <u>Effective Date</u> means January 1, 2015.
- 1.12 <u>Embedded Third Party Software</u> means licensed third party software that is required to provide the functionality of the Judicial Software.
- 1.13 <u>End User</u> means a County that acquires the rights to the Products pursuant to the applicable End User License Agreement or SaaS Agreement.
- 1.14 <u>End User License Agreement</u> means Tyler's Software License and Professional Services Agreement, which is attached hereto as <u>Exhibit B-1</u>. As to any End User, Tyler and such End User may mutually agree to revisions of an End User License Agreement.
- 1.15 <u>Enterprise License Fee</u> means the license fee applicable to each County as set forth on the Enterprise Price List.
- 1.16 <u>Enterprise Modifications</u> means the list of functions to be designated by the CUC Technical Committee as an Enterprise Modification pursuant to this Agreement.
- 1.17 <u>Enterprise Price List</u> means the Enterprise License Fees set forth on <u>Exhibit A-1</u> for each County that executes an End User License Agreement prior to the end of the Term.
- 1.18 Implementation Plan means the Implementation Plan to be developed between Tyler and each End User upon such End User executing a Plan Request. The Implementation Plan shall provide for timetables, milestones, Deliverables, and fees and expenses for, among other things, (a) the development of Local Modifications for the End User, (b) the delivery and installation of the applicable Products for the End User, (c) the training of End User personnel, (d) the staffing plan for the implementation of the Project, and (e) a scheduled timetable for periodic update reporting on the status of the project by parties to the respective End User's Commissioners Court. The Implementation Plan will be incorporated into each respective End User License Agreement.
- 1.19 <u>Individual Case Type</u> means each of (a) District Clerk Civil / Family, (b) County Clerk Civil / Probate, (c) Criminal (District and County), and (d) Justice of the Peace.
- 1.20 <u>Individual Case Type License Fee</u> means the license fees associated with licensing Individual Case Types only, which shall equal the following percentages of the Enterprise Price List: (a) District Clerk Civil / Family 25% of the Enterprise License Fee; (b) County Clerk Civil / Probate 25% of the Enterprise License Fee; (c) Criminal (District and

County) -50% of the Enterprise License Fee; and (d) Justice of the Peace -30% of the Enterprise License Fee.

- 1.21 <u>Judicial Software</u> means: (a) the Court Admin System; (b) Embedded Third party Software; (c) Enterprise Modifications; (d) Local Modifications; and (e) Tyler Modifications.
- 1.22 <u>Local Modifications</u> means the list of functions designated by an End User and set forth in an Implementation Plan to be developed by Tyler and implemented into the Court Admin System for the respective End User.
- 1.23 <u>Member Counties</u> means the Texas counties that are members in good standing of the CUC.
- 1.24 <u>Non-Member County</u> means a county in the State of Texas other than the Member Counties.
- 1.25 <u>Optional Software</u> means the most recent production version of Tyler's software applications and modules (including, without limitation, Jail Manager, Law Enforcement, Check Manager, Jury, and Supervision, and certain add-on features) identified on the Optional Software Price List, and includes the functionality, modifications, and modules set forth on an Optional Software Participation Agreement, and which may be amended by Tyler from time to time.
- 1.26 <u>Optional Software End User</u> means an End User that executes an End User License Agreement or SaaS Agreement, as applicable, for Optional Software.
- 1.27 <u>Optional Software License Fee</u> means the license fee for each Optional Software application set forth on the Optional Software Price List.
- 1.28 <u>Optional Software Saas Fees</u> means the per-user monthly fees charged under a SaaS Agreement for Optional Software.
- 1.29 Optional Software Price List means the then current license fees and maintenance fees associated with each Optional Software application. A copy of the license fees and maintenance fees for the Optional Software as of the date of this Agreement is set forth on Exhibit A-3, which is attached hereto and incorporated by reference herein. The Optional Software Price List may be changed by Tyler from time to time without the approval of the CUC to add new Optional Software to the Agreement. Changes to the prices on Optional Software already identified in Exhibit A-3 will require CUC consent, not to be unreasonably withheld.
- 1.30 Participating County means a County that has executed a Notice to Proceed. For clarity, Participating Counties include all counties that use or are implementing the Products pursuant to the predecessor agreements to this Agreement as of the Effective Date and all Counties that execute a Notice to Proceed after the Effective Date.

- 1.31 <u>Plan Request</u> means written notice issued from a County to Tyler requesting that Tyler prepare an Implementation Plan at the County's expense and consistent with the Budget Proposal.
- 1.32 <u>Products</u> mean, as applicable, the Judicial Software, Individual Case Types, and Optional Software.
- 1.33 <u>Project</u> means the development, delivery, installation, testing, and license of the Products and the performance of all services to be provided by Tyler in accordance with the provisions of an End User License Agreement or SaaS Agreement, as applicable.
 - 1.34 Resale Market has the meaning set forth in Section 6.2.
- 1.35 <u>SaaS Agreement</u> means Tyler's Software as a Service and Professional Services Agreement, which is attached hereto as <u>Exhibit B-2</u>. As to any End User, Tyler and such End User may mutually agree to revisions of a SaaS Agreement.
- 1.36 <u>SaaS Fees</u> means the per-user monthly fees charged under a SaaS Agreement, as set forth on <u>Exhibit A-2</u>.
- 1.37 <u>Standard Maintenance Agreement</u> means Tyler's standard maintenance and support services agreement to be executed by Tyler and each End User, the form of which is attached hereto as <u>Exhibit B-3</u>. As to any End User, Tyler and such End User may mutually agree to revisions of a Standard Maintenance Agreement.
- 1.38 <u>Strategic Initiatives</u> means software development projects that: (a) result in new functionality jointly identified by Tyler and the CUC as having significant value and/or return-on-investment to counties; (b) can be packaged or bundled as discrete software options; and (c) the parties can reasonably determine that a market exists for the resale of such functionality sufficient to provide a return on investment to Tyler and the CUC consistent with industry standards.
 - 1.39 <u>Targeted Sales Activity</u> has the meaning set forth in <u>Section 2.3</u>.
 - 1.40 <u>Territory</u> means the State of Texas.
- 1.41 <u>Tier 1 Support</u> means support services including: (a) defect remediation; (b) help desk user support; (c) server/system support; and (d) desktop support for help desk users.
 - 1.42 <u>Tyler</u> means Tyler Technologies, Inc., a Delaware corporation.
- 1.43 <u>Tyler Confidential Information</u> means all information in any form relating to, used in, or arising out of the Tyler's operations and held by, owned, licensed, or otherwise possessed by Tyler (whether held by, owned, licensed, possessed or otherwise existing in, on, or about Tyler's premises or the CUC's offices, residence(s), or facilities and regardless of how such information came into being, as well as regardless of who created, generated or gathered the information), including, without limitation, all information contained in, embodied in (in any media whatsoever) or relating to Tyler's inventions, ideas, creations,

works of authorship, works of visual art, business documents, operations, manuals, performance manuals, operating data, projections, bulletins, supplier and customer lists, strategic planning data, financial planning data, designs, logos, motifs, test results, product or service literature, product or service concepts, process data, specification data, know how, show how, software, databases, release notes, database layouts, algorithms, techniques, processes, and systems designs. Tyler Confidential Information does not include information that (a) becomes public other than as a result of a disclosure by the CUC in breach hereof, (b) becomes available to the CUC on a non-confidential basis from a source other than Tyler, which is not prohibited from disclosing such information to the CUC by obligation to Tyler, (c) is known by the CUC prior to its receipt from Tyler without any obligation of confidentiality with respect thereto, or (d) is developed by the CUC independently of any disclosures made by Tyler to the CUC of such information.

- 1.44 <u>Tyler Modifications</u> means enhancements and upgrades to the Judicial Software developed by Tyler after the Effective Date.
 - 1.45 <u>Tyler Trademarks</u> has the meaning set forth in <u>Section 7.3</u>.
 - 1.46 <u>Term</u> has the meaning set forth in <u>Section 13.1</u>.
 - 1.47 <u>Verification Procedure(s)</u> has the meaning set forth in <u>Section 4.1</u>.

2. CUC RESPONSIBILITIES

- 2.1 <u>Assistance and Cooperation</u>. After identifying potential End Users, the CUC shall reasonably assist Tyler in coordinating meetings and other activities leading up to proposal requests.
- 2.2 <u>Products</u>. The CUC agrees to provide contract administration and oversight for Counties in accordance with the terms of this Agreement and as follows:
 - (a) with respect to the Judicial Software, to Counties that execute a Plan Request in consideration for the Enterprise License Fee or SaaS Fees, as applicable;
 - (b) with respect to Individual Case Types, to Counties that execute a Plan Request in consideration for the Individual Case Type License Fees or SaaS Fees, as applicable; and
 - (c) with respect to Optional Software, to Counties that execute a Plan Request in consideration for the Optional Software License Fee or Optional Software SaaS Fees, as applicable.
- 2.3 <u>Preserving Tyler Goodwill</u>. The CUC shall conduct its business so as to maintain the goodwill and reputation of Tyler and the Products. The CUC will not initiate Targeted Sales Activity for alternative Court Administration System(s) that provide similar functionality as the Judicial Software to Participating Counties. In this section, "<u>Targeted Sales Activity</u>" means any CUC-initiated communication, whether direct or indirect, that is intended specifically for one or more Participating Counties with the objective of convincing or

encouraging a Participating County to use an alternative Court Administration System. The CUC is not prohibited in any way from making announcements to, or discussing with, any person or entity the activities of CUC, whether past, present, or proposed. The CUC is not prohibited from responding to direct inquiries initiated by Participating Counties regarding alternative Court Administration System(s).

- 2.4 End User License Agreements. The CUC shall ensure that all End User License Agreements are properly executed by End Users, shall collect all license fees associated with such End User License Agreements, and shall promptly deliver to Tyler all such license fees in accordance with the terms of such End User License Agreements, which shall include the following payment provisions:
 - (a) sixty percent (60%) of the applicable license fee shall be due and payable by the End User upon the End User executing an End User License Agreement; and
 - (b) the remaining forty percent (40%) of the applicable license fee shall be due and payable by the End User on the date or schedule specified in the End User License Agreement.
- 2.5 <u>SaaS Agreements</u>. The CUC shall ensure that all SaaS Agreements are properly executed by End Users, shall collect the SaaS Fees as set forth therein, and shall promptly deliver to Tyler all such SaaS Fees in accordance with the terms of such SaaS Agreements.
- 2.6 <u>Maintenance and Support</u>. The CUC shall have no obligation with respect to maintenance and support services for the Products, which services shall be governed by the terms and conditions of Tyler's Standard Maintenance Agreement to be executed between Tyler and the End User.
- 2.7 <u>Compliance</u>. In performing its duties hereunder, the CUC shall comply with all federal, state, and local laws, regulations, and ordinances.
- 2.8 <u>Expenses</u>. The CUC shall be responsible for all expenses incurred by the CUC in the performance of its duties hereunder.
- 2.9 <u>CUC Technical Committee</u>. The CUC shall designate in writing a technical committee (the "<u>CUC Technical Committee</u>"), which shall serve as an intermediary and services aggregator for End Users and perform such other functions as described in <u>Section 4</u>. The members of the CUC Technical Committee may change from time to time at the discretion of each Participating County.
- 2.10 <u>Negative Covenants</u>. Notwithstanding anything herein to the contrary, the CUC shall not, without the prior written approval of Tyler:
 - (a) offer discounts for Tyler products or services other than those set forth in this Agreement without Tyler's prior written consent;

- (b) waive, alter, or change any provision of an End User License Agreement, SaaS Agreement, or Standard Maintenance Agreement;
- (c) make any representation or warranty about the Products other than as presented in Tyler's written materials;
- (d) modify or extend the amount of or time for the payment of any charge or fee under any End User License Agreement, SaaS Agreement, or Standard Maintenance Agreement; or
- (e) incur any expense or obligation in the name of Tyler, other than as provided in this Agreement.
- 2.11 Other Responsibilities. In order to work collaboratively with Tyler to improve the use and operation of the Products in the Participating Counties and other jurisdictions, encourage the adoption and use of the Products in the Counties, and facilitate the activities of the Participating Counties, the CUC shall:
 - (a) encourage the Participating Counties to fund two (2) full time Tyler positions to work exclusively with the CUC program to enhance the implementation and use of the Products;
 - (b) develop marketing materials such as brochures, presentations and, potentially, video presentations to promote the CUC program;
 - (c) provide two full-time dedicated positions at the CUC to work exclusively in the program to enhance the implementation and use of the Products and coordinate activities with the Participating Counties;
 - (d) work with Tyler to complete an assessment and a plan for the Participating Counties to help define how each County will enhance and improve the use of the Products, as well as projects and activities the Counties can work on together through the CUC program;
 - (e) update the assessment and plan on an annual basis to further the efforts to improve the implementation and use of the Products in the Participating Counties; and
 - (f) jointly sponsor, with Tyler, an annual executive leadership meeting to foster a better understanding of the future of criminal justice systems among key elected officials and leaders among the Participating Counties.

3. TYLER RESPONSIBILITIES

3.1 <u>Budget Proposals</u>. Upon the CUC identifying a County to Tyler as a potential End User, Tyler shall meet with the County and the CUC, at no cost, to prepare and issue a Budget Proposal, which shall occur within sixty (60) days of the date Tyler is notified by the CUC. Tyler shall negotiate in good faith any such Budget Proposal with the applicable County.

A Budget Proposal shall be in effect for ninety (90) days unless otherwise agreed to in writing by Tyler, at which time it shall expire unless the County either: (a) issues to Tyler a Plan Request and executes an End User License Agreement; or (b) executes a SaaS Agreement. Any End User License Agreement executed under this <u>Section 3.1</u> shall authorize Tyler to develop an Implementation Plan as set forth in <u>Section 3.2</u>, after which the applicable County will issue either a Notice to Proceed or a Termination Notice.

- 3.2 Plan Requests. Within ninety (90) days of its receipt of a Plan Request and executed End User License Agreement, Tyler shall, at the County's expense, develop an Implementation Plan. The terms of the Implementation Plan shall be in effect for ninety (90) days unless otherwise agreed to in writing by Tyler, at which time it shall expire unless the County executes a Notice to Proceed incorporating the Implementation Plan. If the Implementation Plan expires, the County shall pay Tyler the costs and expenses incurred by Tyler in preparing the Implementation Plan, and Tyler shall refund to the County all license fees previously paid by the County for the Products contemplated in the Implementation Plan, if any. The requirement for a Plan Request, Implementation Plan, and Notice to Proceed, shall not apply to End User License Agreements or SaaS Agreements for acquisition of additional Optional Software subsequent to a Participating County's original Notice to Proceed. Such subsequent End User License Agreements or SaaS Agreements shall be effective upon execution by the Participating County and Tyler, and no right to refund (as set forth in this Section 3.2) shall exist unless specifically set forth in such subsequent End User License Agreement or SaaS Agreement.
- 3.3 <u>Development and Delivery of Products</u>. Tyler shall use commercially reasonable efforts to (i) develop, deliver, install, verify, and test the Products for the End User, (ii) provide Documentation and other materials to the End User, and (iii) perform such modifications, conversion, training, and other services to the End User, all in accordance with the terms and conditions of the applicable End User License Agreement.
- 3.4 <u>Maintenance and Support</u>. For each End User License Agreement, Tyler shall be responsible for obtaining from each End User an executed Standard Maintenance Agreement, which shall provide each End User with Tier 1 Support, the pricing of which shall be set forth in the Budget Proposal. Tyler shall use commercially reasonable efforts to provide maintenance and support services to the End User pursuant to the terms of the Standard Maintenance Agreement, and the CUC shall be responsible for collecting all maintenance and support fees pursuant thereto in accordance with <u>Section 5.3</u>. Tyler will provide a minimum of twelve (12) months written notice to the CUC before Tyler stops providing maintenance and support for a specific version and/or release of a Product. No separate Standard Maintenance Agreement shall be required for a SaaS Agreement.
- 3.5 <u>Compliance</u>. Tyler shall comply with all federal, state, and local laws, regulations, and ordinances.
- 3.6 <u>Service Rates</u>. Tyler shall publish and provide to the CUC and End Users an annual price list (attached as the then current "<u>Exhibit A-4</u>") for its hourly time and materials service rates, not later than March 31 of each year during the Term, which shall be in effect for the next calendar year. Tyler may not increase Service Rates by more than 5% per year.

Service Rates may not be changed for services previously contracted for by a County pursuant to a County Implementation Plan for which a Notice to Proceed been issued.

- 3.7 <u>CUC Trade Name</u>. Notwithstanding anything herein to the contrary, Tyler shall not use any name, trademark, or trade name of the CUC, directly or indirectly, whether in connection with advertising or otherwise, without the prior signed, written consent of the CUC as to each circumstance and occasion of use. Tyler shall not claim that the CUC endorses its products or services without the prior signed, written consent of the CUC. For purposes of this provision, publicity includes notices, informational pamphlets, press releases, research, reports, signs, and similar public notices prepared by or for Tyler individually or jointly with others, or any subcontractors, with respect to this agreement or any services or Deliverables provided herein.
- 3.8 <u>Other Responsibilities</u>. In order to work collaboratively with CUC to improve the use and operation of the Products in the participating Counties and other jurisdictions, encourage the adoption and use of the Products in the Counties, and facilitate the activities of the participating Counties, Tyler shall:
 - (a) provide two (2) full time Tyler positions to work exclusively with the CUC program to enhance the implementation and use of the Products, contingent on the Participating Counties providing funding;
 - (b) provide the opportunity for the CUC to participate in the selection of specific individuals to work in the full-time positions assigned to the CUC program;
 - (c) assist in the development of marketing materials such as brochures, presentations and, potentially, video presentations to promote the CUC program and the Tyler Products;
 - (d) work with CUC to complete an assessment and a plan for the Participating Counties to help define how each County will enhance and improve the use of the Products, as well as projects and activities the Counties can work on together through the CUC program;
 - (e) update the assessment and plan on an annual basis to further the efforts to improve the implementation and use of the Products in the Participating Counties;
 - (f) reasonably assist the CUC in its promotion of the CUC program as an option for licensing, implementing and supporting the Products in all Texas Counties, and as the preferred approach for obtaining the Products in Member Counties and counties eligible to become Member Counties; and
 - (g) jointly sponsor, with the CUC, an annual executive leadership meeting to foster a better understanding of the future of Common Integrated Justice Systems among key elected officials and leaders among the Participating Counties.

3.9 <u>Pricing.</u> During the Term of this Agreement, the pricing set forth in <u>Exhibits A-1</u> through <u>A-3</u> for Tyler license fees and Saas Fees to End Users, as may be amended from time to time, shall be more favorable than Tyler's standard list price. Tyler may, in its sole discretion, propose or provide additional discounts and incentives to any prospective customer through such prospective customer's independent procurement process. If such prospective customer becomes a Participating County instead of contracting independently with Tyler, it shall be eligible for the most favorable pricing from either: (a) the pricing set forth in this Agreement; or (b) the lowest pricing proposed by Tyler during the prospective customer's independent procurement process, including any discounts or incentives previously negotiated with Tyler.

4. ADDITIONAL RESPONSIBILITIES AND COVENANTS

- 4.1 <u>CUC as End User Services Aggregator for Enterprise Modifications</u>. The CUC, as a representative of Participating Counties, shall manage the relationship of End Users and shall represent the End Users collectively for the purpose of identifying and requesting the development of Enterprise Modifications on behalf of the End Users in consideration for the payment of the Enterprise Modification fee, as more particularly set forth in this <u>Section 4.1</u>.
 - (a) After the execution of this Agreement, Tyler agrees to work in good faith with the CUC Technical Committee for the purposes of collecting and analyzing sufficient data to further refine and define in detail additional Enterprise Modifications. After the parties collect and analyze sufficient data to adequately identify and define in detail the scope of any Enterprise Modification, Tyler shall then discuss with the CUC Technical Committee the specifications, conduct a detailed analysis with respect to the Enterprise Modifications, and present to the CUC Technical Committee a detailed written report with respect to the findings of such analysis. For each Enterprise Modification authorized by the CUC Technical Committee pursuant to a signed writing, Tyler shall prepare a CPD Document, at the CUC's expense, and deliver it to the CUC Technical Committee for inspection and approval. Upon approval by the CUC Technical Committee, Tyler shall develop and implement the Enterprise Modifications into the Court Admin System on an hourly basis in accordance with the timetable set forth in the applicable CPD Document. In no event shall Tyler be responsible to the CUC for delays attributable to the CUC's failure to timely inspect and approve a CPD Document. Tyler shall perform its functions hereunder, and shall invoice the CUC, on an hourly basis.
 - (b) Upon delivery, installation, and diagnostic testing of Enterprise Modifications to the CUC Technical Committee, and regardless of whether or not the CUC supplies any test scripts, Tyler shall perform its standard test procedures set forth in the applicable CPD Document and shall certify to the CUC Technical Committee that the Enterprise Modifications in each applicable phase is operating in accordance with the specifications and shall demonstrate such successful operation to the CUC Technical Committee (the "Verification Procedure"). Any defect to an Enterprise Modification that is revealed during the Verification Procedure shall be promptly corrected by Tyler at Tyler's expense, and appropriate documentation for such correction shall be produced and delivered to the CUC Technical Committee within thirty (30) days of such correction. The CUC Technical Committee, in its sole

10

and absolute discretion, may monitor the Verification Procedure. As part of the Verification Procedure, the CUC Technical Committee may, but is not required to, submit to Tyler functional test scripts for each Enterprise Modification, which test scripts will be consistent with the CPD document and will be used by the CUC Technical Committee for purposes of verification testing of the Enterprise Modification. Tyler shall provide to the CUC Technical Committee for its internal use a test script sample set containing test scripts that the CUC Technical Committee may use as examples for the development of its test scripts. The parties must agree on the specifics of any supplied test scripts in order for the test scripts to become a part of the Verification Procedures. The parties will promptly, but in any event not less than five business days, meet in good faith to resolve any issues or disagreements associated with each test script supplied by the CUC Technical Committee. The CUC Technical Committee or End User supplied test scripts, if any, will be in addition to Tyler's Verification Procedures. Upon completion of the Verification Procedure, the Enterprise Modifications shall be deemed part of the Judicial Software and subject to the End User License Agreements.

(c) Enterprise Modifications shall be performed by Tyler at its then current published rates; provided, however, that the CUC shall have the option to purchase, in advance, and on behalf of the End Users 1,000-hour individual blocks of service hours at a discount of 15% off the then current published rates. In no event, however, shall the CUC purchase more than five (5) discounted 1,000-hour blocks in any given calendar year. The CUC may exercise this option by providing Tyler with written notice of its intent to do so, along with an appropriate purchase requisition.

4.2 <u>Multi-Jurisdictional Software</u>. [Intentionally deleted]

5. PAYMENT OBLIGATIONS

- 5.1 <u>License Fees</u>. The CUC shall invoice and timely collect and promptly pay to Tyler all applicable license fees from End Users under the terms of the applicable End User License Agreement. Tyler shall notify the CUC upon the achievement of the applicable billing dates defined in the End User License Agreement, and shall invoice the CUC for the applicable license fees when due.
- 5.2 <u>Tyler Service Charges</u>. Tyler shall invoice and collect directly from End Users all hourly services charges incurred by each End User and expenses incurred by Tyler on behalf of such End User, all in accordance with the End User License Agreement or SaaS Agreement, as applicable, executed between Tyler and the applicable End User. Tyler shall invoice and collect directly from the CUC all hourly services charges incurred in the development of Enterprise Modifications.

5.3 <u>Maintenance and Support Service Charges.</u>

(a) Tyler shall invoice the CUC for maintenance and support fees incurred by a Participating County in accordance with the terms of the Standard Maintenance Agreement, which, in any event, shall provide that the maintenance and support fees shall be billed annually in advance. Tyler shall use reasonable efforts to submit such invoices to the CUC sixty (60) days prior to the initial term of the Standard Maintenance Agreement or the anniversary thereof, as applicable. The CUC shall be

responsible for invoicing each Participating County and collecting payments from the applicable Participating County with respect to the maintenance and support fees. The CUC shall promptly, but in any event within thirty (30) days of receipt of maintenance and support fees, remit to Tyler all maintenance and support fees collected on behalf of a Participating County. In the event that the CUC fails to timely perform its invoice, collection, and remittance obligations under this section, Tyler shall have the right, in its sole discretion and upon written notice to the CUC and the affected Participating County, to invoice such Participating County directly for all future maintenance and support services.

(b) In order to encourage prompt and timely payment of the maintenance and support fees by each Participating County, each Standard Maintenance Agreement shall provide for the following: (i) all maintenance and support fees shall be due and payable by the Participating County to the CUC in accordance with Chapter 2251, Texas Government Code; (ii) Tyler may, in its sole discretion, suspend maintenance and support services with respect to a Participating County for all outstanding invoices that are sixty (60) days or more past due; and (iii) in the event that Software Provider suspends maintenance and support services for a Participating County, such Participating County may reinstate maintenance and support by paying 100% of the maintenance and support fees that would have been owed by the Participating County during the lapsed period plus the maintenance and support fees for the then upcoming maintenance year.

5.4 SaaS Fees.

- (a) Tyler shall invoice the CUC for SaaS Fees incurred by an End User in accordance with the terms of the SaaS Agreement, which, in any event, shall provide that the SaaS Fees shall be billed annually in advance. Tyler shall use reasonable efforts to submit such invoices to the CUC sixty (60) days prior to the initial term of the SaaS Agreement or the anniversary thereof, as applicable. The CUC shall be responsible for invoicing each End User and collecting payments from the applicable End User with respect to the SaaS Fees. The CUC shall remit to Tyler all SaaS Fees collected on behalf of an End User, less a six percent (6%) administrative fee, which shall be retained by the CUC. The CUC shall make such remittances to Tyler promptly, but in no event later than thirty (30) days after receipt by the CUC of SaaS Fees from an End User. In the event that the CUC fails to timely perform its invoice, collection, and remittance obligations under this section, Tyler shall have the right, in its sole discretion and upon written notice to the CUC and the affected End User, to invoice such End User directly for all future SaaS Fees.
- (b) In order to encourage prompt and timely payment of the SaaS Fees by each End User, each SaaS Agreement shall provide for the following: (i) all SaaS Fees shall be due and payable by the End User to the CUC in accordance with chapter 2251, Texas Government Code; and (ii) Tyler may, in its sole discretion, suspend services or take other actions set forth in the SaaS Agreement with respect to an End User for all outstanding invoices that are sixty (60) days or more past due.

- 5.5 <u>Dedicated Tyler Staff.</u> If the participating counties approve and fund the Tyler positions contemplated in sections 2.11(a) and 3.8(a), then CUC will pay Tyler monthly for such positions. Monthly payments will be 1/12th of \$325,000, or \$27,083.33 beginning in January 2015. Notwithstanding the preceding sentence, payments to Tyler for these positions will not exceed amounts received by CUC from Participating Counties and approved by the Stakeholder Committee for this purpose. Further, this obligation on the part of CUC is contingent upon continued participation in the TechShare CIJS Program of no fewer than eleven (11) counties, as determined by the number of counties that have approved the TechShare Common Integrated Justice System 2015 Resource Sharing Addendum on or before December 31, 2014.
- 5.6 <u>Invoice and Payment</u>. Each invoice shall state the total invoiced amount and shall be accompanied by a reasonably detailed itemization of services and expenses. Any undisputed sum not paid when due shall bear interest calculated on an annual basis pursuant to Chapter 2251 of the Texas Government Code.

6. Strategic Initiatives.

- 6.1 <u>CUC as End User Services Aggregator for Strategic Initiatives</u>. The CUC, as a representative of End Users, shall manage the relationship of End Users and shall have the authority to represent the End Users collectively for the purpose of identifying and requesting the development of Strategic Initiatives on behalf of the End Users, as set forth below:
 - (a) At least twice per year, Tyler, the CUC, and the CUC CIJS Technical Committee (or such other representatives as may be designated by the CUC Oversight Committee) shall meet jointly to identify potential Strategic Initiatives.
 - (b) Potential Strategic Initiatives may be identified for any of the modules and/or software components available through this Agreement.
 - (c) End Users, Tyler, or the CUC may suggest that modifications that have already been completed either as Enterprise Modifications or Local Modifications be considered as Strategic Initiatives.
 - (d) For each Strategic Initiative authorized by the CUC pursuant to a signed writing, Tyler shall prepare a CPD document, at the CUC's expense, and deliver it to the CUC for inspection and approval. Upon approval by the CUC, Tyler shall develop and implement the Strategic Initiative on a T&M basis at Tyler's then current hourly rates, subject to the credits and discounts set forth in (e), in accordance with the applicable CPD document. Tyler shall perform its functions hereunder, and shall invoice the CUC, on an hourly basis.
 - (e) Upon approval of each Strategic Initiative as set forth in (b), Tyler shall issue a credit to the CUC for 50% of the cost of developing the CPD. Additional development or other services provided by Tyler to implement the Strategic Initiative shall be invoiced at a discount equal to 50% of Tyler's then current hourly rate for such services as set forth in the Agreement.
 - (f) Upon completion of the Verification Procedure: (i) Tyler shall add the applicable Strategic Initiative to the Optional Software Price List; (ii) the Strategic

Initiative shall be deemed Optional Software; and (iii) the CUC shall designate in writing to Tyler a list of End Users who may license the applicable Strategic Initiative without an Optional Software License Fee.

- (g) On an annual basis, the CUC may update the list(s) of End User(s) who may license any or all applicable Strategic Initiative(s) without an Optional Software License Fee so that newly enrolled and existing End User(s) are encouraged to continue contributing toward the expansion of the Enterprise Modification program.
- (h) Nothing in the Agreement, or any End User License Agreement or SaaS Agreement, shall be deemed to vest in the CUC or any End User any ownership or intellectual property rights in and to the Strategic Initiatives, Tyler's intellectual property, any components and copies thereof, or any derivative works based thereon prepared by Tyler.
- 6.2 <u>Marketing costs</u>. The CUC shall have no liability for the costs to market and sell the Strategic Initiatives to counties, states, and other jurisdictions who are not End Users actively participating in the CUC TechShare program (collectively, the "Resale Market"). The CUC shall reasonably cooperate with Tyler's efforts to market the Strategic Initiatives including, without limitation, providing references, assisting with the development of case studies, and by reasonably participating, at Tyler's expense, in industry events that showcase the Strategic Initiatives.
- 6.3 <u>Further development and maintenance costs.</u> The CUC shall have no liability for subsequent enhancement and maintenance of the Strategic Initiatives for the Resale Market. Such development and maintenance shall be at Tyler's expense.
- 6.4 <u>Revenue Share</u>. Tyler may market, sell, and license the Strategic Initiatives to counties, states, and other jurisdictions who are not End Users actively participating in the CUC TechShare program. For a period of four years after the completion of each such Strategic Initiative, Tyler will annually pay to the CUC a royalty for such Strategic Initiatives sold by Tyler as follows:
 - (a) 25% of the license fees sold in the first year; and
 - (b) 25% of the license fees sold in the second year; and
 - (c) 25% of the license fees sold in the third year; and
 - (d) 25% of the license fees sold in the fourth year.

7. PROPERTY RIGHTS

7.1 <u>Title</u>. Tyler represents and warrants that it is the owner of all right, title, and interest in and to the Products (other than Embedded Third Party Software or other third person software) and all components and copies thereof. Nothing in this Agreement or any End User License Agreement or SaaS Agreement shall be deemed to vest in the CUC or any End User any ownership or intellectual property rights in and to the Tyler's intellectual property, any components and copies thereof, or any derivative works based thereon

prepared by Tyler.

- 7.2 <u>No Other Rights</u>. The CUC may not, directly or indirectly, in any form or manner, copy, distribute, reproduce, incorporate, use or allow access to, the Products, or modify, prepare derivative works of, decompile, reverse engineer, disassemble, or otherwise attempt to derive source code or object code from the Products, except as explicitly permitted under this Agreement or otherwise agreed in writing.
- 7.3 <u>Use of Trademarks and Trade Names</u>. During the Term of this Agreement, Tyler grants the CUC a non-exclusive and non-transferable license to the trademarks, marks, and trade names that Tyler may adopt from time to time solely in connection with the marketing, advertising, and distribution of the Products (collectively, the "<u>Tyler Trademarks</u>"). Nothing herein shall grant the CUC any right, title, or interest in the Tyler Trademarks.

8. CONFIDENTIAL INFORMATION

- 8.1 Non-Disclosure. The CUC will not disclose, disseminate, transmit, publish, distribute, make available, or otherwise convey Tyler Confidential Information, and the CUC will not use, make, sell, or otherwise exploit any such Tyler Confidential Information for any purpose other than performance of this Agreement, without Tyler's written consent except: (i) as may be required by law, regulation, judicial or administrative process; or (ii) as required in litigation pertaining to this Agreement, provided Tyler is given advance notice of such intended disclosure in order to permit Tyler the opportunity to seek a protective order. The CUC will ensure that all of its respective employees shall abide by the terms of this Section 8.1 and will be responsible for breaches by such persons.
- 8.2 <u>Legal Proceedings</u>. If the CUC is requested or required (by oral questions, interrogatories, requests for information or documents in legal proceedings, subpoena, civil investigative demand or other similar process) to disclose any Tyler Confidential Information, then the CUC shall provide Tyler with prompt written notice of such request or requirement so that Tyler may seek protective orders or other appropriate remedies and/or waive compliance with the provisions of this Agreement. If, in the absence of a protective order or other remedy or the receipt of a waiver by Tyler, the CUC nonetheless is legally compelled to disclose Tyler Confidential Information to any court or tribunal or else would stand liable for contempt or suffer other censure or penalty, the CUC may, without liability herein, disclose to such court or tribunal only that portion of the Tyler Confidential Information which the court requires the CUC to disclose, provided that the CUC, at Tyler's expense, use reasonable efforts to preserve the confidentiality of the Tyler Confidential Information, including, without limitation, by cooperating with Tyler to obtain an appropriate protective order or other reliable assurance that confidential treatment will be accorded the Confidential Information by such court or tribunal.

9. REPRESENTATIONS AND WARRANTIES

- 9.1 <u>Standard Warranty to End Users</u>. Tyler shall provide to each End User the representations and warranties set forth in the End User License Agreement or SaaS Agreement, as applicable.
- 9.2 <u>NO OTHER WARRANTIES</u>. EXCEPT AS SPECIFICALLY SET FORTH IN THE APPLICABLE END USER LICENSE AGREEMENT OR SAAS AGREEMENT, TYLER DISCLAIMS ALL

OTHER WARRANTIES, INCLUDING, WITHOUT LIMITATION, IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE.

10. LIMITATION OF LIABILITY

THE LIABILITY OF TYLER FOR DAMAGES ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT, WHETHER BASED ON A THEORY OF CONTRACT OR TORT, INCLUDING NEGLIGENCE AND STRICT LIABILITY, SHALL BE LIMITED TO THE ENTERPRISE MODIFICATION FEE. THE LIABILITY OF TYLER FOR DAMAGES ARISING OUT OF OR IN CONNECTION WITH AN END USER LICENSE AGREEMENT OR SAAS AGREEMENT OR STANDARD MAINTENANCE AGREEMENT SHALL BE GOVERNED BY THE TERMS OF SUCH APPLICABLE AGREEMENT. IN NO EVENT SHALL THE CUC HAVE THE RIGHT TO ENFORCE ANY PROVISION OF AN END USER LICENSE AGREEMENT OR SAAS AGREEMENT OR STANDARD MAINTENANCE AGREEMENT, UNLESS THE CUC SHALL BE A PARTY THERETO.

IN NO EVENT SHALL TYLER OR THE CUC BE LIABLE (NOR TO ANY PERSON CLAIMING ANY RIGHT, TITLE, OR INTEREST DERIVED FROM OR AS SUCCESSOR TO THE CUC OR PARTICIPATING COUNTY'S RIGHT, TITLE, AND INTEREST) FOR INCIDENTAL, CONSEQUENTIAL, OR SPECIAL DAMAGES OF ANY KIND, INCLUDING, WITHOUT LIMITATION, LOST REVENUES OR PROFITS, OR LOSS OF BUSINESS OR LOSS OF DATA ARISING OUT OF THIS AGREEMENT, IRRESPECTIVE OF WHETHER THE PARTIES HAVE ADVANCE NOTICE OF THE POSSIBILITY OF SUCH DAMAGE.

THE FOREGOING LIMITATIONS OF LIABILITY SET FORTH IN THIS SECTION 10 SHALL NOT APPLY WITH RESPECT TO: (A) DAMAGES OCCASIONED BY THE FRAUD, GROSS NEGLIGENCE, OR WILLFUL MISCONDUCTOF EITHER PARTY; (B) CLAIMS THAT ARE THE SUBJECT OF INDEMNIFICATION UNDER SECTION 11 OF THIS AGREEMENT; OR (C) DAMAGES OCCASIONED BY VIOLATION OF LAW.

11. INDEMNIFICATION

- 11.1 <u>Breach of End User License Agreement, SaaS Agreement, or Standard Maintenance Agreement</u>. Tyler shall defend, indemnify, hold, and save harmless the CUC and its employees, agents, successors, and assigns from and against any and all Claims by an End User for breach by Tyler of the applicable End User License Agreement, SaaS Agreement, or Standard Maintenance Agreement.
- 11.2 <u>General Bodily Injury and Property Damage</u>. Tyler shall defend, indemnify, hold, and save harmless the CUC and its employees, agents, successors, and assigns from and against any and all Claims arising from Tyler's participation in this Agreement for bodily injury or property damage sustained by or asserted against the CUC arising out of, resulting from, or attributable to the negligent or willful misconduct of Tyler, its employees, subcontractors, consultants, representatives, and agents; provided, however, that Tyler shall not be liable herein to indemnify the CUC against liability for damages arising out of bodily injury to people or damage to property to the extent that such bodily injury or property damage is caused by or resulting from the actions, negligent or otherwise, of the CUC or any other third party.
- 11.3 <u>Intellectual Property</u>. If any claim is asserted, or action or proceeding brought against the CUC that alleges that all or any part of the Products, in the form supplied, or

modified by Tyler, or its subcontractors pursuant to Tyler's sole directions, infringes or misappropriates any United States intellectual property, intangible asset, or other proprietary right, title or interest (including, without limitation, any copyright or patent or any trade secret right, title, or interest), or violates any other contract, license, grant, or other proprietary right of any third party, the CUC, upon its awareness, shall give Tyler prompt written notice thereof. Tyler shall defend, and hold the CUC harmless against any such claim or action with counsel of Tyler's choice and at Tyler's expense and shall indemnify the CUC against any liability, damages and costs resulting from such claim. Without waiving any rights pursuant to sovereign immunity, the CUC shall cooperate with Tyler in the defense of any claim, action or proceeding and shall, if appropriate, make employees available as Tyler may reasonably request with regard to such defense. This indemnity does not apply to the extent that such a claim is attributable to modifications to the Judicial Software made by the CUC or any third party pursuant to the CUC's directions, or upon the unauthorized use of the Judicial Software by the CUC.

12. INSURANCE

Tyler covenants that it shall provide and maintain the following insurance:

- 12.1 <u>Commercial General Liability Insurance</u>. Commercial liability insurance, including bodily injury, property damage, and products/completed operations coverage in the amount of not less than five million dollars (\$5,000,000) each occurrence or aggregate where applicable.
- 12.2 <u>Property Insurance</u>. Property insurance shall be secured in amounts appropriate to adequately cover the business and personal property of Tyler, with no risk to any co-insurance penalty.
- 12.3 <u>Workers' Compensation and Employer's Liability Insurance</u>. Workers' compensation coverage providing statutory benefits and employer's liability insurance with minimum limits of one million dollars (\$1,000,000) each accident/each employee covering all employees.
- 12.4 <u>Additional Insured Endorsement</u>. Any liability policy required hereunder shall contain an endorsement which applies such policy's coverage to CUC, and the officers, agents, employees and volunteers of CUC, individually and collectively as additional insureds. All liability insurance policies shall name the CUC as an additional insured. Furthermore, the Workers Compensation and Liability Insurance carriers shall grant a waiver of subrogation in the CUC's favor.

13. TERM, SUSPENSION, AND TERMINATION

13.1 <u>Term</u>. The term of this Agreement shall commence on the Effective Date and shall continue until the sooner of: (a) December 31, 2015; or (b) the Agreement is terminated as provided herein (the "<u>Term</u>"). The parties will work cooperatively to achieve their desire to extend this Agreement on mutually agreeable terms. Such extensions, if any, will follow a contract review and extension process to be agreed to by the parties. The parties agree to

meet and negotiate in good faith any future extensions at least ninety (90) days prior to the end of the Term.

- 13.2 <u>Termination for Cause</u>. Either party may terminate this Agreement for Cause, provided that such party follows the procedures set forth in this <u>Section 13.2</u>.
 - (a) For purposes of this Section, "Cause" means either: (i) a material breach of this Agreement, which has not been cured within thirty (30) days of the date such party receives written notice of such breach; (ii) breach of Section 8; (iii) breach by the CUC of Section 2.3, as determined by Tyler in its sole discretion, but subject to the procedures in (a-1) immediately below; or (iv) if Tyler becomes insolvent or bankrupt, or is the subject of any proceedings relating to its liquidation or insolvency or for the appointment of a receiver or similar officer for it, has a receiver of its assets or property appointed or makes an assignment for the benefit of all or substantially all of its creditors, or institutes or causes to be instituted any proceeding in bankruptcy or reorganization or rearrangement of its affairs.
 - (a-1) If Tyler believes that the CUC has breached Section 2.3, it shall notify the CUC in writing of such breach, providing reasonable specificity as to inform the CUC of the actions constituting the alleged breach. Within 30 days of receipt of such notice, the CUC shall convene a meeting of the Stakeholders during which Tyler and the CUC will discuss the alleged breach. Tyler may terminate this Agreement at the conclusion of that meeting if it determines a breach of Section 2.3 has occurred, in Tyler's sole discretion.
 - (b) No party may terminate this Agreement until it notifies the other party in writing of the existence of such material breach, provides the alleged breaching party with thirty (30) days to cure such alleged breach, and cooperates with the alleged breaching party during such thirty (30) day period on a good faith basis to cure such alleged breach. The cooperation procedures set forth in this Section 13.2(b) do not apply for a termination for Cause as defined in Section 13.2(a)(ii) or (iii) or (iv) or if the CUC fails to timely pay any obligation when due to Tyler.
 - (c) In the event either party terminates this Agreement pursuant to this Section 13.2, each party shall return all products, documentation, confidential information, and other information disclosed or otherwise delivered to the other party prior to such termination, and each party may pursue all legal remedies available pursuant to or arising from this Agreement. In no event shall a termination of this Agreement affect the rights and obligations of Tyler or an End User under an executed End User License Agreement, SaaS Agreement, or Standard Maintenance Agreement.
- 13.3 <u>Survival</u>. The following provisions shall survive any termination of this Agreement: 5; 6.4; 7; 8; 9; 10; 11; 13; and 14.

14. MISCELLANEOUS

14.1 <u>Assignment</u>. Neither party may assign this Agreement or any of its respective rights or obligations herein to any third party without the express written consent of the

other party, which consent shall not be unreasonably withheld.

- 14.2 <u>Cumulative Remedies</u>. Except as specifically provided herein, no remedy made available herein is intended to be exclusive of any other remedy, and each and every remedy shall be cumulative and shall be in addition to every other remedy provided herein or available at law or in equity.
- 14.3 <u>Notices</u>. Except as otherwise expressly specified herein, all notices, requests or other communications shall be in writing and shall be deemed to have been given if delivered personally or mailed, by certified or registered mail, postage prepaid, return receipt requested, to the parties at their respective addresses set forth above (or, in the case of a End User, as set forth in the Participation Agreement), or at such other addresses as may be specified in writing by either of the parties. All notices, requests, or communications shall be deemed effective upon personal delivery or three (3) days following deposit in the mail.
- 14.4 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 14.5 <u>Waiver</u>. The performance of any obligation required of a party herein may be waived only by a written waiver signed by the other parties, which waiver shall be effective only with respect to the specific obligation described therein.
- 14.6 <u>Entire Agreement</u>. This Agreement constitutes the entire understanding and contract between the parties and supersedes any and all prior or contemporaneous oral or written representations or communications with respect to the subject matter hereof.
- 14.7 <u>Amendment</u>. This Agreement shall not be modified, amended or in any way altered except by an instrument in writing signed by the properly delegated authority of each party. All amendments or modifications of this Agreement shall be binding upon the parties despite any lack of consideration.
- 14.8 <u>Severability of Provisions</u>. In the event any provision hereof is found invalid or unenforceable pursuant to judicial decree, the remainder of this Agreement shall remain valid and enforceable according to its terms.
- 14.9 Relationship of parties. The parties intend that the relationship between the parties created pursuant to or arising from this Agreement is that of an independent contractor only. Each party shall be solely responsible for all matters relating to, associated with, or arising from the payment of such party's employees, including, without limitation, compliance with social security, disability insurance, withholding and all other wages, salaries, benefits, taxes, demands, and regulations of any nature whatsoever, as well as all employment reporting for such party or any of its employees or agents. Neither party shall be considered an employee of the other party for any purpose, and shall not be entitled to any of the benefits or rights afforded to employees of the other party, including, without limitation, sick leave, vacation leave, holiday pay, public employees retirement system benefits (if applicable), or insurance benefits.
- 14.10 <u>Governing Law</u>. Any dispute arising out of or relating to this Agreement or the breach thereof shall be governed by the laws of the State of Texas, without regard to or application of choice of law rules or principles.

- 14.11 Non-Discrimination. This Agreement is subject to applicable federal and state laws and executive orders relating to equal opportunity and nondiscrimination in employment. Neither Tyler nor its agents or subcontractors shall discriminate in their employment practices against any person by reason of disability, age, race, religion, color, sex, national origin, creed, political affiliation, or veteran status. In addition, Tyler assures that no person will, on the grounds of disability, age, race, creed, national origin, color, religion, sex, political affiliation, or veteran status, be excluded from, be denied the benefit of or be subjected to discrimination pursuant to or arising from any program or activity funded in whole or in part pursuant to or arising from this Agreement. Tyler agrees to comply, and to cause its agents and subcontractors to comply, with the provisions of said laws and orders to the extent any such laws and orders are applicable in the performance of this Agreement.
- 14.12 <u>No Third party Beneficiaries</u>. Nothing in this Agreement is intended to benefit, create any rights in, or otherwise vest any rights upon any third party, including any County (except to the extent that a County becomes an End User and executes an End User License Agreement).
- 14.13 <u>Contra Proferentem</u>. The doctrine of *contra proferentem* shall not apply to this Agreement. If an ambiguity exists in this Agreement, or in a specific provision, neither the Agreement nor the provision shall be construed against the party who drafted the Agreement or provision.
- 14.14 <u>Force Majeure</u>. No party to this Agreement shall be liable for delay or failure in the performance of its contractual obligations arising from any one or more events that are beyond its reasonable control, including, without limitation, acts of God, war, terrorism, and riot. Notwithstanding the foregoing, in every case the delay or failure to perform must be beyond the control and without the fault or negligence of the party claiming excusable delay.
- 14.15 <u>Equitable Relief</u>. The CUC agrees that any violation of this Agreement by the such party with respect to its respective obligations set forth in <u>Section 8</u> shall cause irreparable injury to Tyler and shall entitle Tyler to extraordinary and equitable relief by a court of competent jurisdiction, including, without limitation, temporary restraining orders and preliminary and permanent injunctions, without the necessity of posting bond or security.

20

IN WITNESS WHEREOF, this Agreement has been executed by the parties hereto as of the date first above written.

TEXAS CONFERENCE OF URBAN COUNTIES, INC.

TYLER TECHNOLOGIES, INC.

Name: _ CAA

Title: TechSham Proper Duch

BCC Broke

itle: <u>VP of Sa</u>

Exhibits

Exhibit A Price Lists

A-1 Enterprise Price List

A-2 Software as a Service Price ListA-3 Optional Software Price ListsA-4 2009 T&M Services Hourly Rates

Exhibit B Form Agreements

End User License Agreement and Professional Services Agreement*
(*Includes Maintenance and Support Services Agreement)
Software as a Service and Professional Services Agreement

EXHIBIT A-1 – ENTERPRISE PRICE LIST

Cat	County Population	Undiscounted Member County License Fee	Discounted License Fee (Member Counties before 12/31/2010)	Member Counties after 12/31/2010 or Non-Member Counties
1	1-19,999	140,000	126,000	187,000
2	20,000-29,999	200,000	180,000	277,000
3	30,000-49,999	250,000	225,000	355,000
4	50,000-74,999	350,000	315,000	512,000
5	75,000-149,999	575,000	517,500	751,000
6	150,000-499,999	775,000	697,500	1,040,000
7	500,000-999,999	1,295,000	1,165,500	1,354,000
8	1,000,000-1,749,999	1,750,000	1,575,000	2,080,000
9	1,750,000-2,499,999	1,830,000	1,647,000	3,039,000
10	2,500,000-4,000,000	3,150,000	2,835,000	4,109,000

Individual Case Type License Fees:

- District Clerk Civil / Family 25% of Undiscounted Member County License Fee(or Enterprise License Fee, as appropriate)
- County Clerk Civil / Probate 25% of Undiscounted Member County License Fee (or Enterprise License Fee, as appropriate)
- Criminal (District & County) 50% of Undiscounted Member County License Fee (or Enterprise License Fee, as appropriate)
- Justice of the Peace 30% of Undiscounted Member County License Fee)or Enterprise License Fee, as appropriate)

ANNUAL MAINTENANCE & SUPPORT PRICING

Tier 2 Maintenance & Support:

18% of License Fee

Tier 1 Maintenance & Support:

21% of License Fee

MASTER AGREEMENT EXHIBIT A-2 – SOFTWARE AS A SERVICE PRICE LIST

	Member	County Pricing	Non-Member County Pricing					
Total Users	Monthly Price per User	Monthly Price per User for 24 Hour Shift Personnel (e.g. Sheriff/Jail)	Monthly Price per User	Monthly Price per User for 24 Hour Shift Personnel (e.g. Sheriff/Jail)				
0-10	297.50	157.50	350.00	175.00				
11-20	255.00	135.00	300.00	150.00				
21-40	212.50	112.50	250.00	125.00				
41-60	191.25	101.25	225.00	112.50				
61-80	170.00	90.00	200.00	100.00				
81-100	157.50	78.75	175.00	87.50				
101-120	135.00	67.50	150.00	75.00				
121-150	112.50	56.25	125.00	62.50				
Each user over 150	135.00	67.50	150.00	75.00				

Software as a Service pricing is applicable to:

- Odyssey Case Manager
- Odyssey Jail Manager
- Odyssey Prosecutor
- Odyssey Check Manager
- Odyssey Law Enforcement
- Odyssey Supervision
- Odyssey Financial Manager
- Tyler Jury

Optional features may incur an additional per user monthly fee.

EXHIBIT A-3 - OPTIONAL SOFTWARE PRICE LIST As of 7/27/2010

Section 1. Optional Applications and Modules

Category:	1	2	3	4	5	6	7	8	9	10
Example Clients	Franklin	Gillespie	Kerr	Rockwall	Gregg	Lubbock	Collin	Bexar	Dallas, NM	Harris
医性结合性 医多种性 医多种性 医多种性 医皮肤性 医皮肤性 医皮肤性 医皮肤性 医皮肤性 医皮肤性 医皮肤性 医皮肤				Glynn N	lonroe, Rockd	Lehigh	Lee	NH, Orange	Dade, Wayne	
APPLICATIONS	1	20,000	30,000	50,000	75,000	150,000	500,000	1,000,000	1,750,000	2,500,000
(Based on Population)	19,999	29,999	49,999	74,999	149,999	499,999	999,999	1,749,999	2,499,999	3,999,999
ODYSSEY Session Works: Judges Edition										
Per Judge [10% discount 5 or More, 20% discount 10 or more)	5,000	5,000	5,000	5,000	5,000	5,000	5,000	5,000	5,000	5,000
ODYSSEY CHECK MANAGER	14,000	20,000	26,000	37,000	54,000	75,000	98,000	150,000	219,000	297,000
Check Manager Merchant Website	2,000	2,000	3,000	4,000	5,000	7,000	9,000	14,000	20,000	27,000
ODYSSEY SUPERVISION	24,000	35,000	45,000	65,000	96,000	132,000	172,000	264,000	386,000	522,000
ODYSSEY LAW ENFORCEMENT RMS	16,000	23,000	29,000	42,000	62,000	85,000	111,000	170,000	249,000	336,000
TYLER JURY	11,000	16,000	21,000	30,000	44,000	60,000	78,000	120,000	176,000	237,000
Category:	1	2	3	4	5	6	7	8	9	10
APPLICATIONS	1	10	25	50	100	150	250	1,000	2,000	3,000
(Based on Jail Beds)	9	24	49	99	149	249	999	1,999	2,999	3,999
ODYSSEY JAIL MANAGER										
Jail Management	34,000	53,000	84,000	150,000	180,000	250,000	375,000	450,000	1,000,000	1,500,000
Mugshots	3,000	4,000	7,000	12,000	14,000	19,000	29,000	34,000	75,000	113,000
Jail Commissary Interface	2,000	3,000	5,000	8,000	9,000	13,000	19,000	23,000	50,000	75,000
NorthPointe Classification System	5,880	5,880	5,880	6,374	9,900	15,600	20,340	24,420	24,420	24,420

Section 2. Optional Add-on Features

WINDS TO MAKE THE PARK TO	THE RESIDENCE OF STREET	The second second	and the same of th	America Applications and the		ment of the last o		STREET,	The second second second	M
Franklin	Gillespie	Kerr							Harris	
1	20,000	30,000	50,000	75,000	150,000	500,000	1,000,000	1,750,000	2,500,000	4,000
19,999	29,999	49,999	74,999	149,999	499,999	999,999	1,749,999	2,499,999	3,999,999	Gre
		*** See Featu	ire Sheets for	Implementat	ion detalis, li	stallation, ot	her costs ***			
Ctandard	Ctandard	Ctandard	Ctandard	Chandard	Canadaad	Candoud	Canadaad	Chandard	Chandand	
and the second second second second										
3,000	4,000	5,000	7,000	10,000	13,000	17,000	26,000	38,000	52,000	
9,000	13,000	16,000	24,000	34,000	47,000	62,000	94,000	138,000	186,000	
2,000	3,000	4,000	5,000	7,000	9,000	12,000	18,000	27,000	36,000	
										<u>~</u>
2.000	2.000	2.000	2.500	3.750	5,000	6 500	10,000	14 750	19 750	ğ
										ž
4,000	0,000	7,000	10,000	15,000	20,000	20,000	40,000	33,000	75,000	ž
10,000	22.000	20.000	42.000	62 000	PF PPP	444 000	470.000			ξ
						and the second second second second second				Ϋ́
3,000	4,000	6,000	8,000	11,000	15,000	15,000	15,000	15,000	15,000	SEE SALES MANAGER
										S
Standard	Standard	Standard	Standard	Standard	Standard	Standard	Standard	Standard	Standard	
1.000	2.000	2.000	3.000	4.000	5.000					
				18 (
3,000	4,000	5,000	6,000	9,000	12,000	16,000	24,000	36,000	48,000	
						1		The state of the s		
1	2	3	4	5	6	7	8	9	10	M
1	20,000	30,000	50,000	75,000	150,000	500,000	1,000,000	1,750,000	2,500,000	4,00
19,999	29,999	49,999	74,999	149,999	499,999	999,999	1.749,999	2,499,999	3.999,999	Gri
				implementat					Name and Address of the Owner, when the owner, where the owner, which is the owner, where the owner, which is the owner, which	
Standard	Standard							Ctondand	Considered	
		The second secon								
							22,000	33,000	44,000	
3,000	4,000	5,000	7,000	11,000	14,000	19,000	28,000	41,000	56,000	
6,000	8,000	11,000	15,000	22,000	30,000	39,000	60,000	88,000	119,000	
				See Sale	s Manager					
Standard	Standard	Standard	Standard			Standard	Standard	Ctondard	Ctandard	
5,000	5,000	5,000	5,000				5,000	5,000	5,000	
			-	Will be inclu	ided in produ	ct	-			
1	The second second		4	5	6	7	8	9	10	
1	20,000	30,000	50,000	75,000	150,000	500,000	1,000,000	1,750,000	2,500,000	
19,999	29,999	49,999	74,999	149,999	499,999	999,999	1,749,999	2,499,999	3,999,999	~
										ā
										SEE SALES MGR
23,000	34,000	43,000	62 000	90,000	125 000	163 000	250.000	365,000	494 000	AL.
23,000	54,000	43,000	02,000	30,000	123,000	203,000	250,000	303,000	434,000	ES
										S
2,000	3,000	4,000	5,000	8,000	10,000	13,000	20,000	30,000	40,000	
2,000	3,000	4,000	5,000	8,000	10,000	13,000	20,000	30,000	40,000	
5,000	7,000	9.000	13,000	18.000						
				AND DESCRIPTION OF THE PARTY OF						
	1000						50,000		99,000	
2,000	3,000	4,000	5,000	8,000	10,000	13,000	20,000	30,000	40,000	
2,000	3,000	4,000	5,000	8,000	10,000	13,000	20,000	30,000	40,000	
2,000			5,000	8,000	10,000	13,000	20,000	30,000	40,000	
	3.000	4.000					20,000	25,000	.0,000	
2,000	3,000	4,000	3,000	0,000	10,000	13,000				
2,000							F 000	0.000	10 000	
2,000 1,000	1,000	1,000	2,000	2,000	2,500	4,000	5,000	8,000	10,000	
2,000 1,000 1,000	1,000 1,000	1,000 1,000	2,000 1,000	2,000 2,000	2,500 1,500	4,000 2,000	3,000	5,000	6,000	
2,000 1,000	1,000	1,000 1,000 2,000	2,000 1,000 3,000	2,000 2,000 4,000	2,500	4,000				
2,000 1,000 1,000	1,000 1,000	1,000 1,000	2,000 1,000	2,000 2,000	2,500 1,500	4,000 2,000	3,000	5,000	6,000	M
2,000 1,000 1,000 1,000	1,000 1,000 2,000	1,000 1,000 2,000	2,000 1,000 3,000	2,000 2,000 4,000 S	2,500 1,500 4,500	4,000 2,000 6,000	3,000 9,000 8	5,000 14,000 9	6,000 18,000 10	
2,000 1,000 1,000 1,000 1,000	1,000 1,000 2,000 2	1,000 1,000 2,000 3	2,000 1,000 3,000 4 50,000	2,000 2,000 4,000 5 75,000	2,500 1,500 4,500 6 150,000	4,000 2,000 6,000 7 500,000	3,000 9,000 8 1,000,000	5,000 14,000 9 1,750,000	6,000 18,000 10 2,500,000	4,000
2,000 1,000 1,000 1,000	1,000 1,000 2,000	1,000 1,000 2,000 3 30,000 49,999	2,000 1,000 3,000 4 50,000 74,999	2,000 2,000 4,000 5 75,000 149,999	2,500 1,500 4,500 6 150,000 499,999	4,000 2,000 6,000 7 500,000 999,999	3,000 9,000 8 1,000,000 1,749,999	5,000 14,000 9	6,000 18,000 10	4,000
2,000 1,000 1,000 1,000 1,000	1,000 1,000 2,000 2	1,000 1,000 2,000 3 30,000 49,999	2,000 1,000 3,000 4 50,000 74,999	2,000 2,000 4,000 5 75,000 149,999	2,500 1,500 4,500 6 150,000 499,999	4,000 2,000 6,000 7 500,000	3,000 9,000 8 1,000,000 1,749,999	5,000 14,000 9 1,750,000	6,000 18,000 10 2,500,000	4,000
2,000 1,000 1,000 1,000 1 1 1 19,999	1,000 1,000 2,000 2 20,000 29,999	1,000 1,000 2,000 3 30,000 49,999	2,000 1,000 3,000 4 50,000 74,999 are Sheets for	2,000 2,000 4,000 5 75,000 149,999 Implementati	2,500 1,500 4,500 6 150,000 499,999 ion details, in	4,000 2,000 6,000 7 500,000 999,999 estallation, oti	3,000 9,000 8 1,000,000 1,749,999 her costs ***	5,000 14,000 9 1,750,000 2,499,999	6,000 18,000 10 2,500,000 3,999,999	4,000
2,000 1,000 1,000 1,000 1,000	1,000 1,000 2,000 2	1,000 1,000 2,000 3 30,000 49,999	2,000 1,000 3,000 4 50,000 74,999	2,000 2,000 4,000 5 75,000 149,999	2,500 1,500 4,500 6 150,000 499,999	4,000 2,000 6,000 7 500,000 999,999	3,000 9,000 8 1,000,000 1,749,999	5,000 14,000 9 1,750,000	6,000 18,000 10 2,500,000	4,000
2,000 1,000 1,000 1,000 1 1 1 19,999	1,000 1,000 2,000 2 20,000 29,999	1,000 1,000 2,000 3 30,000 49,999	2,000 1,000 3,000 4 50,000 74,999 are Sheets for	2,000 2,000 4,000 5 75,000 149,999 Implementati	2,500 1,500 4,500 6 150,000 499,999 ion details, in	4,000 2,000 6,000 7 500,000 999,999 estallation, oti	3,000 9,000 8 1,000,000 1,749,999 her costs ***	5,000 14,000 9 1,750,000 2,499,999	6,000 18,000 10 2,500,000 3,999,999	4,000
2,000 1,000 1,000 1,000 1 1 1 19,999	1,000 1,000 2,000 2 20,000 29,999	1,000 1,000 2,000 3 30,000 49,999	2,000 1,000 3,000 4 50,000 74,999 are Sheets for	2,000 2,000 4,000 5 75,000 149,999 Implementati	2,500 1,500 4,500 6 150,000 499,999 ion details, in	4,000 2,000 6,000 7 500,000 999,999 estallation, oti	3,000 9,000 8 1,000,000 1,749,999 her costs ***	5,000 14,000 9 1,750,000 2,499,999	6,000 18,000 10 2,500,000 3,999,999	4,000 Gre
2,000 1,000 1,000 1,000 1 1 19,999	1,000 1,000 2,000 2 20,000 29,999	1,000 1,000 2,000 3 30,000 49,999	2,000 1,000 3,000 4 50,000 74,999 are Sheets for 2500	2,000 2,000 4,000 5 75,000 149,999 Implementation	2,500 1,500 4,500 6 150,000 499,999 lon detalls, Ir	4,000 2,000 6,000 7 500,000 999,999 sstallation, oti	3,000 9,000 8 1,000,000 1,749,999 her costs ***	5,000 14,000 9 1,750,000 2,499,999	6,000 18,000 10 2,500,000 3,999,999	4,000 Gre
2,000 1,000 1,000 1,000 1 1 19,999 2500	1,000 1,000 2,000 2 20,000 29,999 2500	1,000 1,000 2,000 3 30,000 49,999 *** See Featu 2500	2,000 1,000 3,000 4 50,000 74,999 re Sheets for 2500	2,000 2,000 4,000 5 75,000 149,999 Implementation 2500	2,500 1,500 4,500 6 150,000 499,999 Ion detalls, Ir 2500	4,000 2,000 6,000 7 500,000 999,999 sstallation, oti	3,000 9,000 8 1,000,000 1,749,999 ther costs *** 2500	5,000 14,000 9 1,750,000 2,499,999 2500	6,000 18,000 10 2,500,000 3,999,999 2500	4,000 Gre
2,000 1,000 1,000 1,000 1 1 19,999	1,000 1,000 2,000 2 20,000 29,999	1,000 1,000 2,000 3 30,000 49,999	2,000 1,000 3,000 4 50,000 74,999 are Sheets for 2500	2,000 2,000 4,000 5 75,000 149,999 Implementation	2,500 1,500 4,500 6 150,000 499,999 lon detalls, Ir	4,000 2,000 6,000 7 500,000 999,999 sstallation, oti	3,000 9,000 8 1,000,000 1,749,999 her costs ***	5,000 14,000 9 1,750,000 2,499,999	6,000 18,000 10 2,500,000 3,999,999	4,000 Gre
2,000 1,000 1,000 1,000 1 1 19,999 2500	1,000 1,000 2,000 2 20,000 29,999 2500	1,000 1,000 2,000 3 30,000 49,999 *** See Featu 2500 40,000 20,000	2,000 1,000 3,000 4 50,000 74,999 tre Sheets for 2500 50,000 25,000	2,000 2,000 4,000 5 75,000 149,999 Implementation 2500 50,000 25,000	2,500 1,500 4,500 6 150,000 499,999 fon details, in 2500	4,000 2,000 6,000 7 500,000 999,999 stallation, ott	3,000 9,000 8 1,000,000 1,749,999 her costs *** 2500	5,000 14,000 9 1,750,000 2,499,999 2500	6,000 18,000 10 2,500,000 3,999,999 2500	4,000 Gre
2,000 1,000 1,000 1,000 1 1 1,9999 2500	1,000 1,000 2,000 2 20,000 29,999 2500 20,000 10,000	1,000 1,000 2,000 3 30,000 49,999 *** See Featu 2500 40,000 20,000	2,000 1,000 3,000 4 50,000 74,999 re Sheets for 2500 50,000 25,000	2,000 2,000 4,000 5 75,000 149,999 Implementation 2500 50,000 25,000	2,500 1,500 4,500 6 150,000 499,999 ion details, in 2500 100,000 50,000	4,000 2,000 6,000 7 500,000 993,999 sstallation, oti 2500	3,000 9,000 8 1,000,000 1,749,999 her costs *** 2500 125,000 62,500	5,000 14,000 9 1,750,000 2,499,999 2500 125,000 62,500	6,000 18,000 10 2,500,000 3,999,999 2500 150,000 75,000	4,000 Gre
2,000 1,000 1,000 1,000 1 1 19,999 2500	1,000 1,000 2,000 2 20,000 29,999 2500	1,000 1,000 2,000 3 30,000 49,999 *** See Featu 2500 40,000 20,000	2,000 1,000 3,000 4 50,000 74,999 tre Sheets for 2500 50,000 25,000	2,000 2,000 4,000 5 75,000 149,999 Implementation 2500 50,000 25,000	2,500 1,500 4,500 6 150,000 499,999 fon details, in 2500	4,000 2,000 6,000 7 500,000 999,999 stallation, ott	3,000 9,000 8 1,000,000 1,749,999 her costs *** 2500	5,000 14,000 9 1,750,000 2,499,999 2500	6,000 18,000 10 2,500,000 3,999,999 2500	4,000 Gre
2,000 1,000 1,000 1,000 1 1 1 19,999 2500 15,000 7,500	1,000 1,000 2,000 20,000 29,999 2500 20,000 10,000	1,000 1,000 2,000 3 30,000 49,999 2500 40,000 20,000 in to Tyler cor	2,000 1,000 3,000 4 50,000 74,999 re Sheets for 2500 50,000 25,000	2,000 2,000 4,000 5 75,000 149,999 Implementat 2500 50,000 25,000 uures below re 22,000	2,500 1,500 4,500 6 150,000 499,999 Ion detalls, Ir 2500 100,000 50,000	4,000 2,000 6,000 7 500,000 99,999 sstallatlan, at 2500 100,000 50,000 1rty software & 39,000	3,000 9,000 8 1,000,000 1,749,999 her costs *** 2500 125,000 62,500 \$ services ***	5,000 14,000 9 1,750,000 2,499,999 2500 125,000 62,500	6,000 18,000 10 2,500,000 3,999,999 2500 150,000 75,000	4,000 Gre
2,000 1,000 1,000 1,000 1 1 1 1 19,999 2500 15,000 7,500 6,000 4,000	1,000 1,000 2,000 2 20,000 29,999 2500 20,000 10,000	1,000 1,000 2,000 3 30,000 49,999 *** See Featu 2500 40,000 20,000 in to Tyler cor 11,000 7,000	2,000 1,000 3,000 4 50,000 74,999 ore Sheets for 2500 50,000 25,000 15,000 10,000	2,000 2,000 4,000 5 75,000 149,999 Implementati 2500 50,000 25,000 uures below re 22,000 15,000	2,500 1,500 4,500 6 150,000 499,999 lon detalls, in 2500 100,000 50,000 30,000 20,000	4,000 2,000 6,000 7 500,000 999,999 stallatlon, oti 2500 100,000 50,000 30,000 26,000	3,000 9,000 3 1,000,000 1,749,999 ther costs *** 2500 125,000 62,500 & services *** 60,000 40,000	5,000 14,000 9 1,750,000 2,499,999 2500 125,000 62,500 88,000 59,000	6,000 18,000 10 2,500,000 3,999,999 2500 150,000 75,000	4,000
2,000 1,000 1,000 1,000 1 1 1 1,1 19,999 2500 15,000 7,500 6,000 4,000 4,000	1,000 1,000 2,000 2 20,000 29,999 2500 20,000 10,000	1,000 1,000 2,000 3 30,000 49,999 2500 40,000 20,000 in to Tyler cor 11,000 7,000 7,000	2,000 1,000 3,000 4 50,000 74,999 re Sheets for 25,000 25,000 25,000 10,000 10,000	2,000 2,000 4,000 5 75,000 149,999 Implementat. 2500 50,000 25,000 ures below re 22,000 15,000	2,500 1,500 4,500 6 150,000 499,999 Ion details, ir 2500 100,000 50,000 20,000 20,000 20,000 20,000	4,000 2,000 6,000 7 500,000 999,999 stallation, oti 2500 100,000 50,000 1rty software & 39,000 26,000 26,000	3,000 9,000 8 1,000,000 1,749,999 her costs *** 2500 125,000 62,500 & serv/ces *** 60,000 40,000	5,000 14,000 9 1,750,000 2,499,999 2500 125,000 62,500 88,000 59,000 59,000	6,000 18,000 10 2,500,000 3,999,999 2500 150,000 75,000 119,000 79,000 79,000	SEE SALES MANAGER
2,000 1,000 1,000 1,000 1 1 1 1 19,999 2500 15,000 7,500 6,000 4,000	1,000 1,000 2,000 2 20,000 29,999 2500 20,000 10,000	1,000 1,000 2,000 3 30,000 49,999 *** See Featu 2500 40,000 20,000 in to Tyler cor 11,000 7,000	2,000 1,000 3,000 4 50,000 74,999 ore Sheets for 2500 50,000 25,000 15,000 10,000	2,000 2,000 4,000 5 75,000 149,999 Implementati 2500 50,000 25,000 uures below re 22,000 15,000	2,500 1,500 4,500 6 150,000 499,999 lon detalls, in 2500 100,000 50,000 30,000 20,000	4,000 2,000 6,000 7 500,000 999,999 stallatlon, oti 2500 100,000 50,000 30,000 26,000	3,000 9,000 3 1,000,000 1,749,999 ther costs *** 2500 125,000 62,500 & services *** 60,000 40,000	5,000 14,000 9 1,750,000 2,499,999 2500 125,000 62,500 88,000 59,000	6,000 18,000 10 2,500,000 3,999,999 2500 150,000 75,000	4,000 Gre
	19,999 Standard 5,000 3,000 9,000 2,000 4,000 16,000 3,000 Standard 1,000 2,000 3,000 11 19,999 Standard 6,000 5,000 Standard 1,000 2,000 3,000 11 19,999 23,000 2,0	1 20,000 13,000 3,000 4,000 3,000	1	Tranklin Calibaspile Carr Rotwall Calibaspile Calibaspile Carr Calibaspile Caliba	Translin Gillespie	Tenkin			Principle Cert Clyron Moreon Richel Chilery College Chilery College Chilery Ch	Tentah College Colle

^{*}See Exhibit A For Maintenance & Support Rates on Optional Applications, Modules, and Features

Section 3. Optional Add-on Features – Tyler Municipal Court

Level A Level B License Fees Fees Fees Fees License Fees License Fees License License Fees	Populatin Level H License	Population Level I	Population Level J	Population
License Fees Fees		Levell	Lovel	
Criminal Court Case Management \$ 5,500 \$ 6,600 \$ 11,000 \$ 15,400 \$ 17,600 \$ 27,500 \$ 33,000 Probation Management \$ 2,250 \$ 2,500 \$ 3,250 \$ 3,750 \$ 4,400 \$ 6,875 \$ 8,250 Prosecutor Management \$ 2,250 \$ 2,500 \$ 3,250 \$ 3,750 \$ 4,400 \$ 6,875 \$ 8,250 Civil Court Case Management \$ 2,750 \$ 3,300 \$ 5,500 \$ 7,700 \$ 8,800 \$ 13,750 \$ 16,500 Centralized Cash Collections \$ 880 \$ 1,100 \$ 1,650 \$ 2,200 \$ 2,750 \$ 3,300 \$ 6,900 INCODE Scheduling \$ 2,750 \$ 4,000 \$ 4,500 \$ 5,100 \$ 5,700 \$ 6,300 \$ 6,900	Fees	License Fees	License Fees	Level K License Fees
Criminal Court Case Management \$ 5,500 \$ 6,600 \$ 11,000 \$ 15,400 \$ 17,600 \$ 27,500 \$ 33,000 Probation Management \$ 2,250 \$ 2,500 \$ 3,250 \$ 3,750 \$ 4,400 \$ 6,875 \$ 8,250 Prosecutor Management \$ 2,250 \$ 2,500 \$ 3,250 \$ 3,750 \$ 4,400 \$ 6,875 \$ 8,250 Civil Court Case Management \$ 2,750 \$ 3,300 \$ 5,500 \$ 7,700 \$ 8,800 \$ 13,750 \$ 16,500 Centralized Cash Collections \$ 880 \$ 1,100 \$ 1,650 \$ 2,200 \$ 2,750 \$ 3,300 \$ 6,900 INCODE Scheduling \$ 2,750 \$ 4,000 \$ 4,500 \$ 5,100 \$ 5,700 \$ 6,300 \$ 6,900				
Probation Management \$ 2,250 \$ 2,500 \$ 3,250 \$ 3,750 \$ 4,400 \$ 6,875 \$ 8,250	\$ 44,000	\$ 55,000	\$ 75,000	\$ 100,000
Prosecutor Management \$ 2,250 \$ 2,500 \$ 3,250 \$ 3,750 \$ 4,400 \$ 6,875 \$ 8,250 Civil Court Case Management \$ 2,750 \$ 3,300 \$ 5,500 \$ 7,700 \$ 8,800 \$ 13,750 \$ 16,500 Centralized Cash Collections \$ 880 \$ 1,100 \$ 1,650 \$ 2,200 \$ 2,750 \$ 3,300 \$ 4,400 INCODE Scheduling \$ 2,750 \$ 4,000 \$ 5,100 \$ 5,700 \$ 6,300 \$ 6,900	\$ 11.000	,,		
Civil Court Case Management \$ 2,750 \$ 3,300 \$ 5,500 \$ 7,700 \$ 8,800 \$ 13,750 \$ 16,500 Centralized Cash Collections \$ 880 \$ 1,100 \$ 1,650 \$ 2,200 \$ 2,750 \$ 3,300 \$ 4,400 INCODE Scheduling \$ 2,750 \$ 4,000 \$ 4,500 \$ 5,100 \$ 5,700 \$ 6,300 \$ 6,900				
Centralized Cash Collections \$ 880 \$ 1,100 \$ 1,650 \$ 2,200 \$ 2,750 \$ 3,300 \$ 4,400	,			
INCODE Scheduling \$ 2,750 \$ 4,000 \$ 5,100 \$ 5,700 \$ 6,300 \$ 6,900	\$ 22,000 \$ 6,600			9 .
	\$ 7,500			1
- Warrants	\$ 7,500	\$ 6,100	\$ 6,500	\$ 10,500
- Macros				
- Citation Import				
	\$ 4,000	\$ 4,000	\$ 4,000	\$ 4,000
	\$ 4,400			1
	\$ 11,000			
7 0,000 7 0,000 7 0,000 7 2,000	\$ 5,500			
- Import of Export of warrants/dispositions	, 5,500	\$ 0,000	\$ 5,500	\$ 0,000
	\$ 4,400	\$ 4,950	\$ 5,500	\$ 5,500
	\$ 4,400			
	\$ 7,150			
	\$ 2.750			
	\$ 1,100	,	, ,	
	\$ 6,600			
	, -,			
	\$ 13,200	\$ 16,500	\$ 22,500	\$ 30,000

	55/10/12/03	- 200K		OK - 300K	9000	0K - 400K		00K - 500K		00K - 600K	illion	00K - 700K		OK - 800K		0K - 900K		OK - 1.0M	State of	M - 1.5M
				pulation	_	pulation		pulation	_	pulation		pulation		pulation	_	pulation		pulation		pulation
		vel L ense		evel M		Level N License		Level O License		Level P License		Level Q License		Level R License		Level S License		evel T		_evel U .icense
Product Description	F	ees		Fees		Fees		Fees		Fees		Fees		Fees		Fees		Fees		Fees
Municipal Court Case Management																				
Criminal Court Case Management	\$ 2	00,000	Ś	300,000	Ś	400,000	Ś	500,000	Ś	600,000	Ś	700,000	\$	800.000	Ś	900,000	\$ 1	.000,000	s ·	L.500.000
Probation Management	1 1	50,000	Ś	75,000	Ś	100,000	Ś	125.000	Ś	150,000	Ś	175,000	Ś	200,000	Ś	225,000	Ś	250,000	Ś	375,000
Prosecutor Management	\$	50,000	\$	75,000	\$	100,000	Ś	125,000	\$	150,000	Ś	175,000	Ś	200,000	Ś	225,000	Ś	250,000	Ś	375,000
Civil Court Case Management	\$ 1	.00,000	\$	150,000	\$	200,000	\$	250,000	\$	300,000	\$	350,000	\$	400,000	Ś	450,000	Ś	500,000	Ś	750,000
Centralized Cash Collections	\$	15,000	\$	20,000	\$	25,000	\$	30,000	\$	35,000	\$	40,000	\$	40,000	\$	50,000	\$	50,000	\$	60,000
INCODE Scheduling	\$	15,000	\$	20,000	\$	25,000	\$	30,000	\$	35,000	\$	40,000	\$	40,000	\$	50,000	\$	50,000	\$	60,000
- Warrants																				
- Macros																				
- Citation Import																				
Citation Issuing Device Interface	\$	10,000	\$	15,000	\$	15,000	\$	20,000	\$	20,000	\$	25,000	\$	25,000	\$	25,000	\$	25,000	\$	25,000
INCODE Court Web Services	\$	8,800	\$	15,000	\$	20,000	\$	25,000	\$	25,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000
Texas TLETS Warrant Interface	\$	20,000	\$	30,000	\$	40,000	\$	50,000	\$	50,000	\$	60,000	\$	60,000	\$	60,000	\$	60,000	\$	60,000
Court/Police (Non-INCODE) Interface - Import or Export of warrants/dispositions	\$	10,000	\$	15,000	\$	20,000	\$	25,000	\$	25,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000
General Ledger Interface (Non-INCODE)	\$	10,000	\$	15,000	\$	20,000	\$	25,000	\$	25,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000
GIS/Street Index Extract	\$	10,000	\$	15,000	\$	20,000	\$	25,000	\$	25,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000
Red Light Camera Interface	\$	20,000	\$	30,000	\$	40,000	\$	50,000	\$	50,000	\$	60,000	\$	60,000	\$	60,000	\$	60,000	\$	60,000
Collection Agency Export Interface	\$	10,000	\$	15,000	\$	20,000	\$	25,000	\$	25,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000
SETCIC (Harris Co Courts) Collections Export Interface	\$	10,000	\$	15,000	\$	20,000	\$	25,000	\$	25,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000	\$	30,000
Tyler Output Processor Server	\$	30,000	\$	45,000	\$	60,000	\$	75,000	\$	90,000	\$	105,000	\$	120,000	\$	135,000	\$	150,000	\$	225,000
Tyler Content Management	\$	60,000	\$	90,000	\$	120,000	\$	150,000	\$	180,000	\$	210,000	\$	240,000	\$	270,000	\$	300,000	\$	450,000

Maintenance and Support Services on Municipal Court software is 25% of License Fees.

Section 4. Optional Add-on Features – Tyler Public Safety

		0 - 10		11 - 20		21 - 30		31 - 40		41 - 50	I	50 - 75		76 - 100		101 - 125		126 - 150		151 - 175		176 - 300
	1000	Sworn	4	Sworn		worn		Sworn		Sworn		Sworn		Sworn		Sworn		Sworn		Sworn		Sworn
		Officers Level A		Officers	_	fficers	-	Officers	_	Officers	_	Officers	,	Officers	_	Officers	-	Officers		Officers	The second	Officers
	1 '	Level A		Level B License	_	evel C		Level D License	1	Level E .icense	1	Level F _icense		_evel G .icense	1	Level H License	1	Level I	- 1	Level J License		Level K License
Product Description		Fees		Fees		Fees	_	Fees		Fees		Fees	_	Fees	Ι.	Fees		Fees		Fees		Fees
Tyler Public Safety																						
Tyler Public Safety CAD																						
CAD	\$	11,000	\$	13,200	\$	16,500	\$	19,800	\$	23,100	\$	29,700	\$	36,300	\$	42,900	\$	49,500	\$	56,100	\$	62,700
Dispatcher (additional seat)	\$	2,200	Ś	2,475	\$	2,750	\$		\$	3,575	1 .	3,850	\$	4,400	\$	-	\$	5,500				
External Agency (per agency)	\$	825	S	1,100	\$	1,375	\$	1,650	\$	1,925	1 .	2,200	\$	2,475	\$		\$	3,025			1.	
E-911 Server Interface	\$	4,400	Ś	4,950	\$	5,500	\$	6,050	\$	6,600	1 '	7,700	Ś	8,800	\$		\$	11,000	1 '		1.	
E-911 Client Interface (additional clients)	\$	550	s	550	\$	550	\$	550	\$	825	\$	825	\$	1,100	1 '		\$	1,650	1.		1 '	
NCIC Server Interface	s	6,600	\$	6,600	\$	7,150	\$	7,700	\$	8,250	1 '	9,900	Š	11,000	\$		\$		1 '			
NCIC Client Interface (additional clients)	\$	825	\$	825	\$	1,100	\$	1,375	\$	1,650	\$	1,925	\$	2,200	\$		1 .	13,200	1 '		1	
CAD 24x7 Support	Ś	5,000	\$	5,000	\$	6,000	\$	7,500	\$	7,500		10,000	\$	10,000	\$	2,200	\$	2,200			1 '	
- When utilized each call \$250	,	5,000	١,	3,000	7	0,000	7	7,300	٦	7,300	٦	10,000	۶	10,000	Þ	12,000	\$	15,000	\$	15,000	\$	20,000
Combined Search	\$	2,200	\$	2,475	\$	2,750	\$	2 200	1	2 575		2.050		4 400					١.		١.	
10000	7	2,200	7	2,473	٦	2,730	٦	3,300	\$	3,575	\$	3,850	\$	4,400	\$	4,950	\$	5,500	\$	5,500	\$	5,500
Tyler Public Safety Mobile Applications																						
Mobile Communication Server	\$	7,700	\$	8,800	\$	9,900	\$	1,100	\$	12,100	\$	13,200	\$	16,500	\$	19,800	\$	23,100	\$	26,400	\$	30,000
Mobile RMS Client (per seat)		N/C		N/C		N/C		N/C		N/C		N/C		N/C	ľ	N/C	Ĺ	N/C	ľ	N/C	ľ	N/C
Mobile CAD Client (per seat)	\$	1,650	\$	1,650	\$	1,650	\$	1,650	\$	1,650	\$	1,650	\$	1,650	\$	1,650	\$	1,650	\$	1,650	\$	
Mobile AVL Client (per seat)	\$	275	\$	275	\$	275	\$	275	\$	275	\$	275	\$	275	Ś	275	Ś	275	\$	275	\$	•
Mobile Ciations Client (per seat)	\$	550	\$	550	\$	550	\$	550	\$	550	\$	550	\$	550	\$	550	\$	550	Ś	550	1 '	
** pricing subject to change based on # of mobile clients							ľ		ľ				•	-	Ť	550	1	330	,	330	"	330
Tyler Public Safety Mapping																						
CAD Mapping Interface	\$	5,650	\$	5,650	\$	5,650	\$	5,650	\$	E CE0	ė	E CEO	4	5 650	4	F 650	_	F 650				
RMS Mapping Interface	\$	2,350	\$		\$					5,650	\$	5,650	\$	5,650	\$	5,650	\$	5,650	\$	5,650	\$	
The mapping member	٦	2,330	٦	2,350	÷	2,350	\$	2,350	\$	2,350	\$	2,350	\$	2,350	\$	2,350	\$	2,350	\$	2,350	\$	2,350
Tyler Public Safety RMS	\$	19,800	4	26,400	,	22.000	,	20.000	_	46 200		44 400							١.		١.	
- Base RMS System	۶	19,000	\$	26,400	\$	33,000	\$	39,600	\$	46,200	\$	44,400	\$	72,600	\$	85,800	\$	99,000	\$	112,200	\$	125,400
· · · · · · · · · · · · · · · · · · ·									١.		١.											
Case Management/Events Center	\$	6,600	\$	8,800	\$	11,000	\$	13,200	\$	15,400	\$	18,800	\$	24,200	\$	28,600	\$	33,000	\$	37,400	\$	41,800
Vehicle Impound	\$	1,650	\$	2,200	\$	2,750	\$	3,300	\$	3,850	\$	4,950	\$	6,050	\$	7,150	\$	6,050	\$	9,350	\$	10,450
Pawn Tickets	\$	1,100	\$	1,100	\$	1,375	\$	1,650	\$	1,925	\$	2,750	\$	3,300	\$	3,850	\$	3,850	\$	4,950	\$	5,500
Personnel	\$	2,200	\$	2,750	\$	3,300	\$	3,850	\$	4,400	\$	5,500	\$	6,600	\$	7,700	\$	7,150	\$	9,900	\$	11,000
Messaging/Message Board	\$	2,200	\$	2,750	\$	3,300	\$	3,850	\$	4,400	\$	5,500	\$	6,600	\$	7,700	\$	7,150	\$	9,900	\$	11,000
Jail Intake & Booking Module	\$	4,400	\$	4,950	\$	5,500	\$	6,050	\$	6,600	\$	7,700	\$	8,800	\$	9,900	\$	10,800	\$	12,100	\$	13,000
Tyler Public Safety Permits & Registration																						
Alarm Tracking/Permitting	\$	1,100	\$	1,100	\$	1,375	\$	1,650	\$	1,925	Ś	2,750	\$	3,300	\$	3,850	۸,	4.450	_	4.050		F F00
Sex Offender Registration	\$	1,100	Ś	1,100	\$	1,375	\$	1,650	\$	1,925	\$		\$	3,300	\$	3,850	\$	4,450	\$	4,950	\$	5,500
Pet Registration	Ś	1,100	\$	1,375	\$	1,650	\$	1,925	\$	2,200	\$					'		4,450	\$	4,950	\$	5,500
Bicycle Registration	\$	1,100	\$	1,375	\$	1,650	\$	1,925	\$	2,200	\$		\$	3,300	\$	3,850	\$	4,450	\$	4,950	\$	5,500
	,	1,100	7	1,3/3	Ų	1,030	Þ	1,525	Þ	2,200	Þ	2,750	Þ	3,300	\$	3,850	\$	4,450	\$	4,950	\$	5,500
Tyler Public Safety Report Solutions																						
Report Writer	\$	3,850	\$	4,125	\$	4,400	\$	4,675	\$	4,950	\$	5,500	\$	6,050	\$	6,600	\$	7,000	\$	7,700	\$	8,250
Additional Designer (per seat)	\$	550	\$	550	\$	550	\$	550	\$	550	\$	550	\$	550	\$	550	\$	550	\$	550	\$	550
Tyler Public Safety Property Room/Evidence Management																						
Property Room	Ś	3,300	Ś	4,400	\$	5,500	4	6 600		7.700											١.	
PDA Server Software	\$	2,200	\$	2,475	\$		\$	6,600	\$	7,700	\$		\$	12,100	\$	14,300	\$	16,500	\$	18,700	\$	20,900
PDA Interface (per handheld)	\$	330				2,750	\$	3,300	\$	3,850	\$		\$	4,400	\$	4,950	\$	5,500	\$	6,050	\$	6,600
by menace (per nanunera)	٦	330	\$	330	\$	330	\$	330	\$	330	\$	330	\$	330	\$	330	\$	330	\$	330	\$	330
Tyler Public Safety Interfaces														1								
Tyler Court Case Management Interface		N/C		N/C		N/C		N/C		N/C		N/C		N/C		N/C		N/C		N/C		N/C
Brazos Technology Citation Interface	\$	2,750	\$	2,750	\$	2,750	\$	2,750	\$	2,750	\$	1	\$	2,750	\$		\$	2,750	\$	2,750	\$	2,750
Third Party Citation Issuing Device Interface	\$	5,000	\$	5,000	\$	5,000	\$		\$		\$		\$	5,000	\$	5,000	\$	5,000	\$	5,000	\$	5,000
LEAP Interface	\$	2,500	\$	2,500	\$	2,500			\$	2,500	\$	1	\$		\$		\$	2,500	\$	2,500	\$	2,500
LiveScan Fingerprint System Interface	\$	4,400	\$	4,400	\$	4,400	\$		\$	4,400	\$		\$		\$		\$	4,400				
Firehouse CAD Monitor Interface	\$	5,500	\$	5,500	\$	5,500	\$		\$		\$		\$	5,500		1			\$	4,400	\$	4,400
Accident Reporting Diagramming Software Interface	Ś	1,000	\$	1	\$	1,000	\$		\$	1	\$		\$				\$	5,500		5,500	\$	5,500
CAD Call Sheet Export to Another TPS Agency	\$	1,500	\$		\$	1,500	\$								\$		\$	1,000		1,000	\$	1,000
	7	1,300	÷	1,500	ų	1,500	P	1,500	Þ	1,500	\$	1,500	\$	1,500	\$	1,500	\$	1,500	\$	1,500	\$	1,500

Maintenance and Support Services on Public Safety software is 25% of License Fees.

EXHIBIT A-3 - OPTIONAL SOFTWARE PRICE LIST As of 3/18/2014

Section 1. Optional Applications and Modules

Category:	1	2	3	4	5	6	7	8	9	1
Example Clients	Franklin	Gillespie	Кепт	Rockwall	Gregg	Lubbock	Collin	Bexar	Dallas, NM	Harri
				Glynn J	Monroe, Rockd	Lehigh	Lee	NH, Orange	Dade, Wayne	
APPLICATIONS	1	20,000	30,000	50,000	75,000	150,000	500,000	1,000,000	1,750,000	2,500,00
(Based on Population)	19,999	29,999	49,999	74,999	149,999	499,999	999,999	1,749,999	2,499,999	3,999,99
ODYSSEY FINANCIAL MANAGER										
Full GL Lite	8,000	12,000	15,000	22,000	31,000	43,000	56,000	86,000	126,000	170,000
ODYSSEY ATTORNEY MANAGER										
Prosecutor	16,000	23,000	29,000	42,000	62,000	85,000	111,000	170,000	249,000	336,000
Public Defender	9,600	13,800	17,400	25,200	37,200	51,000	66,600	102,000	149,400	201,600
Both	20,480	29,440	37,120	53,760	79,360	108,800	142,080	217,600	318,720	430,080
ODYSSEY CHECK MANAGER	14,000	20,000	26,000	37,000	54,000	75,000	98,000	150,000	219,000	297,000
Check Manager Merchant Website	2,000	2,000	3,000	4,000	5,000	7,000	9,000	14,000	20,000	27,000
ODYSSEY SUPERVISION	24,000	35,000	45,000	65,000	96,000	132,000	172,000	264,000	386,000	522,000
ODYSSEY LAW ENFORCEMENT RMS	16,000	23,000	29,000	42,000	62,000	85,000	111,000	170,000	249,000	336,000
ODYSSEY INDEX	14,000	20,000	26,000	37,000	54,000	74,000	97,000	148,000	217,000	293,000
SUPERVISION - PRETRIAL										
Pretrial Release Module (decision-making component)	12,000	17,500	22,500	32,500	48,000	66,000	86,000	132,000	173,700	234,900
Pretrial Services (plus Supervision Lite/no CSTS reporting)	18,000	26,250	33,750	48,750	72,000	99,000	129,000	198,000	260,550	352,350
SUPERVISION - ALL FUNCTIONS	24,000	35,000	45,000	65,000	96,000	132,000	172,000	264,000	386,000	522,000
TYLER JURY	11,000	16,000	21,000	30,000	44,000	60,000	78,000	120,000	176,000	237,000
Tyler Jury Web Portal	15,000	15,000	15,000	15,000	15,000	15,000	15,000	15,000	15,000	15,000
Category:	1	2	3	4	5	6	7	8	9	10
APPLICATIONS	1	10	25	50	100	150	250	1,000	2,000	3,000
(Based on Jail Beds)	9	24	49	99	149	249	999	1,999	2,999	3,999
ODYSSEY JAIL MANAGER					1					
Jail Management	34,000	53,000	84,000	150,000	180,000	250,000	375,000	450,000	1,000,000	1,500,000
Mugshots	3,000	4,000	7,000	12,000	14,000	19,000	29,000	34,000	75,000	113,000
Jail Data Export Enterprise	2,000	3,000	5,000	8,000	9,000	13,000	19,000	23,000	50,000	75,000
Jali Data Export Single Vendor	500	750	1,250	2,000	2,250	3,250	4,750	5,750	12,500	18,750
NorthPointe Classification System	5,880	5,880	5,880	6,374	9,900	15,600	20,340	24,420	24,420	24,420
Biometric Identification (requires hardware)	3,350	3,600	4,100	4,850	5,100	6,100	12,325	20,875	35,200	48,995
LiveScan	5,000	5,000	5,000	5,000	5,000	5,000	5,000	5,000	5,000	5,000
TDEx	1,000	2,000	2,000	3,000	4,000	5,000	7,000	10,000	15,000	20,000

Section 2. Optional Add-on Features

Category:	1	2	3	4	5	6	7	8	9	10
ODYSSEY FUNCTIONAL OPTIONS	1	20,000	30,000	50,000	75,000	150,000	500,000	1,000,000	1,750,000	2,500,000
(By Population)	19,999	29,999	49,999	74,999	149,999	499,999	999,999	1,749,999	2,499,999	3,999,999
ODYSSEY Session Works: Clerks Edition										
Enterprise - All Case Types	17,000	25,000	32,000	46,000	68,000	94,000	122,000	187,000	273,000	369,000
1 Court Type (Crim/District/etc)	5,000	8,000	10,000	14,000	20,000	28,000	36,000	55,000	80,000	108,000
2 Court Types (Civ/County/etc)	10,000	14,000	18,000	25,000	37,000	51,000	66,000	101,000	148,000	200,000
3 Court Types (Traf/Justice/etc)	14,000	20,000	25,000	36,000	53,000	73,000	94,000	144,000	211,000	285,000
ODYSSEY Session Works: Judges Edition			= =							
Per Judge [10% discount 5 or More, 20% discount 10 or more]	5,000	5,000	5,000	5,000	5,000	5,000	5,000	5,000	5,000	5,000
Document Management										
Enterprise DMS (Batch/Workflow)	12,000	17,000	22,000	32,000	47,000	64,000	84,000	128,000	187,000	253,000
Auto Attach (Req. Enterprise)	5,000	7,000	9,000	13,000	19,000	26,000	34,000	52,000	76,000	103,000
Record on Appeal Creator	3,000	4,000	5,000	7,000	10,000	13,000	17,000	26,000	38,000	52,000
e-Signatures + merged to TIFF OCR (Optical Character Recognition)	7,000	10,000	12,000	17,000	25,000	34,000	45,000	68,000	100,000	135,000
Level 2 - Auto Convert all new documents to PDF (1)	5,000	7,000	9,000	13,000	19,000	26,000	34,000	52,000	76,000	103,000
Level 3 - L2 + Scheduled back conversion of active files (1)	6,000	8,400	10,800	15,600	22,800	31,200	40,800	62,400	91,200	123,600
Integration Toolkits										
Case Manager Libraries	15,000	20,000	40,000	50,000	50,000	100,000	100,000	125,000	125,000	150,000
Jail Manager Libraries	7,500	10,000	20,000	25,000	25,000	50,000	50,000	62,500	62,500	75,000
Third Party Connectors										
Generic E-Filing Connector	4,000	6,000	7,000	10,000	15,000	20,000	26,000	40,000	59,000	79,000
PRIA eRecording Standards										
1.0 Recording Interface	1,000	2,000	2,000	3,000	4,000	5,000	7,000	10,000	15,000	20,000
2.0 Recording Interface	1,000	2,000	2,000	3,000	4,000	5,000	7,000	10,000	15,000	20,000

EXHIBIT A-4 – 2009 T&M SERVICES HOURLY RATES

Project Management

\$160.00 per hour

Technical Services

\$149.00 per hour

Training Services

\$138.00 per hour



End User License Agreement and Professional Services Agreement

This End User License Agreement and Professional Services Agreement (this "Agreement") is made and entered into by and between Tyler Technologies, Inc., a Delaware corporation ("Tyler"), and ______ (the "End User").

Background

WHEREAS, the Texas Conference of Urban Counties (the "CUC") has previously published a Request for Proposal that defines the CUC's requirements for certain judicial administration software and Tyler responded with a Proposal that met the CUC's requirements; and thereafter the CUC and Tyler entered into a contract (the "CUC Master Agreement") which defines the terms and conditions whereby Tyler will provide to End Users certain licenses, professional services, and maintenance and support services; and

WHEREAS, End User is authorized to procure the Licensed Property and services under the CUC Master Agreement; and

WHEREAS, End User desires to engage Tyler to license certain software and to provide certain professional services related thereto, all on the terms and conditions set forth in this Agreement.

NOW, THEREFORE, in consideration of the mutual promises contained herein, along with other good and valuable consideration, the receipt and sufficiency of which all parties mutually acknowledge, Tyler and End User agree as follows:

- A. Tyler shall furnish the products and services described in this Agreement, and End User shall pay the prices set forth in this Agreement.
- B. This Agreement consists of this cover and signature page and the following attachments and exhibits attached hereto and to be attached throughout the Term of this Agreement, all of which are incorporated by reference herein:
 - Schedule 1. Investment Summary
 - Exhibit A. Software License and Professional Services Agreement
 - Exhibit B. Software Maintenance Agreement

IN WITNESS WHEREOF, this Agreement has been executed by a duly authorized officer of each Party hereto to be effective as of the date last set forth below (the "Effective Date"):

TYLER TECHNOLOGIES, INC.

END USER

Signature:	Signature:	
Date:	Date:	
Name:	Name:	
Title:	Title:	
Address:	Address:	

(Schedule 1) Investment Summary

Subtotal \$

Estimated Travel Expenses \$
Total Contract Price \$

(M&S Fees due annually in advance)

Software		Maintenance &	Support			
oftware Licenses			Licens	se Fees	Support Type	Annual M& Fee
Licensed Software						
						-
						-
						_
						_
						_
rofessional Services			IO SID HTG TO HE			
T&M Services	Rate	Hours		Cost		
Project Management	-	-	\$	-		
Operational Analysis	-	-	\$	-		
Data Conversion	-	-	\$	-		
Customization & Interfaces	-	-	\$	-		
Configuration & Consulting	-	-	\$ \$ \$	-		
Training	-	-	\$	-		
Go-Live Assistance	_	-	\$	-		
Forms Setup Assistnace	-	-	\$	-		
Services Subtotal						
nbedded Third Party Software			(
None						
	Total Lice	ense Fees	\$	_	Maintenance &	Support Fee
			Ψ.			ouppoit i ot

(Exhibit A) Software License and Professional Services Agreement

This Software License and Professional Services Agreement is made and entered into as of the Effective Date by and between Tyler and End User

WHEREAS, End User desires to engage Tyler to license certain software and to provide certain professional services related thereto, all on the terms and conditions set forth in this Agreement.

NOW, THEREFORE, in consideration of the mutual promises contained herein, along with other good and valuable consideration, the receipt and sufficiency of which all parties mutually acknowledge, Tyler and End User agree as follows:

1. CERTAIN DEFINITIONS

- 1.1. <u>Agreement</u> means this Software License and Professional Services Agreement, including all exhibits attached hereto and to be attached throughout the Term of this Agreement, all of which are incorporated by reference herein.
- 1.2. <u>Business Day</u> means any day, Monday through Friday, excepting any federal holiday.
- 1.3. <u>Claims</u> mean any and all claims, liens, demands, damages, liability, actions, causes of action, losses, judgments, costs, and expenses, including reasonable attorneys' fees and expenses.
- 1.4. <u>Current Production Software Version</u> means the current production version of Tyler's software listed on the Investment Summary.
- 1.5. <u>Defect</u> means any bug, error, contaminate, malfunction, or other defect in the Licensed Software caused by, arising from, or emanating from the reasonable control of Tyler that renders the Licensed Software in non-conformance with Tyler's then current published specifications.
- 1.6. <u>Documentation</u> means the user's operating manuals and any other materials in any form or media provided by Tyler to the users of the Licensed Software.
- 1.7. <u>Embedded Third Party Software</u> means licensed third party software (other than Third Person Software) that is required to provide the functionality of the Licensed Software, which as of the date of this Agreement, consists of the software set forth on Schedule 1 labeled as "Embedded Third Party Software".
- 1.8. <u>Escrow Agent</u> means Iron Mountain Intellectual Property Management, Inc.
- 1.9. <u>Escrow Agreement</u> means the Master Escrow Service Agreement between Tyler and Escrow Agent.
- 1.10. <u>Indemnified Parties</u> mean End User and each of its personnel, agents, successors, and permitted assigns.
- 1.11. <u>Investment Summary</u> means the summary of fees and services set forth on Schedule 1.
- 1.12. <u>License Fee</u> means the "Total License Fees" as set forth on the Investment Summary, which is due and payable as set forth in Section 4.1.
- 1.13. $\underline{\text{Licensed Property}}$ means the Licensed Software and the Documentation.
- 1.14. <u>Licensed Software</u> means: (a) the Current Production Software Version; (b) Embedded Third Party Software; and (c) any Local Enhancements
- 1.15. <u>Local Enhancements</u> means any refinement, enhancement, or other customization to the Current Production Software Version to be developed by Tyler per the Investment Summary.
- 1.16. <u>Maintenance and Support Fees</u> has the meaning set forth in Exhibit B Software Maintenance Agreement.

- 1.17. Party means, individually, Tyler and End User.
- 1.18. <u>Project</u> means the delivery and license of the Licensed Property and the performance of all services to be provided by Tyler in accordance with the provisions of this Agreement.
- 1.19. <u>Project Manager</u> means the person designated by each Party who is responsible for the management of the Project.
- 1.20. <u>Software Maintenance Agreement</u> means the maintenance and support services agreement attached hereto as Exhibit B.
 - 1.21. T&M means time and materials.
- 1.22. Third Person Hardware means the CPUs, servers, and other hardware to be leased, purchased, or otherwise acquired by End User from a third party that is minimally required to operate the Licensed Software and such other CPUs, servers, and other hardware that End User has actually leased, purchased or otherwise acquired and/or may be minimally required in the future to operate the Licensed Software.
- 1.23. Third Person Software means the operating systems and other software to be licensed, purchased, or otherwise acquired by End User from a third party that is minimally required to operate the Licensed Software and such operating systems and other software that End User has actually licensed, purchased, or otherwise acquired and/or may be minimally required in the future to operate the Licensed Software.
- 1.24. Confidential and Proprietary Information means all information in any form relating to, used in, or arising out of a Party's operations and held by, owned, licensed, or otherwise possessed by such Party (the "Owner") (whether held by, owned, licensed, possessed, or otherwise existing in, on or about the Owner's premises or the other Party's offices, residence(s), or facilities and regardless of how such information came into being, as well as regardless of who created, generated or gathered the information). including, without limitation, all information contained in, embodied in (in any media whatsoever) or relating to the Owner's inventions, ideas, creations, works of authorship, business documents, licenses, correspondence, operations, manuals, performance manuals, operating data, projections, bulletins, customer lists and data, sales data, cost data, profit data, financial statements, strategic planning data, financial planning data, designs, logos, proposed trademarks or service marks, test results, product or service literature, product or service concepts, process data, specification data, know how, software, databases, database layouts, design documents, release notes, algorithms, source code, screen shots, and other research and development information and data. Notwithstanding the foregoing, Confidential and Proprietary Information does not include information that: (a) becomes public other than as a result of a disclosure by the other Party in breach hereof; (b) becomes available to the other Party on a non-confidential basis from a source other than teh Owner, which is not prohibited from disclosing such information by obligation to the Owner; (c) is known by the other Party prior to its receipt from the Owner without any obligation of confidentiality with respect thereto; or (d) is developed by the other Party independently of any disclosures made by the Owner.

2. IMPLEMENTATION PLAN; NOTICE TO PROCEED

2.1. <u>Project Framework.</u> This Agreement sets forth the terms whereby Tyler shall provide to the End User, and the End User shall acquire from the Tyler, the following (and each on the terms and subject to the conditions of this Agreement): (a) a license for the Licensed Property for the License Fee; (b) certain implementation, installation, testing, and training services related to the Licensed Software for the T&M fees set forth in the Investment Summary; and (c) certain maintenance and support services.

2.2. Implementation Plan.

(a) Within ninety (90) days of the Effective Date (or upon such other schedule as the Parties may mutually agree), Tyler shall devote sufficient resources to develop an Implementation Plan. The Implementation Plan shall provide for timetables, fees and expenses for, among other things, (i) the development of Local Enhancements for the End User, (ii) the delivery and installation of the applicable Licensed Software for the End User, (iii) the training of End User personnel, (iv) the staffing plan for the implementation of the Project, and (v) a scheduled timetable for periodic update reporting on the status of the project by parties to the respective End User's Commissioners Court. Tyler shall provide End User with a revised Investment Summary which shall, upon issuance of a Notice to Proceed by the End User, be incorporated into this End User License Agreement.

- (b) Tyler is authorized to invoice the End User for the Implementation Plan, on a T&M basis as set forth in the Investment Summary, which shall be invoiced and paid in accordance with Section 4.4.
- (c) Within ninety (90) days of the completion of the Implementation Plan (or upon such other schedule as the Parties may mutually agree in writing), End User shall, in End User's sole discretion, either (i) issue a Notice to Proceed with the Project, subject to the terms and conditions of this Agreement or (ii) issue a Termination Notice.
- (d) If End User issues a Notice to Proceed, the license fee shall be due and payable as set forth in Section 4.1.
- (e) If End User issues a Termination Notice, then: (i) this Agreement shall terminate as of the date of such termination notice; (ii) End User shall have no further obligation to participate in the Project, except for paying Tyler for all costs and expenses incurred by Tyler in performing the Implementation Plan, which shall be paid within thirty (30) business days of the Termination Notice; and (iii) within thirty (30) Business Days of any such termination, each party shall return all products, documentation, confidential information, and other information disclosed or otherwise delivered to the other party prior to such termination.

3. TITLE AND LICENSE

- 3.1. <u>License Grant</u>. In consideration for the License Fee, which shall be due and payable as set forth in Section 4, Tyler hereby grants to End User a non-exclusive, royalty-free, revocable license (and sublicense with respect to the Embedded Third Party Software) to use the Licensed Property for End User's internal administration, operation, and/or conduct of End User's business operations by an unlimited number of users employed by End User on an unlimited number of computers and/or computer stations utilized by End User. Upon End User's payment of the License Fee in full, the foregoing licenses shall become irrevocable, subject to the restrictions on use set forth herein.
- 3.2. <u>Restrictions</u>. Unless otherwise expressly set forth in this Agreement, End User shall not (a) reverse engineer, de-compile, or disassemble any portion of the Licensed Software or (b) sublicense, transfer, rent, or lease the Licensed Software or its usage. To the extent End User employs contractors, subcontractors, or other third parties to assist in the Project, End User shall obtain from such third parties an executed Tyler confidentiality agreement prior to such parties being permitted access to Tyler Confidential and Proprietary Information.
- 3.3. <u>Copies</u>. End User may make and maintain such copies of the Licensed Property as are reasonably appropriate for its use and for archival and backup purposes; provided, however, that End User shall retain all proprietary notices, logos, copyright notices, and similar markings on such copies.
- 3.4. Embedded Third Party Software. The license grant set forth in Section 3.1 includes the right to use any Embedded Third Party Software; provided, however, that such access to and use of such Embedded Third Party Software shall be according to such terms, conditions, and licenses as are imposed by the manufacturers and/or third party licensors of such Embedded Third Party Software. All such Embedded Third Party Software shall be included in the License Fee. Tyler shall pass through to End User any and all warranties granted to Tyler by the owners, licensors, and/or distributors of such Embedded Third Party Software. End User shall be responsible for procuring and paying for all Third Person Software.

3.5. <u>Title</u>.

- (a) Tyler represents and warrants that it is the owner of all right, title, and interest in and to the Licensed Software (other than Embedded Third Party Software) and all components and copies thereof. Nothing in this Agreement shall be deemed to vest in End User any ownership or intellectual property rights in and to Tyler's intellectual property (including, without limitation, Tyler Confidential and Proprietary Information), any components and copies thereof, or any derivative works based thereon prepared by Tyler.
- (b) All training materials developed solely by either Party shall be the sole property of such Party. Any training materials developed jointly by the Parties shall be owned jointly by the Parties, and each Party shall be entitled to exercise all rights of ownership of such materials without any duty to account to the other, subject to Section 10.
- (c) All End User data shall remain the property of End User. Tyler shall not use End User data other than in connection with providing the services pursuant to this Agreement.
- 3.6. End User Modifications. Tyler shall have no liability pursuant to this Agreement or the Software Maintenance Agreement for any damages or defects to the Licensed Software caused, directly or indirectly, by End User Modifications or other changes to the Licensed Software that are implemented without the prior written consent of Tyler.

4. FEES AND INVOICING

4.1. <u>License Fee</u>. End User shall pay the License Fee in accordance with the following payment plan:

Payment Event	% of License Fee Payable
Issuance of a Notice to Proceed	60%
Commencement of Operational Use	40%

Invoices shall be sent to End User upon each Payment Event, which shall be paid in accordance with Section 4.4 and Section 4.5.

- 4.2. <u>Professional Services Charges</u>. T&M charges for all professional services to be performed hereunder shall be invoiced and paid by End User in accordance with Section 4.4.
- 4.3. <u>Expenses</u>. End User shall reimburse Tyler for travel, lodging, and food expenses actually and reasonably incurred by Tyler in performing its professional services herein in accordance with Section 4.4.
- 4.4. Invoice and Payment for Implementation Services. Tyler shall invoice End User for implementation services and associated expenses billed on a T&M basis herein on a monthly basis. Each invoice shall state the total invoiced amount and shall be accompanied by a reasonably detailed itemization of services and expenses. End User's payment shall be governed by chapter 2251, Texas Government Code. All payments shall be made in U.S. currency. Interest charges for any overdue payments shall be paid by End User in accordance with section 2251.025, Texas Government Code.

In the event that an error appears in an invoice submitted by Tyler, End User shall notify Tyler of the error not later than the twenty first (21st) day after the date End User receives the invoice. If the error is resolved in favor of Tyler, Tyler shall be entitled to receive interest on the unpaid balance of the invoice submitted by Tyler beginning on the date that the payment for the invoice became overdue. If the error is resolved in favor of the End User, Tyler shall submit a corrected invoice that must be paid in accordance with section 2251.021, Texas Government Code. The unpaid balance accrues interest as provided by chapter 2251, Texas Government Code, if the corrected invoice is not paid by the appropriate date.

- 4.5. <u>Invoice and Payment of License Fees and Maintenance and Support Fees</u>. License Fees and Maintenance and Support Fees, as set forth on the Investment Summary, shall be invoiced and paid as set forth below:
 - (a) Tyler shall invoice the Texas Conference of Urban Counties (the "CUC") for License Fees and Maintenance and Support Fees incurred by the End User in accordance with the terms of the CUC Master Agreement. Tyler shall use reasonable

- efforts to submit such invoices to the CUC sixty (60) days prior to the initial License Fee and/or Maintenance and Support Fee due date or the anniversary thereof, as applicable.
- (b) The CUC shall be responsible for invoicing End User and collecting payments from End User with respect to the License Fees and Maintenance and Support Fees. The CUC shall promptly, but in any event within thirty (30) days of receipt of License Fees and Maintenance and Support Fees from End User, remit to Tyler all License Fees and Maintenance and Support Fees collected on behalf of an End User. In the event that the CUC fails to timely perform its invoice, collection, and remittance obligations under this section, Tyler shall have the right, in its sole discretion and upon written notice to the CUC and End User, to invoice End User directly for all future License Fees and Maintenance and Support Fees.

5. PROJECT IMPLEMENTATION

- 5.1. <u>Professional Services</u>. Attached hereto as Schedule 1 is Tyler's good faith estimate of the hours and fees associated with the services to be performed by Tyler for End User, including travel time by Tyler's personnel from Tyler's place of business to and from End User's place of business, and for which End User shall pay on a T&M basis. Additional services requested by End User which are beyond those hours detailed in Schedule 1 will be billed at Tyler's then current services rates.
- 5.2. Office Space. End User shall, at its sole expense, provide reasonable access to office space, telephone access, network access (including providing Tyler reasonable access to a secure virtual private network connection or other comparable connection for use by Tyler from time to time on a non-dedicated basis), Internet connections, and such other facilities as may be reasonably requested by Tyler for use by Tyler personnel for the purpose of performing this Agreement while such personnel are working on-site and engaged in Project-related services.
- 5.3. Third Person Hardware and Third Person Software. End User shall be responsible to purchase, install, and configure all Third Person Hardware and Third Person Software. Tyler shall have no liability for defects in the Third Person Hardware or Third Person Software.
- 5.4. <u>Cooperation</u>. End User acknowledges that the implementation of the Project is a cooperative process requiring the time and resources of End User personnel. End User shall, and shall cause its personnel to, use all reasonable efforts to cooperate with and assist Tyler as may be reasonably required to timely implement the Project, including, without limitation, providing reasonable information regarding its operations and reasonable access to its facilities. Tyler shall not be liable for failure to timely implement the Project when such failure is due to Force Majeure (as identified in Section19.15) or to the failure by End User personnel to provide such cooperation and assistance (either through action or omission).

6. DELIVERY AND INSTALLATION OF THE LICENSED SOFTWARE

- 6.1. <u>Delivery: Risk of Loss</u>. Tyler shall deliver the Licensed Software to End User's place of business. Risk of loss of the Licensed Software, and media on which such may be delivered, shall remain with Tyler at all times until completed delivery.
- 6.2. <u>Installation; Diagnostic Testing.</u> Tyler shall install the Licensed Software at End User's place of business. Upon installation, Tyler shall conduct its standard diagnostic evaluation to determine that the Licensed Software is properly installed and shall notify the End User's Project Manager in writing after successful completion thereof.

7. VERIFICATION OF THE LICENSED SOFTWARE; FINAL ACCEPTANCE

7.1. <u>Verification Procedure.</u> Upon installation of the Licensed Software in accordance with the timetables set forth in the Implementation Plan, Tyler shall perform its standard test procedures and shall certify to End User that the Licensed Software is in substantial conformance with Tyler's then current published specifications (the "Verification Procedure") and is ready to commence Operational Use.

- 7.2. Optional End User Validation. End User may, in its sole and absolute discretion, monitor the Verification Procedure by performing its own defined internal validation process to test the software to determine if it substantially complies with Tyler's then current published specifications. Such validation test shall constitute End User's validation.
- 7.3. Results Final; Correction. Tyler's verification or End User's validation that the Licensed Software substantially complies with the then current published specifications shall be final and conclusive except for latent defect, fraud, and such gross mistakes that amount to fraud. In the event said verification / validation becomes other than final, End User's right and remedy against Tyler shall be to require Tyler to correct the cause thereof or exercise its rights under Section 17.2. If End User has made modifications to the software programs, Tyler will not make such corrections, unless such modifications were specifically authorized in writing by Tyler.
- 7.4. Operational Use. Notwithstanding anything to the contrary herein, End User's use of the Licensed Software for its intended purpose ("Operational Use") shall constitute Tyler's verification or End User's validation of the software products, without exception and for all purposes.
- 7.5. <u>Final Acceptance</u>. When the Licensed Software is ready to commence Operational Use, End User shall be deemed to have "Final Acceptance" of the Licensed Software, this Agreement shall terminate (subject to Section 17.3 (Survival)), and the Licensed Software shall then become subject to the terms and conditions of the Software Maintenance Agreement.

8. TRAINING

To the extent that training services are included in Schedule 1, Tyler shall train End User in accordance with a mutually agreeable training plan. The training plan shall outline the training required for personnel to operate the Licensed Software. Tyler shall provide End User personnel with only the number of hours of training for the respective portions of the Licensed Software as set forth in the Schedule 1. Training shall be provided at End User's principal place of business or other site selected by End User. Training shall be performed according to the training plan, but in any event shall be "hands-on" using production-ready versions of the Licensed Software. The courses shall train End User's employees or agents in a manner to provide basic end user training. End User shall be responsible for providing an adequately equipped training facility to operate the Licensed Software.

9. MAINTENANCE SERVICES

- 9.1. <u>Maintenance and Support Agreement</u>. Upon the commencement of Operational Use, Tyler shall provide End User with maintenance and support services for the Licensed Software, and End User shall pay the Maintenance and Support Fees.
- 9.2. Responsibilities of End User. In addition to the other responsibilities set forth herein, End User shall: (a) provide all training of its personnel; (b) collect, prepare, and enter all data necessary for the day-to-day operations of the Licensed Software; (c) retain separate copies of all conversion data delivered to Tyler; (d) provide the computer system on which the Licensed Software will be loaded and operated; (e) provide the requisite networks; (f) maintain an internal help desk function; (g) prior to Project completion, install all changes or updates into the Licensed Software and Third Person Software products that are furnished by Tyler for the purpose of correcting failures of the Licensed Software to conform to, and perform in accordance with, the requirements of this Agreement; and (h) maintain, as part of End User's computer system, a secure Microsoft VPN connection for use by Tyler.

10. CONFIDENTIAL AND PROPRIETARY INFORMATION

10.1. Protection of Confidential and Proprietary Information. Each Party shall not disclose, disseminate, transmit, publish, distribute, make available, or otherwise convey the other Party's Confidential and Proprietary Information, and each Party shall not use, make, sell, or otherwise exploit any such other Party's Confidential and Proprietary Information for any purpose other than the performance of this Agreement, without the other Party's written consent, except: (a) as may be required by law, regulation, judicial, or administrative process; or (b) as required in litigation pertaining to this Agreement, provided that the other Party is given advance notice of such intended disclosure in order to permit it the opportunity to seek a protective order. The Parties shall ensure that all individuals assigned to

perform services herein shall abide by the terms of this Section 10.1 and a Party shall be responsible for breaches by such persons acting by or for such Party.

10.2. Judicial and Administrative Proceedings. If a Party is requested or required (by oral questions, interrogatories, requests for information or documents in legal proceedings, subpoena, civil investigative demand, requirements of the Texas Public Information Act, or other similar process) to disclose any Confidential and Proprietary Information of the other Party (the "Owner"), such Party shall provide the Owner with prompt written notice of such request or requirement so that the Owner may seek protective orders or other appropriate remedies and/or waive compliance with the provisions of this Agreement. If, in the absence of a protective order or other remedy or the receipt of a waiver by the Owner, the Party nonetheless is legally compelled to disclose the Owner's Confidential and Proprietary Information to any court or tribunal or else would stand liable for contempt or suffer other censure or penalty, the Party may, without liability herein, disclose only that portion of the Owner's Confidential and Proprietary Information required to be disclosed, provided that the Party uses reasonable efforts to preserve the confidentiality of the Owner's Confidential and Proprietary Information, including, without limitation, by cooperating with the Owner to obtain an appropriate protective order or other administrative relief.

11. ESCROW

Tyler maintains an Escrow Agreement with an Escrow Agent under which Tyler places the source code of each major release. At End User's request, Tyler will add End User as a beneficiary on its Escrow Agreement upon payment in full of the License Fee. End User will be invoiced the annual beneficiary fee by Tyler and is solely responsible for maintaining its status as a beneficiary. Release of the escrowed material shall be governed by the terms of the Escrow Agreement and the use thereof shall be restricted by Sections 2.2 and 10 of this Agreement.

12. REPRESENTATIONS AND WARRANTIES

- 12.1. <u>Project Personnel</u>. All Tyler personnel utilized in connection with fulfilling its obligations pursuant to or arising from this Agreement shall be employees of Tyler or, if applicable, Tyler's subcontractor(s), shall be qualified to perform the tasks assigned them, and shall be in compliance with all applicable laws relating to employees generally, including, without limitation, immigration laws.
- 12.2. <u>Media Defects</u>. The media on which the Licensed Software is provided shall, at the time of delivery and installation, be free of Defects in material and workmanship.
- 12.3. <u>Pass-Through of Warranties</u>. Tyler hereby passes through the benefits of all third party warranties that it receives in connection with any product provided to End User.
- 12.4. <u>No Actions, Suits, or Proceedings</u>. There are no actions, suits, or proceedings, pending or, to the knowledge of Tyler, threatened, that shall have a material adverse effect on Tyler's ability to fulfill its obligations pursuant to or arising from this Agreement.
- 12.5. <u>Compliance with Laws</u>. In performing this Agreement, Tyler shall comply with all applicable material licenses, legal certifications, or inspections. Tyler shall also comply in all material respects with applicable federal, state, and local statutes, laws, ordinances, rules, and regulations.
- 12.6. Ownership. Tyler is a Delaware corporation that is listed for trading on the New York Stock Exchange. No director, officer, or 5% or more stockholder shall, during the course of this Agreement, receive or confer improper personal benefits or gains associated with the performance of the services outlined in this Agreement.
- 12.7. <u>Certain Business Practices</u>. Neither Tyler nor any of its principals is presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from participating in this Agreement by any federal department or agency. Tyler further represents and warrants that it is not listed on any local, End User, state or federal consolidated list of debarred, suspended, and ineligible contractors and grantees. No person (other than permanent employees of Tyler) has been engaged or retained by Tyler to solicit, procure, receive, accept, arrange, or secure this Agreement for any compensation, consideration, or value.

12.8. Illicit Code. The Licensed Software, when delivered and installed by Tyler, does not contain, and Tyler has not knowingly introduced through any media, any virus, worm, trap door, back door, bomb, bug, or other contaminant or disabling device, including, without limitation, any timer, clock, counter or other limiting routines, codes, commands, or instructions that may have the effect or be used to access, alter, delete, limit, control, damage, or disable any End User property.

EXCEPT AS SPECIFICALLY SET FORTH IN THIS SECTION 12 OR ELSEWHERE IN THIS AGREEMENT, TYLER DISCLAIMS ALL OTHER WARRANTIES, INCLUDING, WITHOUT LIMITATION, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE.

13. LIMITATION OF LIABILITY

TYLER'S LIABILITY TO END USER FOR DAMAGES ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT, WHETHER BASED ON A THEORY OF CONTRACT OR TORT, INCLUDING NEGLIGENCE AND STRICT LIABILITY, SHALL BE LIMITED TO: (A) PRIOR TO OPERATIONAL USE, THE LICENSE FEES PAID BY END USER; AND (B) AFTER OPERATIONAL USE, TYLER'S OBLIGATIONS AS SET FORTH IN THE TERMS AND CONDITIONS OF THE SOFTWARE MAINTENANCE AGREEMENT. THE FOREGOING LIMITATIONS DO NOT APPLY TO THE FOLLOWING CIRCUMSTANCES: (1) FRAUD; OR (2) BREACH OF SECTION 14.1 (CLAIMS FOR BODILY INJURY OR PROPERTY DAMAGE) OR SECTION 14.2 (INTELLECTUAL PROPERTY INFRINGEMENT).

IN NO EVENT SHALL TYLER BE LIABLE FOR INCIDENTAL, CONSEQUENTIAL, OR SPECIAL DAMAGES OF ANY KIND, INCLUDING, WITHOUT LIMITATION, LOST REVENUES OR PROFITS, OR LOSS OF BUSINESS OR LOSS OF DATA ARISING OUT OF THIS AGREEMENT, IRRESPECTIVE OF WHETHER THE PARTIES HAVE ADVANCE NOTICE OF THE POSSIBILITY OF SUCH DAMAGE.

14. INDEMNIFICATION

14.1. <u>General – Bodily Injury and Property Damage</u>. Notwithstanding any other provision of this Agreement, Tyler shall defend, indemnify, hold, and save harmless the Indemnified Parties from and against any and all Claims for bodily injury or property damage sustained by or asserted against End User arising out of, resulting from, or attributable to the negligent or willful misconduct of Tyler, its employees, subcontractors, representatives, and agents; provided, however, that Tyler shall not be liable herein to indemnify End User against liability for damages arising out of bodily injury to people or damage to property to the extent that such bodily injury or property damage is caused by or resulting from the actions, negligent or otherwise, of End User, its agents, contractors, subcontractors, or employees.

14.2. Intellectual Property.

(a) Notwithstanding any other provision of this Agreement, if any claim is asserted, or action or proceeding brought against Indemnified Parties that alleges that all or any part of the Licensed Property, in the form supplied, or modified by Tyler, or Indemnified Party's use thereof, infringes or misappropriates any United States intellectual property, intangible asset, or other proprietary right, title, or interest (including, without limitation, any copyright or patent or any trade secret right, title, or interest), or violates any other contract, license, grant, or other proprietary right of any third party, the Indemnified Party, upon its awareness, shall give Tyler prompt written notice thereof. Tyler shall defend, and hold the Indemnified Party harmless against, any such claim or action with counsel of Tyler's choice and at Tyler's expense and shall indemnify the Indemnified Party against any liability, damages, and costs resulting from such claim. Without waiving any rights pursuant to sovereign immunity, the Indemnified Party shall cooperate with and may monitor Tyler in the defense of any claim, action, or proceeding and shall, if appropriate, make employees available as Tyler may reasonably request with regard to such defense. This indemnity does not apply to the extent that such a claim is attributable to modifications to the Licensed Software made by the Indemnified Party, or any third party pursuant to the Indemnified Party's directions, or upon the unauthorized use of the Licensed Property by the Indemnified Party.

(b) If the Licensed Property becomes the subject of a claim of infringement or misappropriation of a copyright, patent, or trade secret or the violation of any other contractual or proprietary right of any third party, Tyler shall, at its sole cost and expense, select and provide one of the following remedies, which selection shall be in Tyler's sole discretion: (i) promptly replace the Licensed Property with a compatible, functionally equivalent, non-infringing; system; or (iii) promptly modify the Licensed Property to make it non-infringing; or (iii) promptly procure the right of End User to use the Licensed Property as intended.

15. TAXES

- 15.1. <u>Tax Exempt Status</u>. End User is a governmental tax-exempt entity and shall not be responsible for any taxes for any Licensed Property or services provided for herein, whether federal or state. The fees paid to Tyler pursuant to this Agreement are inclusive of any applicable sales, use, personal property, or other taxes attributable to periods on or after the Effective Date of this Agreement.
- 15.2. <u>Employee Tax Obligations</u>. Each Party accepts full and exclusive liability for the payment of any and all contributions or taxes for Social Security, Workers' Compensation Insurance, Unemployment Insurance, or Retirement Benefits, Pensions, or annutites now or hereafter imposed pursuant to or arising from any state or federal laws which are measured by the wages, salaries, or other remuneration paid to persons employed by such Party for work performed under this Agreement.

16. INSURANCE

Tyler shall provide, upon the written request of End User (which shall not be less than thirty (30) days after the Effective Date), proof of insurance for and maintain, at Tyler's sole cost and expense, the following insurance coverage issued with an insurance carrier with a Best Key rating of "A VII" or higher: (a) Industrial/Workers' Compensation Insurance protecting Tyler and End User from potential Tyler employee claims based upon job-related sickness, injury, or accident during performance of this Agreement; and (b) Comprehensive General Liability (including, without limitation, bodily injury and property damage) insurance with respect to Tyler's agents and vehicles assigned to perform the services herein with policy limits of not less than \$1,000,000 combined single limit per occurrence and \$2,000,000 in the aggregate. End User shall be named as an additional insured party and such notation shall appear on the certificate of insurance furnished by Tyler's insurance carrier.

17. TERM, SUSPENSION, AND TERMINATION

- 17.1. <u>Term</u>. The term of this Agreement (the "Term") shall commence on the Effective Date and shall continue until the sooner of: (a) Final Acceptance; or (b) the Agreement is terminated for Cause pursuant to Section 17.2.
- 17.2. <u>Termination for Cause</u>. Either Party may terminate this Agreement for Cause, provided that such Party follows the procedures set forth in this Section 17.2.
 - (a) For purposes of this Section, "Cause" means either:
 - a material breach of this Agreement, which has not been cured within ninety (90) days of the date such Party receives written notice of such breach;
 - (ii) the failure by End User to timely pay when due any fees and expenses owed to Tyler pursuant to this Agreement and any delinquent amounts remain outstanding for a period of thirty (30) days after Tyler provides written notice of its intent to terminate for failure to pay;
 - (iii) breach of Section 10; or
 - (iv) if Tyler becomes insolvent or bankrupt, or is the subject of any proceedings relating to its liquidation or insolvency or for the appointment of a receiver or similar officer for it, has a receiver of its assets or property appointed or makes an assignment for the benefit of all or substantially all of its creditors, or institutes or causes to be instituted any proceeding in bankruptcy or reorganization or rearrangement of its affairs.

- (b) No Party may terminate this Agreement under Section 17.2(a)(i) unless it cooperates in good faith with the alleged breaching Party during the cure period and complies in good faith with the dispute resolution procedures set forth in Section 18 following such period.
- (c) In the event either Party terminates this Agreement pursuant to this Section 17.2, each Party shall return all products, documentation, confidential information, and other information disclosed or otherwise delivered to the other Party prior to such termination and all revocable licenses granted herein shall terminate.
- 17.3. <u>Survival</u>. The following provisions shall survive after the Term of this Agreement: 1;3; 10; 11; 13; 14; 15; 17; 18; and 19.

18. DISPUTE RESOLUTION

Disputes arising out of, or relating to, this Agreement shall first be discussed by the Project Managers. Any dispute that cannot be resolved within five (5) Business Days at the Project Manager level (or such other date as agreed upon by the Project Managers) shall be referred to the individual reasonably designated by End User and Tyler's Vice President of Courts and Justice Systems Division assigned to End User's account ("Intermediary Dispute Level"). Any dispute that cannot be resolved in ten (10) Business Days at the Intermediary Dispute Level shall then be referred to End User's chief executive officer or other individual reasonably designated by End User and Tyler's President of Courts and Justice Systems Division ("Executive Dispute Level"), at such time and location reasonably designated by the Parties. Any negotiations pursuant to this Section 18 are confidential and shall be treated as compromise and settlement negotiations for purposes of the applicable rules of evidence. For any dispute that the Parties are unable to resolve through informal discussions or negotiations or pursuant to the dispute resolution and escalation procedures set forth in this Agreement, the matter shall be addressed through non-binding mediation under the Commercial Mediation Rules of the American Arbitration Association ("AAA"). A single mediator engaged in the practice of law, who is knowledgeable about subject matter of this Agreement, will conduct the mediation under the then current rules of the AAA. Any mediation under this Agreement shall be conducted in the county boundaries of End User. The costs of the mediation shall be borne equally between the Parties, except that each party shall bear its own attorneys fees. Nothing herein is intended to prevent either party from seeking any other remedy available at law including seeking redress in a court of competent jurisdiction. This provision shall survive the termination of this Agreement.

19. MISCELLANEOUS

- 19.1. <u>Assignment</u>. Neither Party may assign this Agreement or any of its respective rights or obligations herein to any third party without the express written consent of the other Party, which consent shall not be unreasonably withheld.
- 19.2. <u>Subcontractors</u>. Tyler shall not utilize any subcontractor(s) without the prior written consent of End User's Project Manager, which consent shall not be unreasonably withheld. The approval by End User of Tyler's right to use subcontractor(s) shall not waive or relieve Tyler from Tyler's obligations pursuant to this Agreement.
- 19.3. <u>Cumulative Remedies</u>. Except as specifically provided herein, no remedy made available herein is intended to be exclusive of any other remedy, and each and every remedy shall be cumulative and shall be in addition to every other remedy provided herein or available at law or in equity.
- 19.4. Notices. Except as otherwise expressly specified herein, all notices, requests or other communications shall be in writing and shall be deemed to have been given if delivered personally or mailed, by certified or registered mail, postage prepaid, return receipt requested, to the Parties at their respective addresses set forth on the signature page hereto, or at such other addresses as may be specified in writing by either of the Parties. All notices, requests, or communications shall be deemed effective upon personal delivery or three (3) days following deposit in the mail.
- 19.5. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

- 19.6. <u>Waiver</u>. The performance of any obligation required of a Party herein may be waived only by a written waiver signed by the other Party, which waiver shall be effective only with respect to the specific obligation described therein.
- 19.7. Entire Agreement. This Agreement constitutes the entire understanding and contract between the Parties and supersedes any and all prior or contemporaneous oral or written representations or communications with respect to the subject matter hereof.
- 19.8. <u>Amendment</u>. This Agreement shall not be modified, amended, or in any way altered except by an instrument in writing signed by the properly delegated authority of each Party. All amendments or modifications of this Agreement shall be binding upon the Parties despite any lack of consideration.
- 19.9. <u>Severability of Provisions</u>. In the event any provision hereof is found invalid or unenforceable pursuant to judicial decree, the remainder of this Agreement shall remain valid and enforceable according to its terms.
- 19.10. <u>Relationship of Parties</u>. The Parties intend that the relationship between the Parties created pursuant to or arising from this Agreement is that of an independent contractor only. Neither Party shall be considered an agent, representative, or employee of the other Party for any purpose.
- 19.11. <u>Governing Law</u>. Any dispute arising out of or relating to this Agreement or the breach thereof shall be governed by the laws of the state of the domicile of End User, without regard to or application of choice of law rules or principles.
- 19.12. <u>Audit</u>. Tyler shall maintain complete and accurate records of all work performed pursuant to and arising out of this Agreement. End User may, upon the written request of the Project Manager, audit any and all work or expense records of Tyler relating to materials and/or services provided herein. End User shall provide Tyler twenty-four hour notice of such audit or inspection. Tyler shall have the right to exclude from such inspection any Tyler Confidential and Proprietary Information not otherwise required to be provided to End User as a part of this Agreement. Tyler shall make such books and records available to End User during normal business hours. Any such audit shall be conducted at Tyler's principal

- place of business within the State of Texas during Tyler's normal business hours and at End User's sole expense.
- 19.13. No Third Party Beneficiaries. Nothing in this Agreement is intended to benefit, create any rights in, or otherwise vest any rights upon any third party.
- 19.14. <u>Contra Proferentem</u>. The doctrine of *contra proferentem* shall not apply to this Agreement. If an ambiguity exists in this Agreement, or in a specific provision, neither the Agreement nor the provision shall be construed against the Party who drafted the Agreement or provision.
- 19.15. Force Majeure. No Party to this Agreement shall be liable for delay or failure in the performance of its contractual obligations arising from any one or more events that are beyond its reasonable control, including, without limitation, acts of God, war, terrorism, and riot. Upon such delay or failure affecting one Party, that Party shall notify the other Party and use all reasonable efforts to cure or alleviate the cause of such delay or failure with a view to resuming performance of its contractual obligations as soon as practicable. Notwithstanding the foregoing, in every case the delay or failure to perform must be beyond the control and without the fault or negligence of the Party claiming excusable delay. Any performance times pursuant to or arising from this Agreement shall be considered extended for a period of time equivalent to the time lost because of any delay that is excusable herein.
- 19.16. Equitable Relief. Each Party covenants, represents, and warrants that any violation of this Agreement by such Party with respect to its respective obligations set forth in Sections 3.2 and 10 shall cause irreparable injury to the other Party and shall entitle the other Party to extraordinary and equitable relief by a court of competent jurisdiction, including, without limitation, temporary restraining orders and preliminary and permanent injunctions, without the necessity of posting bond or security.
- 19.17. Attorneys' Fees and Costs. If attorneys' fees or other costs are incurred by either Party to secure the performance of any obligations under this Agreement, or to establish damages for the breach thereof or to obtain any other appropriate relief, whether by way of prosecution or defense, the prevailing Party shall be entitled to recover from the other Party its reasonable attorneys' fees and costs incurred in connection therewith.

[Remainder of this page intentionally left blank]

(Exhibit B) Maintenance and Support Services Agreement

This Maintenance and Support Services Agreement (this "M&S Agreement") is made and entered into as of the Effective Date by and between Tyler and End User.

WHEREAS, Tyler and End User have entered into that certain Software License and Professional Services Agreement (the "License Agreement") pursuant to which, among other things, End User has acquired a license to Tyler's Licensed Software.

WHEREAS, End User desires Tyler to perform, and Tyler desires to perform, certain maintenance and support services related to the Licensed Software.

NOW, THEREFORE, in consideration of the promises contained herein, along with other good and valuable consideration, the receipt and sufficiency of which all parties acknowledge, the parties agree as follows:

1. CERTAIN DEFINITIONS

- 1.1. <u>Terms Not Defined</u>. Terms not otherwise defined herein shall have the meanings assigned to such terms in the License Agreement.
- 1.2. <u>Business Day means Monday through Friday, excluding Tyler Holidays.</u>
- 1.3. <u>Business Hour</u> means 7:00 a.m. to 7:00 p.m., Central Time during Business Days.
- 1.4. <u>Circumvention or Circumvention Procedures</u> means, as applied to a Documented Defect, a change in operating procedures whereby End User can reasonably avoid any deleterious effects of such Documented Defect. If a Circumvention Procedure is not acceptable to End User, End User may escalate this Defect as set forth in Section 3.11.
- 1.5. <u>Defect</u> means any bug, error, malfunction, or other defect in the Licensed Software caused by, arising from, or emanating from the reasonable control of Tyler that renders the Licensed Software in non-conformance with Tyler's then current published specifications.
- 1.6. <u>Documented Defect</u> means a Defect that End User documents for Tyler pursuant to Section 2.1.
- 1.7. <u>Essential Functionality</u> means any operational aspect of the Licensed Software that is required for immediate and ongoing business continuity by one or more users and which adversely impacts business in a crucial or critical manner.
- 1.8. <u>Non-Essential Functionality</u> means any operational aspect of the Licensed Software that will not interrupt business continuity or which will not adversely impact business in a crucial or critical manner.
- 1.9. <u>Legislative Change</u> means a refinement, enhancement, or other modification to the Licensed Software necessary to comply with final, statewide legislation or administrative regulation affecting in the same manner all clients in End User's state that are similarly situated with respect to the legislation and pertaining to: (a) existing reports, exports, or data exchanges; (b) new reports; (c) new data entry fields for state reporting; (d) new fee calculations; (e) new disposition templates; (f) new sentence templates; or (g) new citation templates. Legislative Changes do not include the expansion of End User's constitutional or operational responsibilities beyond those that exist as of the Effective Date.
 - 1.10. Effective Date has the meaning set forth in Section 8.1.
- 1.11. <u>Service Level 1 Defect</u> means a Documented Defect that causes (a) complete application failure or application unavailability; (b) application failure or unavailability in one or more of End User's remote locations; or (c) systemic loss of multiple essential system functions.
- 1.12. <u>Service Level 2 Defect</u> means a Documented Defect that causes (a) repeated, consistent failure of Essential Functionality affecting more than one user or (b) loss or corruption of data.

- 1.13. <u>Service Level 3 Defect</u> means a Service Level 1 Defect with an existing Circumvention Procedure, or a Service Level 2 Defect that affects only one user or for which there is an existing Circumvention Procedure.
- 1.14. <u>Service Level 4 Defect</u> means a Documented Defect that causes failure of Non-Essential Licensed Software functionality or a cosmetic or other Documented Defect that does not qualify as any other Service Level Defect
- 1.15. <u>Third Person Software</u> means all third party software required for the operation and use by End User of the Licensed Software consistent with the license granted to End User.
- 1.16. <u>Version Release</u> means new versions of <u>the Licensed</u>
 <u>Software</u> that contain technical improvements, functional enhancements, updates, extensions, and/or maintenance changes to the Licensed Software.
- 1.17. <u>Tyler Holidays</u> means one (1) day for a New Year's holiday, Good Friday, Memorial Day, a one (1) day holiday for Independence Day, Labor Day, Thanksgiving Day and the day after, and two (2) days during Christmas time. The exact date for any rolling holiday will be published on the Tyler website in advance of the date.

2. END USER RESPONSIBILITIES

2.1. <u>Documenting Defects</u>. End User must document all Defects in writing with sufficient information to recreate the Defect or otherwise clearly and convincingly document or evidence its occurrence, including, but not limited to, the operating environment, data set, user, or any other such information that Tyler may reasonably request. End User shall deliver such information to Tyler concurrently with its notification to Tyler of a Defect. End User shall use all reasonable efforts to eliminate any non-application related issues prior to its notification to Tyler of such Defect, including, but not limited to, issues related to the network, user training, End User-produced extensions, and data problems not caused by the Licensed Software. Any technical or other issue for which End User requests services, but which is not a Documented Defect, shall be treated as a request for other services and governed by Section 4.

2.2. Other End User Responsibilities. End User shall:

- (a) maintain all required Third Person Software to the release level compatible with the installed version(s) of the Licensed Software;
- (b) establish and maintain an internal help desk to be the central point of contact and communication between the end users and Tyler's support staff. In the event that the End User is unable to establish and maintain an internal help desk, End User may select up to twenty (20) "super users" who may contact Tyler's help desk.
- (c) provide training on the Licensed Software to its employees;
- (d) allow Tyler to install patches and other maintenance releases provided by Tyler;
- (e) allow remote access by Tyler to End User's servers and data via a Microsoft VPN connection or CISCO VPN client or other mutually agreeable protocol, provided, however, that End User acknowledges that failure to provide a timely and practical remote access method may negatively impact Tyler's ability to perform its responsibilities under this M&S Agreement;
- (f) implement and perform appropriate data backup and data recovery procedures related to the Licensed Software. In no event shall Tyler be held liable for any loss or other damage associated with the loss or destruction of any data related to the Licensed Software that is attributable to End User's failure to implement and perform such procedures on a timely and regular basis; and
- (g) provide onsite installation, new integration, training, and other responsibilities with respect to Version Releases as set forth in Section 5.

3. TYLER RESPONSIBILITIES - SUPPORT SERVICES

- 3.1. General Services for Reporting Production Documented Defects.
- (a) Tyler shall provide End User with procedures for contacting support staff during normal business hours (7:00 a.m. to 7:00 p.m., Central Time, Monday through Friday, excluding Tyler Holidays) for reporting Documented Defects. Tyler shall assist End User in the diagnosis of any Documented Defect, including the assigned Service Level and Tyler's tracking number.
- (b) For each reported Documented Defect, Tyler shall assign appropriate personnel to diagnose and correct the Documented Defect, and where appropriate, identify Circumvention Procedures. Tyler's initial response shall include an acknowledgement of notice of the Documented Defect, confirmation that Tyler has received sufficient information concerning the Documented Defect, and an action plan for resolving the Documented Defect and avoiding further deleterious consequences of the Documented Defect.
- 3.2. <u>Service Level 1 Defects</u>. Tyler shall provide an initial response to Service Level 1 Defects within one (1) Business Hour of receipt of the Documented Defect. Tyler shall use commercially reasonable efforts to resolve such Documented Defects or provide a Circumvention Procedure within one (1) Business Day. Tyler's responsibility for loss or corrupted data is limited to assisting End User in restoring its database to a known, accurate state.
- 3.3. <u>Service Level 2 Defects</u>. Tyler shall provide an initial response to Service Level 2 Defects within four (4) Business Hours of receipt of the Documented Defect. Tyler shall use commercially reasonable efforts to resolve such Documented Defects or provide a Circumvention Procedures within five (5) Business Days. Tyler's responsibility for loss or corrupted data is limited to assisting End User in restoring its database to a known, accurate state.
- 3.4. <u>Service Level 3 Defects</u>. Tyler shall provide an initial response to Service Level 3 Defects within one (1) Business Day of receipt of the Documented Defect. Tyler shall use commercially reasonable efforts to resolve such Documented Defect without the need for a Circumvention Procedure with the next published maintenance update or service pack, which shall occur at least quarterly. Tyler's responsibility for lost or corrupted data is limited to assisting End User in restoring its database to a known, accurate state.
- 3.5. <u>Service Level 4 Defects</u>. Tyler shall provide an initial response to Service Level 4 Defects within two (2) Business Days. Tyler shall use commercially reasonable efforts to resolve such Non-Essential Documented Defect within two version release cycles and a cosmetic or other Documented Defect that does not qualify as any other Service Level Defect with a future Version Release.
- 3.6. <u>Technical Server & Systems Support</u>. Tyler shall use commercially reasonable efforts to provide End User with technical support to assist End User with troubleshooting the loss of functionality of Licensed Software for reasons other than a Documented Defect. Tyler technical support shall be limited to:
 - (a) assisting the End User with isolating the source of Licensed Software failure due to systems-level hardware, Third Party Software, network, client-level hardware or peripherals;
 - (b) providing recommendations to End User regarding resolution of said non-defect failure(s); and
 - (c) providing End User with assistance on basic maintenance and administration of the Licensed Software environment, including basic data backup and restore procedures, deployment of Version Releases, and setup of supported peripheral devices for use with the Licensed Software
- 3.7. <u>24 X 7 Emergency Support</u>. Tyler shall provide the End User with procedures for contacting support staff after normal business hours for the limited purpose of reporting emergency application unavailability issues (such as a Level 1 Defect) within the Licensed Software. Tyler shall use commercially reasonable efforts to provide the response set forth in Section 3.2.

- 3.8. <u>Saturday Technical Support</u>. Tyler shall use commercially reasonable efforts to be available for one pre-scheduled Saturday of each month to allow assistance to End User IT staff. This option is available for the application of patches and full release upgrades as well as consulting with the End User IT staff for server maintenance and configuration for the licensed software environment.
- 3.9. <u>Base Version Level for Correction</u>. Tyler shall correct or otherwise cure Documented Defects to the current Version Release of Licensed Software made available to End User and either the immediately preceding Version Release or all Version Releases released to End User within the prior one (1) year, whichever is greater.
- 3.10. <u>Legislative Change Support</u>. Tyler will use its commercially reasonable efforts to implement Legislative Changes within the time frames set forth in the applicable legislation, but in any event in the next Version Release. Tyler's sole liability for implementing Legislative Changes in any calendar year shall be limited to the number of hours of analysis, development, post release data migration, and testing services, at Tyler's then current hourly rates, equal to not more than 20% of the total Annual Maintenance Fees for the Licensed Software paid by all clients with Legislative Change Support in End User's state during such calendar year; to the extent additional programming services are required, such services shall be billed to End User at End User's contractual billing rates, or at Tyler's then current hourly rates if no contractual billing rates are in effect. Notwithstanding the foregoing, End User shall be responsible for the cost of any other services required to implement a Legislative Change, including, without limitation, training, configuration, project management, or data conversion from external sources. Upon the mutual determination of the need for a Legislative Change that exceeds the limitations set forth above, Tyler shall provide End User with a written statement identifying the total number of hours that Tyler is liable for Legislative Change Support as calculated above plus a good faith estimate of the additional cost to End User. Such additional costs, if any, shall be prorated as a percentage of Annual Maintenance and Support Fees among all clients in End User's state with Legislative Change Support.
- 3.11. <u>Escalation Procedure</u>. If Tyler is unable to resolve any Service Level 1 or Service Level 2 Defect as provided in this Section 3, End User may immediately escalate the issue to End User's Project Manager or Designee and Tyler's Director of Client Services. Tyler and End User will use good faith reasonable efforts to meet, discuss, and agree upon a resolution plan for the affected Defect. If End User's Project Manager or Designee and Tyler's Director of Client Services cannot agree upon an acceptable resolution plan within 24 hours of such initial escalation, or such other reasonable time as the parties may agree, End User may further escalate the issue to End User's next Administrative Level and Tyler's Division Chief Operating Officer or Division President who shall have final authority to negotiate an acceptable resolution plan.

4. ADDITIONAL SUPPORT SERVICES

End User may request support services in addition to the standard maintenance offering (a "Service Request"). Such other support services may include, without limitation, services related to: (a) additional training; (b) technical assistance; (c) programming services; (d) installation of add-on components; and/or (e) business analysis. Tyler shall provide to End User a written response to the request which describes in detail the anticipated impact of the request on the existing Licensed Software, the time required to perform such services, an implementation plan, and a schedule of the fees related thereto. Fees for additional support services shall be billed by Tyler directly to End User and shall be invoiced monthly, which shall be due and payable in accordance with Section 7.2.

5. VERSION RELEASES

Tyler shall notify End User of the occurrence of a new Version Release and shall provide End User with such Version Releases for the Licensed Software. The delivery of each Version Release shall include a complete, installable copy of the Licensed Software, together with release notes and other appropriate documentation. Tyler will provide installation software and instruction for use by End User in installing new Version Releases provided, however, that if Tyler does not provide installation software and instructions, then Tyler shall provide installation assistance to End User at no additional cost. End User shall, at its own expense, be responsible for any configuration assistance, new integration, and training with respect to each Version Release.

6. THIRD PERSON SOFTWARE

6.1. <u>Notice of New Third Person Software</u>. Tyler shall provide End User with advanced notice of any mandated new Third Person Software

revision that shall be required to load a Version Release. Tyler shall use commercially reasonable efforts to minimize the need for End User to rely upon updates of Third Person Software.

- 6.2. <u>Tyler Certification</u>. At Tyler's expense, Tyler shall certify the compatibility of Third Person Software components used by the Licensed Software and maintain a list of supported Third Person Software release levels. Version Releases shall be certified to supported versions of all required Third Person Software. Tyler shall certify new releases of Third Person Software within a reasonable timeframe.
- 6.3. <u>Costs</u>. End User is responsible for all costs associated with installing and maintaining Third Person Software versions that are identified on Tyler's list of certified Third Person Software.
- 6.4. <u>Maintenance</u>. End User is responsible for maintaining software maintenance/update agreements with Third Person Software vendors at End User's expense. At the request of End User, Tyler shall participate with End User in discussions with Third Person Software providers on all software maintenance issues.

7. FEES

- 7.1. <u>Annual Maintenance Fee</u>. End User shall pay the annual maintenance and support fees as set forth on and in accordance with the timetables of Schedule 1 (the "Maintenance and Support Fees").
- 7.2. <u>Invoice and Payment</u>. Maintenance and Support Fees shall be invoiced annually in advance as set forth below:
 - (a) Tyler shall invoice the Texas Conference of Urban Counties (the "CUC") for maintenance and support fees incurred by an End User in accordance with the terms of the CUC Master Agreement and this M&S Agreement. Tyler shall use reasonable efforts to submit such invoices to the CUC sixty (60) days prior to the initial term of the M&S Agreement or the anniversary thereof, as applicable. The CUC shall be responsible for invoicing the End User and collecting payments from the applicable End User with respect to the maintenance and support fees. The CUC shall promptly, but in any event within thirty (30) days of receipt of License Fees and Maintenance and Support Fees, remit to Tyler all maintenance and support fees collected on behalf of an End User. In the event that the CUC fails to timely perform its invoice, collection, and remittance obligations under this section, Tyler shall have the right, in its sole discretion and upon written notice to the CUC and the affected End User, to invoice such End User directly for all future maintenance and support services.
 - (b) Any undisputed sum not paid when due shall bear interest calculated on an annual basis pursuant to Chapter 2251 of the Texas Government Code.
- 7.3. Each invoice shall include, at a minimum, the total invoiced amount and a reference to the specific items being invoiced under this M&S Agreement. Following receipt of a properly submitted invoice, End User shall pay amounts owed in accordance with chapter 2251, Texas Government Code. All payments shall be made in U.S. currency. Interest charges for any overdue payments shall be paid by End User in accordance with section 2251.025, Texas Government Code.

In the event that an error appears in an invoice submitted by Tyler and sent to End User by CUC, End User shall notify Tyler of the error not later than the twenty first (21st) day after the date End User receives the invoice. If the error is resolved in favor of Tyler, Tyler shall be entitled to receive interest on the unpaid balance of the invoice submitted by Tyler beginning on the date that the payment for the invoice became overdue. If the error is resolved in favor of the End User, Tyler shall submit a corrected invoice that must be paid in accordance with section 2251.021, Texas Government Code. The unpaid balance accrues interest as provided by chapter 2251, Texas Government Code, if the corrected invoice is not paid by the appropriate date.

7.4. <u>Maintenance on End User-Specific Customer Enhancements</u>. The annual Maintenance and Support Fee may be further increased by agreement of the Parties with respect to (a) maintenance and support of End User-Specific Customer Enhancements requested by End User and (b) material functional enhancements contained in new Version Releases that

are not merely technical improvements, updates, extensions and/or maintenance changes to the Licensed Software. End User will have the option to accept or decline any such material functional enhancement that would result in an increase in the Maintenance and Support Fee without affecting End User's entitlement to receive the remainder of any Version Release in which such enhancement is offered.

- 7.5. <u>Suspension of Services for Non-payment</u>. Tyler may suspend its performance of services hereunder during any period for which End User does not pay any undisputed Maintenance and Support Fees for a period of time exceeding sixty (60) days. Tyler shall promptly reinstate maintenance and support services upon receipt of payment of all undisputed Maintenance and Support Fees, including all such fees for the period(s) during which services were suspended.
- 7.6. <u>Fee Increases</u>. Any annual increase to maintenance and support fees may not exceed five percent (5%) of the preceding year's maintenance and support fees.

8. TERM AND TERMINATION

- 8.1. <u>Term.</u> This M&S Agreement shall commence in accordance with Schedule 1 of this M&S Agreement (the "Effective Date") and shall continue in effect for a period of one (1) year; provided, however, that at the end of such initial term, and on each subsequent anniversary of the Effective Date, the term shall automatically extend for an additional year unless a Party provides, at least ninety (90) days prior to the end of the then current term, written notice that it does not wish to extend the term or otherwise terminates the agreement as provided in this Section 8.
- 8.2. <u>Termination by End User at the End of a Term.</u> End User may terminate this M&S Agreement effective as of the end of the initial term or any subsequent term by giving not less than ninety (90) days' notice of its intent to terminate. End User may, at its option, reinstate maintenance by providing notice to Tyler and making payment of fifty percent (50%) of each year's Maintenance and Support Fees that would have been owed by End User during the lapsed period plus the Maintenance and Support Fees for the then upcoming maintenance year.
- 8.3. <u>Termination by End User for Cause</u>. End User may terminate this M&S Agreement for "cause" in accordance with this Section 8.3. For purposes of this Section, "cause" means a continuous or repeated failure to cure Documented Defects timely as provided in Section 3. In such event, End User shall deliver written notice of its intent to terminate along with a description in reasonable detail of the problems for which End User is invoking its right to terminate. Following such notice, Tyler shall have ninety (90) days to cure such problems. Following such ninety (90) day period, Tyler and End User shall meet to discuss any outstanding issues. In the event that "cause" still exists at the end of such period, then End User may terminate this Agreement. In the event of a termination under this subsection, Tyler shall return all monies paid to Tyler by End User under this M&S Agreement for the remainder of the then current maintenance period.

9. LIMITATION OF LIABILITY

TYLER'S LIABILITY TO END USER FOR DAMAGES ARISING OUT OF OR IN CONNECTION WITH THIS M&S AGREEMENT, WHETHER BASED ON A THEORY OF CONTRACT OR TORT, INCLUDING NEGLIGENCE AND STRICT LIABILITY, SHALL BE LIMITED TO FIXING DEFECTS IN ACCORDANCE WITH SECTION 3 OR AS OTHERWISE SET FORTH IN SECTION 8.3.

IN NO EVENT SHALL TYLER BE LIABLE TO END USER FOR INCIDENTAL, CONSEQUENTIAL, OR SPECIAL DAMAGES OF ANY KIND, INCLUDING, WITHOUT LIMITATION, LOST REVENUES OR PROFITS, OR LOSS OF BUSINESS OR LOSS OF DATA ARISING OUT OF THIS AGREEMENT, IRRESPECTIVE OF WHETHER THE PARTIES HAVE ADVANCE NOTICE OF THE POSSIBILITY OF SUCH DAMAGE.

10. DISPUTE RESOLUTION

The parties agree to use good faith, reasonable efforts to meet, discuss, and try to resolve any disputes arising out of, or relating to, this M&S Agreement for a period of sixty (60) days. The parties shall include in any such informal meetings persons with appropriate knowledge and authority, including, without limitation, End User's Information Technology Manager and Tyler's Support Manager. Any negotiations pursuant to this Section 10 are confidential and shall be treated as compromise and settlement negotiations for purposes of the applicable rules of evidence. For any dispute that the Parties are unable to resolve through informal

discussions or negotiations, the Parties shall have the right to pursue any remedies at law.

11. MISCELLANEOUS

- 11.1. <u>Assignment</u>. Neither party may assign this M&S Agreement or any of its respective rights or obligations herein to any third party without the express written consent of the other party.
- 11.2. Notices. Except as otherwise expressly specified herein, all notices, requests or other communications shall be in writing and shall be deemed to have been given if delivered personally or mailed, by certified or registered mail, postage prepaid, return receipt requested, to the parties at their respective addresses set forth on the signature page, or at such other addresses as may be specified in writing by either of the parties. All notices, requests, or communications shall be deemed effective upon personal delivery or three (3) days following deposit in the mail.
- 11.3. <u>Counterparts</u>. This M&S Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 11.4. Waiver. The performance of any obligation required of a party herein may be waived only by a written waiver signed by the other Parties, which waiver shall be effective only with respect to the specific obligation described therein.

- 11.5. Entire Agreement. This M&S Agreement constitutes the entire understanding and contract between the parties and supersedes any and all prior or contemporaneous oral or written representations or communications with respect to the subject matter hereof.
- 11.6. <u>Amendment</u>. This M&S Agreement shall not be modified, amended or in any way altered except by an instrument in writing signed by the properly delegated authority of each Party. All amendments or modifications of this M&S Agreement shall be binding upon the parties despite any lack of consideration.
- 11.7. <u>Governing Law</u>. Any dispute arising out of or relating to this M&S Agreement or the breach thereof shall be governed by the laws of the state of the domicile of End User, without regard to or application of choice of law rules or principles.
- 11.8. <u>No Third Party Beneficiaries</u>. Nothing in this M&S Agreement is intended to benefit, create any rights in, or otherwise vest any rights upon any third party.
- 11.9. <u>Contra Proferentem</u> The doctrine of contra proferentem shall not apply to this M&S Agreement. If an ambiguity exists in this Agreement, or in a specific provision, neither the Agreement nor the provision shall be construed against the party who drafted the M&S Agreement or provision.

[Remainder of this page intentionally left blank]



Software as a Service and Professional Services Agreement

This Software as a Service ("SaaS") and Professional Services Agreement (this "Agreement") is made and entered into by and between Tyler Technologies, Inc., a Delaware corporation ("Tyler"), and ______ County, Texas (the "Client").

Background

Client desires to engage Tyler to provide certain products and professional services, all on the terms and conditions set forth in this Agreement.

NOW, THEREFORE, in consideration of the mutual promises contained herein, along with other good and valuable consideration, the receipt and sufficiency of which all parties mutually acknowledge, Tyler and Client agree as follows:

- A. Tyler shall furnish the products and services described in this Agreement, and Client shall pay the prices set forth in this Agreement.
- B. This Agreement consists of this cover and signature page and the following attachments and exhibits attached hereto and to be attached throughout the Term of this Agreement, all of which are incorporated by reference herein:
 - Schedule 1. Investment Summary
 - Exhibit A. General Terms & Conditions
 - Exhibit B. Service Level Terms and Conditions
 - Schedule B-1. Application Availability Period Service Level

IN WITNESS WHEREOF, this Agreement has been executed by a duly authorized officer of each Party hereto to be effective as of the date last set forth below (the "Effective Date"):

TYLER TECHNOLOGIES, INC.

CLIENT

Signature:		Signature:	
Date:		Date:	
Name:		Name:	
Title:		. Title:	
Address:	6500 International Parkway, Suite 2000		
	Plano, Texas 75093		

(Schedule 1) Investment Summary

Software Licenses and ASP	Fee		
Term			
Start Date:			
SaaS Fee Payments	SaaS Fee A	Annual Amour	it
First payment due upon commencment of training and then	\$0	per Year	
annually in advance thereafter.			
Software Licenses			
Odyssey Online - Licensed Software	No. Users	Cost/User per Month	SaaS Fee (Annual)
Total Users			
	Total Ann	ual SaaS Fee	\$0
Embedded Third Party Software			
None			
Implementation Services	;		
Professional Services			
T&M Services			Cost
E		「&M Services vel Expenses _	\$0
Total I	mplementat	ion Services	\$0

(Exhibit A) General Terms and Conditions

1. CERTAIN DEFINITIONS

- 1.1. <u>Agreement</u> means this Software as a Service and Professional Services Agreement, including all exhibits attached hereto and to be attached throughout the Term of this Agreement, all of which are incorporated by reference herein.
- 1.2. <u>Business Day</u> means any day, Monday through Friday, excluding any Tyler holiday.
- 1.3. <u>Business Hour</u> means 7:00 a.m. to 7:00 p.m., Central Time during Business Days.
- 1.4. <u>Claims</u> mean any and all claims, liens, demands, damages, liability, actions, causes of action, losses, judgments, costs, and expenses, including reasonable attorneys' fees and expenses.
- 1.5. <u>Current Production Software Version</u> means the current production version of Tyler's software listed on the Investment Summary.
- 1.6. <u>Defect</u> means any bug, error, contaminate, malfunction, or other defect in the Licensed Software caused by, arising from, or emanating from the reasonable control of Tyler that renders the Licensed Software in non-conformance with Tyler's then current published specifications.
- 1.7. <u>Documentation</u> means the user's operating manuals and any other materials in any form or media provided by Tyler to the users of the Licensed Software.
- 1.8. <u>Embedded Third Party Software</u> means licensed third party software (other than Third Person Software) that is required to provide the functionality of the Licensed Software, which as of the date of this Agreement, consists of the software set forth on Schedule 1 labeled as "Embedded Third Party Software".
- 1.9. <u>Indemnified Parties</u> mean Client and each of its personnel, agents, successors, and permitted assigns.
- 1.10. <u>Investment Summary</u> means the summary of fees and services set forth on Schedule 1.
- 1.11. $\underline{\text{Licensed Property}}$ means the Licensed Software and the Documentation.
- 1.12. <u>Licensed Software</u> means: (a) the Current Production Software Version; (b) Embedded Third Party Software; and (c) any Local Enhancements.
- 1.13. <u>Local Enhancements</u> means any refinement, enhancement, or other customization to the Current Production Software Version to be developed by Tyler per the Investment Summary.
 - 1.14. Party means, individually, Tyler and Client.
- 1.15. <u>Project</u> means the delivery and license of the Licensed Property and the performance of all services to be provided by Tyler in accordance with the provisions of this Agreement.
- 1.16. <u>Project Manager</u> means the person designated by each Party who is responsible for the management of the Project.
- 1.17. <u>SaaS Fee</u> means the "Total Annual SaaS Fee" as set forth on the Investment Summary, which is due and payable as set forth in Section 3.1.
- 1.18. <u>Service Level Terms and Conditions</u> means the terms and conditions for Tyler's maintenance and support of the Licensed Software, which are set forth in Exhibit B.
 - 1.19. T&M means time and materials.
- 1.20. <u>Third Person Hardware</u> means the workstations and other hardware to be leased, purchased, or otherwise acquired by Client from a third party that is minimally required to operate the Licensed Software and

such other hardware that Client has actually leased, purchased or otherwise acquired and/or may be minimally required in the future to operate the Licensed Software.

- 1.21. <u>Third Person Software</u> means the operating systems and other software to be licensed, purchased, or otherwise acquired by Client from a third party that is minimally required to operate the Licensed Software and such operating systems and other software that Client has actually licensed, purchased, or otherwise acquired and/or may be minimally required in the future to operate the Licensed Software.
- 1.22. Confidential and Proprietary Information means all information in any form relating to, used in, or arising out of a Party's operations and held by, owned, licensed, or otherwise possessed by such Party (the "Owner") (whether held by, owned, licensed, possessed, or otherwise existing in, on or about the Owner's premises or the other Party's offices, residence(s), or facilities and regardless of how such information came into being, as well as regardless of who created, generated or gathered the information), including, without limitation, all information contained in, embodied in (in any media whatsoever) or relating to the Owner's inventions, ideas, creations, works of authorship, business documents, licenses, correspondence, operations, manuals, performance manuals, operating data, projections, bulletins, customer lists and data, sales data, cost data, profit data, financial statements, strategic planning data, financial planning data, designs, logos, proposed trademarks or service marks, test results, product or service literature, product or service concepts, process data, specification data, know how, software, databases, database layouts, design documents, release notes, algorithms, source code, screen shots. and other research and development information and data. Notwithstanding the foregoing, Confidential and Proprietary Information does not include information that: (a) becomes public other than as a result of a disclosure by the other Party in breach hereof; (b) becomes available to the other Party on a non-confidential basis from a source other than the Owner, which is not prohibited from disclosing such information by obligation to the Owner; (c) is known by the other Party prior to its receipt from the Owner without any obligation of confidentiality with respect thereto; or (d) is developed by the other Party independently of any disclosures made by the Owner.
- 1.23. <u>Users</u> means individuals who are employed and authorized by Client to use the Licensed Property, and who have been supplied with user identifications and passwords by Client (or by Tyler at Client's request).

2. TITLE AND LICENSE

2.1. <u>License Grant</u>. In consideration for the SaaS Fee, which shall be due and payable as set forth in Section 3, Tyler hereby grants to Client a limited, non-exclusive, revocable and non-transferable license (and sublicense with respect to the Embedded Third Party Software) to use the Licensed Property for Client's internal administration, operation, and/or conduct of Client's business operations by the number of Users as set forth on the Investment Summary.

TYLER HAS THE RIGHT TO REVOKE THIS LICENSE IF CLIENT TERMINATES, CANCELS OR FAILS TO RENEW THIS AGREEMENT. TYLER HAS THE RIGHT TO UNILATERALLY REVOKE THIS LICENSE AND DENY CLIENT ACCESS TO THE LICENSED PROPERTY IF CLIENT FAILS TO REMIT ANY REQUIRED FEES WITHIN THIRTY DAYS OF THE DATE SUCH FEES BECOME DUE AS SET FORTH HEREIN AND SUCH AMOUNTS REMAIN OUTSTANDING FOR A PERIOD OF THIRTY DAYS FOLLOWING TYLER'S WRITTEN NOTICE OF ITS INTENT TO REVOKE THE LICENSE.

2.2. <u>User Licenses</u>. Unless otherwise specified on the Investment Summary: (a) the Licensed Property is purchased as User subscriptions and may be accessed by no more than the specified number of Users; (b) additional User subscriptions may be added during the Term at the same pricing as that for the pre-existing subscriptions, prorated for the remainder of the Term in effect at the time the additional User subscriptions are added; and (c) the added User subscriptions shall terminate on the same date as the pre-existing subscriptions. User subscriptions are for designated Users and cannot be shared or used by more than one User; provided, however, that User subscriptions may be reassigned to new Users replacing former Users who no longer require ongoing use of the Licensed Property.

- 2.3. Restrictions. Unless otherwise expressly set forth in this Agreement, Client shall not (a) reverse engineer, de-compile, or disassemble any portion of the Licensed Software or (b) sublicense, transfer, rent, or lease the Licensed Software or its usage. To the extent Client employs contractors, subcontractors, or other third parties to assist in the Project, Client shall obtain from such third parties an executed Tyler confidentiality agreement prior to such parties being permitted access to Tyler Confidential and Proprietary Information.
- 2.4. Embedded Third Party Software. The license grant set forth in Section 2.1 includes the right to use any Embedded Third Party Software; provided, however, that such access to and use of such Embedded Third Party Software shall be according to such terms, conditions, and licenses as are imposed by the manufacturers and/or third party licensors of such Embedded Third Party Software. All such Embedded Third Party Software shall be included in the SaaS Fee. Tyler shall pass through to Client any and all warranties granted to Tyler by the owners, licensors, and/or distributors of such Embedded Third Party Software.

2.5. <u>Title</u>.

- (a) Tyler represents and warrants that it is the owner of all right, title, and interest in and to the Licensed Software (other than Embedded Third Party Software) and all components and copies thereof. Nothing in this Agreement shall be deemed to vest in Client any ownership or intellectual property rights in and to Tyler's intellectual property (including, without limitation, Tyler Confidential and Proprietary Information), any components and copies thereof, or any derivative works based thereon prepared by Tyler.
 - (b) All training materials shall be the sole property of Tyler.
- (c) All Client data shall remain the property of Client. Tyler shall not use Client data other than in connection with providing the services pursuant to this Agreement.

3. FEES AND INVOICING

- 3.1. <u>SaaS Fee</u>. Invoice and payment of the SaaS Fee shall be governed by Section 3.5.
- 3.2. <u>Professional Services Charges</u>. T&M charges for all professional services to be performed hereunder shall be invoiced and paid by Client in accordance with Section 3.4.
- 3.3. <u>Expenses</u>. Client shall reimburse Tyler for travel, lodging, and food expenses actually and reasonably incurred by Tyler in performing its professional services herein in accordance with Section 3.4.
- 3.4. <u>Invoice and Payment for Professional Services</u>. Tyler shall invoice Client for professional services and associated expenses herein on a monthly basis. Each invoice shall state the total invoiced amount and shall be accompanied by a reasonably detailed itemization of services and expenses. Following receipt of a properly submitted invoice, Client shall pay amounts owing therein thirty (30) days in arrears. All payments shall be made in U.S. currency
- 3.5. <u>Invoice and Payment of Annual SaaS Fees</u>. Annual SaaS Fees, as set forth on the Investment Summary, shall be invoiced and paid as set forth below:
 - (a) Tyler shall invoice the Texas Conference of Urban Counties (the "CUC") for Annual SaaS Fees incurred by the Client in accordance with the terms of the CUC Master Agreement. Tyler shall use reasonable efforts to submit such invoices to the CUC sixty (60) days prior to the initial SaaS due date or the anniversary thereof, as applicable.
 - (b) The CUC shall be responsible for invoicing Client and collecting payments from Client with respect to the Annual SaaS Fees. The CUC shall promptly, but in any event within thirty (30) days of receipt of Annual SaaS Fees, remit to Tyler all Annual SaaS Fees collected on behalf of Client. In the event that the CUC fails to timely perform its invoice, collection, and remittance obligations under this section, Tyler shall have the right, in its sole discretion and upon written

- notice to the CUC Client, to invoice the Client directly for all future Annual SaaS Fees.
- 3.6. Each invoice shall include, at a minimum, the total invoiced amount and a reference to the specific items being invoiced under this Agreement. Following receipt of a properly submitted invoice, Client shall pay amounts owed in accordance with Chapter 2251, Texas Government Code. All payments shall be made in U.S. currency.
- 3.7. Any undisputed sum not paid when due shall bear interest calculated on an annual basis pursuant to Chapter 2251 of the Texas Government Code.
- 3.8. Annual increases of SaaS Fees, if any, beyond the initial term in the Investment Summary shall not exceed five percent (5%) of the preceding year's SaaS Fees.

4. PROJECT IMPLEMENTATION

- 4.1. <u>Professional Services</u>. Attached hereto as Schedule 1 is Tyler's good faith estimate of the hours and fees associated with the services to be performed by Tyler for Client, including travel time by Tyler's personnel from Tyler's place of business to and from Client's place of business, and for which Client shall pay on a T&M basis. Additional services requested by Client which are beyond those hours detailed in Schedule 1 will be billed at Tyler's then current services rates.
- 4.2. Office Space. Client shall, at its sole expense, provide reasonable access to office space, telephone access, network access (including providing Tyler reasonable access to a secure virtual private network connection or other comparable connection for use by Tyler from time to time on a non-dedicated basis), Internet connections, and such other facilities as may be reasonably requested by Tyler for use by Tyler personnel for the purpose of performing this Agreement.
- 4.3. <u>Third Person Hardware and Third Person Software</u>. Client shall be responsible to purchase, install, and configure all Third Person Hardware and Third Person Software. Tyler shall have no liability for defects in the Third Person Hardware or Third Person Software.
- 4.4. <u>Cooperation</u>. Client acknowledges that the implementation of the Project is a cooperative process requiring the time and resources of Client personnel. Client shall, and shall cause its personnel to, use all reasonable efforts to cooperate with and assist Tyler as may be reasonably required to timely implement the Project, including, without limitation, providing reasonable information regarding its operations and reasonable access to its facilities. Tyler shall not be liable for failure to timely implement the Project when such failure is due to Force Majeure (as identified in Section17.14) or to the failure by Client personnel to provide such cooperation and assistance (either through action or omission).

5. INSTALLATION OF THE LICENSED SOFTWARE

Tyler shall use commercially reasonable efforts to promptly install the Licensed Software on Tyler's servers in accordance with a mutually agreed upon timetable. Upon installation, Tyler shall conduct its standard diagnostic evaluation to determine that the Licensed Software is properly installed, and upon completion, shall deliver written instructions for accessing the Licensed Software to Client.

6. VERIFICATION OF THE LICENSED SOFTWARE

- 6.1. <u>Verification Procedure.</u> Upon installation of the Licensed Software, Tyler shall perform its standard test procedures and shall certify to Client that the Licensed Software is in substantial conformance with Tyler's then current published specifications and is ready for Client's use. A copy of the certification shall be provided to the CUC. In the event Tyler cannot so certify, Client's right and remedy against Tyler shall be to require Tyler to correct the cause thereof or exercise its rights under Section 15.3.
- 6.2. <u>Certification Final</u>. Tyler's certification that the Licensed Software substantially complies with the then current published specifications shall be final and conclusive, except for latent defect, fraud, and such gross mistakes that amount to fraud. If Client disputes the certification for any reason, it shall notify Tyler in writing within fifteen (15) business days of receipt of the certification from Tyler, and provide a copy of such notification to the CUC.

6.3. <u>Use</u>. Notwithstanding anything to the contrary herein, Client's use of the Licensed Software for its intended purpose shall constitute Tyler's verification of the software products, without exception and for all purposes.

7. TRAINING

To the extent that training services are included in Schedule 1, Tyler shall train Client in accordance with a mutually agreeable training plan. The training plan shall outline the training required for personnel to operate the Licensed Software. Tyler shall provide Client personnel with only the number of hours of training for the respective portions of the Licensed Software as set forth in Schedule 1. Training shall be provided at Client's principal place of business or other site selected by Client. Training shall be performed according to the training plan, but in any event shall be "handson" using production-ready versions of the Licensed Software. The courses shall train Client's employees or agents in a manner to provide basic end user training. Client shall be responsible for providing an adequately equipped training facility to operate the Licensed Software.

8. MAINTENANCE AND SUPPORT SERVICES

- 8.1. <u>Service Level Terms and Conditions</u>. Upon Tyler's certification of the Licensed Software or Client's use, whichever occurs first, Tyler shall provide Client with the maintenance and support services for the Licensed Software as set forth in Exhibit B.
- 8.2. <u>Responsibilities of Client.</u> In addition to the other responsibilities set forth herein, Client shall: (a) provide all training of its personnel; (b) collect, prepare, and enter all data, other than such data converted by Tyler pursuant to this Agreement, necessary for the day-to-day operations of the Licensed Software; (c) retain separate copies of all conversion data delivered to Tyler; (d) provide end user workstations that conform to Tyler's minimum requirements; and (e) provide the requisite networks.

9. CONFIDENTIAL AND PROPRIETARY INFORMATION

- 9.1. Protection of Confidential and Proprietary Information. Each Party shall not disclose, disseminate, transmit, publish, distribute, make available, or otherwise convey the other Party's Confidential and Proprietary Information, and each Party shall not use, make, sell, or otherwise exploit any such other Party's Confidential and Proprietary Information for any purpose other than the performance of this Agreement, without the other Party's written consent, except: (a) as may be required by law, regulation, judicial, or administrative process; or (b) as required in litigation pertaining to this Agreement, provided that the other Party is given advance notice of such intended disclosure in order to permit it the opportunity to seek a protective order. The Parties shall ensure that all individuals assigned to perform services herein shall abide by the terms of this Section 9.1 and a Party shall be responsible for breaches by such persons acting by or for such Party.
- 9.2. <u>Judicial and Administrative Proceedings</u>. If a Party is requested or required (by oral questions, interrogatories, requests for information or documents in legal proceedings, subpoena, civil investigative demand, requirements of the Texas Public Information Act, or other similar process) to disclose any Confidential and Proprietary Information of the other Party (the "Owner"), such Party shall provide the Owner with prompt written notice of such request or requirement so that the Owner may seek protective orders or other appropriate remedies and/or waive compliance with the provisions of this Agreement. If, in the absence of a protective order or other remedy or the receipt of a waiver by the Owner, the Party nonetheless is legally compelled to disclose the Owner's Confidential and Proprietary Information or else would stand liable for contempt or suffer other censure or penalty, the Party may, without liability herein, disclose only that portion of the Owner's Confidential and Proprietary Information required to be disclosed, provided that the Party uses reasonable efforts to preserve the confidentiality of the Owner's Confidential and Proprietary Information, including, without limitation, by cooperating with the Owner to obtain an appropriate protective order or other administrative relief.

10. REPRESENTATIONS AND WARRANTIES

10.1. <u>Project Personnel.</u> All Tyler personnel utilized in connection with fulfilling its obligations pursuant to or arising from this Agreement shall be employees of Tyler or, if applicable, Tyler's subcontractor(s), shall be qualified to perform the tasks assigned them, and shall be in compliance with all applicable laws relating to employees generally, including, without limitation, immigration laws.

- 10.2. <u>Pass-Through of Warranties</u>. Tyler hereby passes through the benefits of all third party warranties that it receives in connection with any product provided to Client.
- 10.3. No Actions, Suits, or Proceedings. There are no actions, suits, or proceedings pending or, to the knowledge of Tyler, threatened, that shall have a material adverse effect on Tyler's ability to fulfill its obligations pursuant to or arising from this Agreement.
- 10.4. Compliance with Laws. In performing this Agreement, Tyler shall comply with all applicable material licenses, legal certifications, or inspections. Tyler shall also comply in all material respects with applicable federal, state, and local statutes, laws, ordinances, rules, and regulations.
- 10.5. Ownership. Tyler is a Delaware corporation that is listed for trading on the New York Stock Exchange. No director, officer, or 5% or more stockholder shall, during the course of this Agreement, receive or confer improper personal benefits or gains associated with the performance of the services outlined in this Agreement.
- 10.6. <u>Certain Business Practices</u>. Neither Tyler nor any of its principals is presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from participating in this Agreement by any federal department or agency. Tyler further represents and warrants that it is not listed on any local, Client, state or federal consolidated list of debarred, suspended, and ineligible contractors and grantees. No person (other than permanent employees of Tyler) has been engaged or retained by Tyler to solicit, procure, receive, accept, arrange, or secure this Agreement for any compensation, consideration, or value.

EXCEPT AS SPECIFICALLY SET FORTH IN THIS SECTION 10 OR ELSEWHERE IN THIS AGREEMENT, TYLER DISCLAIMS ALL OTHER WARRANTIES, INCLUDING, WITHOUT LIMITATION, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE.

11. LIMITATION OF LIABILITY

TYLER'S LIABILITY TO CLIENT FOR DAMAGES ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT, WHETHER BASED ON A THEORY OF CONTRACT OR TORT, INCLUDING NEGLIGENCE AND STRICT LIABILITY, SHALL BE LIMITED TO: (A) PRIOR TO TYLER'S CERTIFICATION OF THE LICENSED SOFTWARE AND CLIENT'S USE THEREOF, THE SAAS FEES PAID BY CLIENT, IF ANY; AND (B) AFTER TYLER'S CERTIFICATION OF THE LICENSED SOFTWARE AND CLIENT'S USE THEREOF, FIXING DEFECTS IN ACCORDANCE WITH EXHIBIT B. THE FOREGOING LIMITATIONS DO NOT APPLY TO THE FOLLOWING CIRCUMSTANCES: (1) FRAUD; OR (2) BREACH OF SECTION 12.1 (CLAIMS FOR BODILY INJURY OR PROPERTY DAMAGE) OR SECTION 12.2 (INTELLECTUAL PROPERTY INFRINGEMENT).

IN NO EVENT SHALL TYLER BE LIABLE FOR INCIDENTAL, CONSEQUENTIAL, OR SPECIAL DAMAGES OF ANY KIND, INCLUDING, WITHOUT LIMITATION, LOST REVENUES OR PROFITS, LOSS OF BUSINESS, OR LOSS OR CORRUPTION OF DATA ARISING OUT OF THIS AGREEMENT, IRRESPECTIVE OF WHETHER THE PARTIES HAVE ADVANCE NOTICE OF THE POSSIBILITY OF SUCH DAMAGE.

12. INDEMNIFICATION

12.1. <u>General – Bodily Injury and Property Damage</u>. Notwithstanding any other provision of this Agreement, Tyler shall defend, indemnify, hold, and save hamless the Indemnified Parties from and against any and all Claims for bodily injury or property damage sustained by or asserted against Client arising out of, resulting from, or attributable to the negligent or willful misconduct of Tyler, its employees, subcontractors, representatives, and agents; provided, however, that Tyler shall not be liable herein to indemnify Client against liability for damages arising out of bodily injury to people or damage to property to the extent that such bodily injury or property damage is caused by or resulting from the actions, negligent or otherwise, of Client, its agents, contractors, subcontractors, or employees.

12.2. Intellectual Property.

- (a) Notwithstanding any other provision of this Agreement, if any claim is asserted, or action or proceeding brought against Indemnified Parties that alleges that all or any part of the Licensed Property, in the form supplied, or modified by Tyler, or an Indemnified Party's use thereof, infringes or misappropriates any United States intellectual property, intangible asset, or other proprietary right, title, or interest (including the property). interest (including, without limitation, any copyright or patent or any trade secret right, title, or interest), or violates any other contract, license, grant, or other proprietary right of any third party, the Indemnified Party, upon its awareness, shall give Tyler prompt written notice thereof. Tyler shall defend, and hold Indemnified Party harmless against, any such claim or action with counsel of Tyler's choice and at Tyler's expense and shall indemnify Indemnified Party against any liability, damages, and costs resulting from such claim. Without waiving any rights pursuant to sovereign immunity, Indemnified Party shall cooperate with and may monitor Tyler in the defense of any claim, action, or proceeding and shall, if appropriate, make employees available as Tyler may reasonably request with regard to such defense. This indemnity does not apply to the extent that such a claim is attributable to modifications to the Licensed Property made by the Indemnified Party, or any third party pursuant to Indemnified Party's directions, or upon the unauthorized use of the Licensed Property by the Indemnified Party.
- (b) If the Licensed Property becomes the subject of a claim of infringement or misappropriation of a copyright, patent, or trade secret or the violation of any other contractual or proprietary right of any third party, Tyler shall, at its sole cost and expense, select and provide one of the following remedies, which selection shall be in Tyler's sole discretion: (i) promptly replace the Licensed Property with a compatible, functionally equivalent, non-infringing system; or (ii) promptly modify the Licensed Property to make it non-infringing; or (iii) promptly procure the right of Client to use the Licensed Property as intended.

13. TAXES

- 13.1. <u>Tax Exempt Status</u>. Client represents and warrants that it is a governmental tax-exempt entity and shall not be responsible for any taxes for any Licensed Property or services provided for herein, whether federal or state. The fees paid to Tyler pursuant to this Agreement are inclusive of any applicable sales, use, personal property, or other taxes attributable to periods on or after the Effective Date of this Agreement.
- 13.2. Employee Tax Obligations. Each Party accepts full and exclusive liability for the payment of any and all contributions or taxes for Social Security, Workers' Compensation Insurance, Unemployment Insurance, or Retirement Benefits, Pensions, or annuities now or hereafter imposed pursuant to or arising from any state or federal laws which are measured by the wages, salaries, or other remuneration paid to persons employed by such Party for work performed under this Agreement.

14. INSURANCE

Tyler shall provide, upon the written request of Client (which shall not be less than thirty (30) days after the Effective Date), proof of insurance for and maintain, at Tyler's sole cost and expense, the following insurance coverage issued with an insurance carrier with a Best Key rating of "A VII" or higher: (a) Industrial/Workers' Compensation Insurance protecting Tyler and Client from potential Tyler employee claims based upon job-related sickness, injury, or accident during performance of this Agreement; and (b) Comprehensive General Liability (including, without limitation, bodily injury and property damage) insurance with respect to Tyler's agents and vehicles assigned to perform the services herein with policy limits of not less than \$1,000,000 combined single limit per occurrence and \$2,000,000 in the aggregate. Client shall be named as an additional insured party and such notation shall appear on the certificate of insurance furnished by Tyler's insurance carrier.

15. TERM, SUSPENSION, AND TERMINATION

15.1. <u>Term.</u> The term of this Agreement (the "Term") shall commence on the start date specified in the Investment Summary and continue for the subscription term specified therein. Except as otherwise specified in the Investment Summary, the term of this Agreement, and the corresponding payment of all SaaS Fees, shall automatically renew for additional periods equal to the then expiring subscription term or one year (whichever is longer), unless either party gives the other notice of non-renewal at least 30 days before the end of the relevant subscription term. The per-unit pricing during any such renewal term shall be the same as that during the prior term unless Tyler has given Client written notice of any pricing change at

least 90 days before the end of such prior term, in which case the change in pricing shall be effective upon renewal and thereafter.

- 15.2. <u>Early Termination by Client</u>. This Agreement may be terminated by Client prior to the end of the then current term by Client providing Tyler with (a) ninety (90) days written notice of its intent to terminate, and (b) payment equal to the lesser of (i) 50% of the SaaS Fees still due for the remainder of the then current term or (ii) one year's SaaS Fees.
- 15.3. <u>Termination for Cause</u>. Either Party may terminate this Agreement for Cause, provided that such Party follows the procedures set forth in this Section 15.3.
 - (a) For purposes of this Section, "Cause" means either:
 - (i) a material breach of this Agreement, which has not been cured within ninety (90) days of the date such Party receives written notice of such breach;
 - (ii) the failure by Client to timely pay when due any fees and expenses owed to Tyler pursuant to this Agreement and any delinquent amounts remain outstanding for a period of thirty (30) days after Tyler provides written notice of its intent to terminate for failure to pay;
 - (iii) breach of Section 9; or
 - (iv) if Tyler becomes insolvent or bankrupt, or is the subject of any proceedings relating to its liquidation or insolvency or for the appointment of a receiver or similar officer for it, has a receiver of its assets or property appointed or makes an assignment for the benefit of all or substantially all of its creditors, or institutes or causes to be instituted any proceeding in bankruptcy or reorganization or rearrangement of its affairs.
 - (b) No Party may terminate this Agreement under Section 15.3(a)(i) unless it cooperates in good faith with the alleged breaching Party during the cure period and complies in good faith with the dispute resolution procedures set forth in Section 16 following such period.
- (c) Upon any termination for Cause by Client, Tyler shall refund any prepaid SaaS Fees covering the remainder of the Term after the effective date of termination. Upon any termination for Cause by Tyler, Client shall pay Tyler the lesser of (a) any unpaid SaaS Fees covering 50% of remainder of the Term after the effective date of termination; or (b) one year's SaaS Fees. In no event shall any termination relieve Client of the obligation to pay any fees payable to Tyler for the period prior to the effective date of termination.
- 15.4. Effect of Termination. Upon termination of this Agreement for any reason: (a) the licenses provided hereunder shall automatically terminate as of the effective date of the termination and Client's access to the licensed applications shall be denied; (b) subject to payment of all amounts due hereunder, and upon written request, Tyler will provide to Client such contents of the database that are owned by Client, as such contents exist on the date of termination, in a standard industry data file format within five business days; and (c) upon written request, Client shall return all documentation, products, Tyler Confidential and Proprietary Information, and other information disclosed or otherwise delivered to Client by Tyler.
- 15.5. <u>Survival</u>. The following provisions shall survive after the Term of this Agreement: 1; 2; 9; 11; 12; 13; 15; 16; and 17.

16. DISPUTE RESOLUTION

Disputes arising out of, or relating to, this Agreement shall first be discussed by the Project Managers. Any dispute that cannot be resolved within five (5) Business Days at the Project Manager level (or such other date as agreed upon by the Project Managers) shall be referred to the individual reasonably designated by Client and Tyler's Vice President of Courts and Justice Systems Division assigned to Client's account ("Intermediary Dispute Level"). Any dispute that cannot be resolved in ten (10) Business Days at the Intermediary Dispute Level shall then be referred to Client's chief executive officer or other individual reasonably designated by Client and Tyler's President of Courts and Justice Systems Division

("Executive Dispute Level"), at such time and location reasonably designated by the Parties. Any negotiations pursuant to this Section 16 are confidential and shall be treated as compromise and settlement negotiations for purposes of the applicable rules of evidence. For any dispute that the Parties are unable to resolve through informal discussions or negotiations or pursuant to the dispute resolution and escalation procedures set forth in this Agreement, the matter shall be addressed through non-binding mediation under the Commercial Mediation Rules of the American Arbitration Association ("AAA"). A single mediator engaged in the practice of law, who is knowledgeable about subject matter of this Agreement, will conduct the mediation under the then current rules of the AAA. Any mediation under this Agreement shall be conducted in the county boundaries of Client. The costs of the mediation shall be borne equally between the Parties, except that each party shall bear its own attorneys fees. Nothing herein is intended to prevent either party from seeking any other remedy available at law including seeking redress in a court of competent jurisdiction. This provision shall survive the termination of this Agreement.. Any such arbitration proceeding shall be governed by the rules of the American Arbitration Association. Any award or other relief granted by the arbitrators may be enforced in any court of competent jurisdiction. The foregoing shall not apply to claims for equitable relief under Section 9.

17. MISCELLANEOUS

- 17.1. <u>Assignment</u>. Neither Party may assign this Agreement or any of its respective rights or obligations herein to any third party without the express written consent of the other Party, which consent shall not be unreasonably withheld.
- 17.2. <u>Cumulative Remedies</u>. Except as specifically provided herein, no remedy made available herein is intended to be exclusive of any other remedy, and each and every remedy shall be cumulative and shall be in addition to every other remedy provided herein or available at law or in equity.
- 17.3. Notices. Except as otherwise expressly specified herein, all notices, requests or other communications shall be in writing and shall be deemed to have been given if delivered personally or mailed, by certified or registered mail, postage prepaid, return receipt requested, to the Parties at their respective addresses set forth on the signature page hereto, or at such other addresses as may be specified in writing by either of the Parties. All notices, requests, or communications shall be deemed effective upon personal delivery or three (3) days following deposit in the mail.
- 17.4. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 17.5. <u>Waiver</u>. The performance of any obligation required of a Party herein may be waived only by a written waiver signed by the other Party, which waiver shall be effective only with respect to the specific obligation described therein.
- 17.6. <u>Entire Agreement</u>. This Agreement constitutes the entire understanding and contract between the Parties and supersedes any and all prior or contemporaneous oral or written representations or communications with respect to the subject matter hereof.
- 17.7. <u>Amendment</u>. This Agreement shall not be modified, amended, or in any way altered except by an instrument in writing signed by the properly delegated authority of each Party. All amendments or modifications of this Agreement shall be binding upon the Parties despite any lack of consideration.

- 17.8. <u>Severability of Provisions</u>. In the event any provision hereof is found invalid or unenforceable pursuant to judicial decree, the remainder of this Agreement shall remain valid and enforceable according to its terms.
- 17.9. <u>Relationship of Parties</u>. The Parties intend that the relationship between the Parties created pursuant to or arising from this Agreement is that of an independent contractor only. Neither Party shall be considered an agent, representative, or employee of the other Party for any purpose.
- 17.10. <u>Governing Law</u>. Any dispute arising out of or relating to this Agreement or the breach thereof shall be governed by the laws of the state of the domicile of Client, without regard to or application of choice of law rules or principles.
- 17.11. Audit. Tyler shall maintain complete and accurate records of all work performed pursuant to and arising out of this Agreement. Client may, upon the written request of the Project Manager, audit any and all work or expense records of Tyler relating to materials and/or services provided herein. Client shall provide Tyler twenty-four hour notice of such audit or inspection. Tyler shall have the right to exclude from such inspection any Tyler Confidential and Proprietary Information not otherwise required to be provided to Client as a part of this Agreement. Tyler shall make such books and records available to Client during normal business hours. Any such audit shall be conducted at Tyler's principal place of business during Tyler's normal business hours and at Client's sole expense.
- 17.12. No Third Party Beneficiaries. Nothing in this Agreement is intended to benefit, create any rights in, or otherwise vest any rights upon any third party.
- 17.13. <u>Contra Proferentem</u>. The doctrine of *contra proferentem* shall not apply to this Agreement. If an ambiguity exists in this Agreement, or in a specific provision, neither the Agreement nor the provision shall be construed against the Party who drafted the Agreement or provision.
- 17.14. Force Majeure. No Party to this Agreement shall be liable for delay or failure in the performance of its contractual obligations arising from any one or more events that are beyond its reasonable control, including, without limitation, acts of God, war, terrorism, and riot. Upon such delay or failure affecting one Party, that Party shall notify the other Party and use all reasonable efforts to cure or alleviate the cause of such delay or failure with a view to resuming performance of its contractual obligations as soon as practicable. Notwithstanding the foregoing, in every case the delay or failure to perform must be beyond the control and without the fault or negligence of the Party claiming excusable delay. Any performance times pursuant to or arising from this Agreement shall be considered extended for a period of time equivalent to the time lost because of any delay that is excusable herein.
- 17.15. <u>Equitable Relief.</u> Each Party covenants, represents, and warrants that any violation of this Agreement by such Party with respect to its respective obligations set forth in Sections 2.3 and 9 shall cause irreparable injury to the other Party and shall entitle the other Party to extraordinary and equitable relief by a court of competent jurisdiction, including, without limitation, temporary restraining orders and preliminary and permanent injunctions, without the necessity of posting bond or security.
- 17.16. Attornevs' Fees and Costs. If attorneys' fees or other costs are incurred by either Party to secure the performance of any obligations under this Agreement, or to establish damages for the breach thereof or to obtain any other appropriate relief, whether by way of prosecution or defense, the prevailing Party shall be entitled to recover from the other Party its reasonable attorneys' fees and costs incurred in connection therewith.

[Remainder of this page intentionally left blank]

(Exhibit B) Service Level Terms and Conditions

1. CERTAIN DEFINITIONS

- 1.1. <u>Terms Not Defined</u>. Terms not otherwise defined in this Exhibit B shall have the meanings assigned to such terms in the Software as a Service and Professional Services Agreement (the "Agreement").
- 1.2. <u>Application Availability Period</u> has the meaning set forth in Schedule B-1.
- 1.3. <u>Circumvention or Circumvention Procedures</u> means, as applied to a Documented Defect, a change in operating procedures whereby Client can reasonably avoid any deleterious effects of such Documented Defect.
- 1.4. <u>Defect</u> means any bug, error, malfunction, or other defect in the Licensed Software caused by, arising from, or emanating from the reasonable control of Tyler that renders the Licensed Software in non-conformance with Tyler's then current published specifications.
- 1.5. <u>Documented Defect</u> means a Defect that Client documents for Tyler pursuant to Section 2.1.
- 1.6. <u>Downtime</u> means minutes during the Application Availability Period where the Licensed Software is not available as set forth in Section 3.1
- 1.7. <u>Operational Maintenance Window</u> has the meaning set forth in Schedule B-1.
- 1.8. <u>Service Level 1 Defect</u> Documented Defect that causes (a) complete application failure or application unavailability.; (b) application failure or unavailability in one or more of Client's remote locations; (c) systemic loss of multiple essential system functions.
- 1.9. <u>Service Level 2 Defect</u> means a Documented Defect that causes (a) repeated, consistent failure of essential functionality affecting more than one user or (b) loss or corruption of data.
- 1.10. <u>Service Level 3 Defect</u> means a Priority 1 Defect with an existing Circumvention Procedure, or a Priority 2 Defect that affects only one User or for which there is an existing Circumvention Procedure.
- 1.11. <u>Service Level 4 Defect</u> means a Documented Defect that causes failure of non-essential Licensed Software functionality or a cosmetic or other Documented Defect that does not qualify as any other Service Level Defect.
- 1.12. <u>Version Release</u> means new versions of the Licensed Software that contain technical improvements, functional enhancements, updates, extensions, and/or maintenance changes to the Licensed Software.
- 1.13. <u>Tyler Holidays</u> means one (1) day for a New Year's holiday, Good Friday, Memorial Day, a one (1) day holiday for Independence Day, Labor Day, Thanksgiving Day and the day after, and two (2) days during Christmas time. The exact date for any rolling holiday will be published on the Tyler website in advance of the date.

2. CLIENT RESPONSIBILITIES

- 2.1. <u>Documenting Defects</u>. Client must document all Defects in writing with sufficient information to recreate the Defect or otherwise clearly and convincingly document or evidence its occurrence, including, but not limited to, the operating environment, data set, user, or any other such information that Tyler may reasonably request. Client shall deliver such information to Tyler concurrently with its notification to Tyler of a Defect. Client shall use all reasonable efforts to eliminate any non-application related issues prior to its notification to Tyler of such Defect, including, but not limited to, issues related to the network, User training, Client-produced extensions, and data problems not caused by the Licensed Software. Any technical or other issue for which Client requests services, but which is not a Documented Defect, shall be treated as a request for other services and governed by Section 6.
- 2.2. <u>Training</u>. Except as provided in Exhibit A of the Agreement, Client shall provide training to its employees on the Licensed Software and any Version Releases related thereto.

3. TYLER RESPONSIBILITIES – APPLICATION AVAILABILITY AND OPERATIONS SUPPORT

3.1. Application Availability.

- (a) Tyler shall use its commercially reasonable efforts to provide access to the Licensed Software during the Application Availability Period as set forth in the goals listed in Schedule 8-1
- (b) Tyler shall maintain a log of any system issues that result in Downtime of more than 1 hour, excluding: (i) scheduled maintenance by Tyler's Internet Service Provider or colocated data center; (ii) periods needed to deter or correct problems due to malicious attacks or denial of service attempts; (iii) Client hardware or network failure; (iv) negligent actions by Client's agents, employees, or vendors; and (v) events of Force Majeure (as set forth in Exhibit A, Section 17.14).

3.2. Operations Support: Procedures for Reporting Downtime.

- (a) Tyler shall provide Client with procedures for contacting support staff on a twenty-four hour, seven days a week basis for the limited purpose of reporting Downtime. Client agrees to designate no more than two (2) of Client's employees who are authorized to utilize this procedure after normal Business Hours
- (b) For each reported Downtime incident, Tyler shall assign appropriate personnel to diagnose and correct the Downtime. Tyler's initial response shall include an acknowledgement of notice of the Downtime, confirmation that Tyler has received sufficient information concerning the Downtime, and an action plan for resolving the Downtime.

3.3. Credit for Downtime.

- (a) At the end of each calendar quarter, Tyler shall prepare a report for the prior three months detailing the average percentage of Downtime during that three month period.
- (b) Tyler shall issue a credit towards the next quarterly payment as follows:
 - if the average percentage of Downtime is less than 4%, no Downtime credit shall be issued;
 - (ii) if the average percentage of Downtime is between 4% and 6% of the Application Availability Period, Tyler shall issue a Downtime credit towards 3% of the next quarterly payment due; and
 - (iii) if the average percentage of Downtime is greater than 6% of the Application Availability Period, Tyler shall issue a Downtime credit towards 5% of the next quarterly payment due.
- (c) The issuance by Tyler of any Downtime credit shall not relieve Tyler of its obligations to coπect the problem that resulted in Downtime in accordance with its obligations herein. However, Client acknowledges that correction may occur in the following quarter and, because of the time reasonably needed to perform any such correction, the quarter in which the correction occurs may also be affected by Downtime.
- (d) Notwithstanding the foregoing, the total of all credits that would be due under this SLA shall not exceed 5% of the quarterly SaaS Fee for any one quarter.

4. TYLER RESPONSIBILITIES - HELP DESK

Tyler shall provide Client with procedures for contacting support staff during normal business hours (7:00 a.m. to 7:00 p.m., Central Time,

0Monday through Friday, excluding Tyler holidays) for general application assistance.

5. TYLER RESPONSIBILITIES -DOCUMENTED DEFECTS

- 5.1. General Services for Reporting Documented Defects.
- (a) Tyler shall provide Client with procedures for contacting support staff during normal business hours (7:00 a.m. to 7:00 p.m., Central Time, Monday through Friday, excluding Tyler holidays) for reporting Documented Defects. Tyler shall assist Client in the diagnosis of any Documented Defect, including the assigned Priority and Tyler's tracking number.
- (b) For each reported Documented Defect, Tyler shall assign appropriate personnel to diagnose and correct the Documented Defect, and where appropriate, identify Circumvention Procedures. Tyler's initial response shall include an acknowledgement of notice of the Documented Defect, confirmation that Tyler has received sufficient information concerning the Documented Defect, and an action plan for resolving the Documented Defect and avoiding further deleterious consequences of the Documented Defect.
- 5.2. <u>Service Level 1 Defects</u> Tyler shall provide an initial response to Service Level 1 Defects within one (1) Business Hour of receipt of the Documented Defect. Tyler shall use commercially reasonable efforts to resolve such Documented Defects or provide a Circumvention Procedure within one (1) Business Day. Tyler's responsibility for loss or corrupted data is limited to assisting Client in restoring its database to a known, accurate state.
- 5.3. <u>Service Level 2 Defects</u>. Tyler shall provide an initial response to Service Level 2 Defects within one (1) Business Hour of receipt of the Documented Defect. Tyler shall use commercially reasonable efforts to resolve such Documented Defects or provide a Circumvention Procedure within five (5) Business Days. Tyler's responsibility for loss or corrupted data is limited to assisting Client in restoring its database to a known, accurate state
- 5.4. <u>Service Level 3 Defects</u>. Tyler shall provide an initial response to Service Level 3 Defects within one (1) Business Day of receipt of the Documented Defect. Tyler shall use commercially reasonable efforts to resolve such Documented Defect without the need for a Circumvention Procedure with the next published maintenance update or service pack, which shall occur at least quarterly. Tyler's responsibility for lost or corrupted data is limited to assisting Client in restoring its database to a known, accurate state.
- 5.5. <u>Service Level 4 Defects</u>. Tyler shall provide an initial response to Service Level 4 Defects within two (2) Business Days. Tyler shall use commercially reasonable efforts to resolve such Documented Defect with a future Version Release.
- 5.6. <u>Technical Server & Systems Support.</u> Tyler shall use commercially reasonable efforts to provide Client with technical support to assist Client with troubleshooting the loss of functionality of Licensed

Software for reasons other than a Documented Defect. Tyler technical support shall be limited to:

- (a) assisting Client with isolating the source of Licensed Software failure due to systems-level hardware, Third Party Software, network, client-level hardware or peripherals;
- (b) providing recommendations to Client regarding resolution of said non-defect failure(s); and
- (c) providing Client with assistance on basic maintenance and administration of the Licensed Software environment, including basic data backup and restore procedures, deployment of Version Releases, and setup of supported peripheral devices for use with the Licensed Software

6. ADDITIONAL SUPPORT SERVICES

Client may request support services in addition to the correction of Documented Defects by delivering to Tyler a written request outlining the nature of the services desired (a "Service Request"). Such other support services may include, without limitation, services related to: (a) additional training; (b) technical assistance; (c) programming services; (d) installation of add-on components; and/or (e) business analysis. Tyler shall provide to Client a written response to the request which describes in detail the anticipated impact of the request on the existing Licensed Software, the time required to perform such services, an implementation plan, and a schedule of the fees related thereto. Fees for additional support services shall be billed by Tyler directly to Client and shall be invoiced monthly, which shall be due and payable within thirty (30) days.

7. VERSION RELEASES

Tyler shall provide Version Releases of the Licensed Software on not less than an annual basis. Tyler shall notify Client of the occurrence of a new Version Release and shall provide Client with access to such Version Releases for the Licensed Software in a prompt manner that is consistent with Tyler's business practices for client's utilizing the Licensed Software under a software as a service agreement.

8. THIRD PERSON SOFTWARE

- 8.1. <u>Notice of New Third Person Software</u>. Tyler shall provide Client with advanced notice of any mandated new Third Person Software revision that shall be required to use the Licensed Software. Tyler shall, to the extent practicable, minimize the need for Client to rely upon updates of Third Person Software.
- 8.2. <u>Tyler Certification</u>. At Tyler's expense, Tyler shall certify the compatibility of Third Person Software components used by the Licensed Software and maintain a list of supported Third Person Software release levels. Version Releases shall be certified to supported versions of all required Third Person Software. Tyler shall certify new releases of Third Person Software within a reasonable timeframe.
- 8.3. <u>Costs</u>. Client is responsible for all costs associated with installing and maintaining Third Person Software versions that are identified on Tyler's list of certified Third Person Software.

[Remainder of this page intentionally left blank]

(Schedule B-1) Application Availability Period Service Levels

Туре	Description	Goal
Application Availability Period	All operational time as set forth under "Goal", and which is outside the Operational Maintenance Window, and where Tyler has not announced its intent to perform maintenance at least forty-eight (48) hours in advance.	10:00 a.m. Central Time Sunday to 06:00 a.m. Central Time Sunday Total of 166 hours per week.
Operational Maintenance Window	The Operational Maintenance Window happens weekly. During this time, Tyler can take its Odyssey servers off-line (no Internet access) and perform work on supporting hardware. Tyler will provide 48 hours notice to the Client if the Odyssey application will be unavailable during the maintenance period. Application maintenance includes upgrades or replacements of Tyler servers, data storage, data backup, and supporting hardware. This period also covers software maintenance items that include scheduled hot fixes, quarterly service releases, operating system security patches and upgrades, and so forth. If an Odyssey application hot fix must be performed outside the Operational Maintenance Window and impacts application availability, Tyler will provide 24-hour notice to the Client.	06:00 a.m. to 10:00 a.m. Central Time Sunday.
Backups	Nightly backups of the following files will be completed: production databases, images, forms, and other documents. Client data transactions are saved every 15 minutes during the Application Availability Period. Every night, a full database backup is performed, including client images, forms, and other documents. Back-up media will be cycled off-site nightly to a fireproof vault. Nightly backups are stored offsite.	Nightly

CERTIFICATE OF INTERESTED PARTIES

FORM 1295

1 of 2

	Complete Nos. 1 - 4 and 6 if there are interested parties. Complete Nos. 1, 2, 3, 5, and 6 if there are no interested parties.	OFFICE USE ONLY CERTIFICATION OF FILING
1	-86	Certificate Number: 2016-146341
	Plano, TX United States	Date Filed:
2	Name of governmental entity or state agency that is a party to the contract for which the form is being filed.	12/14/2016
	Fort Bend County	Date Acknowledged:

3 Provide the identification number used by the governmental entity or state agency to track or identify the contract, and provide a description of the services, goods, or other property to be provided under the contract.

14006 Tyler Technologies, Inc.

Court Administration System Maintenance

4 Name of Interested Party	City, State, Country (place of business)	Nature of interest (check applicable)	
		Controlling	Intermediary
Pope, Daniel M.	Plano, TX United States	×	
Brattain, Donald R.	Plano, TX United States	×	
Carter, Glenn A.	Plano, TX United States	х	
Cline, Brenda A.	Plano, TX United States	×	
King, Jr., J. Luther	Plano, TX United States	×	
Leinweber, Larry D.	Plano, TX United States	×	
Marr, Jr., John S.	Yarmouth, ME United States	х	
Miller, Brian K.	Plano, TX United States	×	
Moore, Jr., H. Lynn	Plano, TX United States	х	
Womble, Dustin R.	Lubbock, TX United States	×	
Yeaman, John M.	Plano, TX United States	×	12.

CERTIFICATE OF INTERESTED PARTIES FORM 1295 OFFICE USE ONLY Complete Nos. 1 - 4 and 6 if there are interested parties. Complete Nos. 1, 2, 3, 5, and 6 if there are no interested parties. **CERTIFICATION OF FILING** 1 Name of business entity filing form, and the city, state and country of the business entity's place Certificate Number: of business. 2016-146341 Tyler Technologies, Inc. Date Filed: Plano, TX United States 12/14/2016 2 Name of governmental entity or state agency that is a party to the contract for which the form is being filed. Date Acknowledged: **Fort Bend County** Provide the identification number used by the governmental entity or state agency to track or identify the contract, and provide a description of the services, goods, or other property to be provided under the contract. 14006 Tyler Technologies, Inc. Court Administration System Maintenance Nature of interest City, State, Country (place of business) (check applicable) Name of Interested Party Intermediary Controlling 5 Check only if there is NO Interested Party. AFFIDAVIT I swear, or affirm, under penalty of perjury, that the above disclosure is true and correct, SOPHIA L. THOMAS Notary Public, State of Texas Comm. Expires 06-23-2018 Signature of authorized agent of contracting business entity Notary ID 126488736 AFFIX NOTARY STAMP / SEAL ABOVE Sworn to and subscribed before me, by the said __Craig Seekamp _____, this the 14 ____ day of December , to certify which, witness my hand and seal of office. na L Thomas

Signature of officer administering oath

Printed name of officer administering oath

Title of officer administering oath